

Implementation Agreement

Dated: 12 December 2017

Unibail-Rodamco SE (Unibail)

Westfield Corporation Limited (ABN 12 166 995 197) (WCL), Westfield America Management Limited (ABN 66 072 780 619) (WAML) as responsible entity of Westfield America Trust (ABN 27 374 714 905) (WAT) and as responsible entity of WFD Trust (ABN 50 598 857 938) (WFDT) (together or any of them individually as the context requires, Westfield)

King & Wood Mallesons

Level 61
Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia
T +61 2 9296 2000
F +61 2 9296 3999
DX 113 Sydney
www.kwm.com

Implementation AgreementContents

Detail	s	1
Gene	ral terms	2
1	Definitions and interpretation	2
1.1	Definitions	2
1.2	General interpretation	23
2	Agreement to propose Transaction	24
2.1	Contemplated Transaction	24
2.2	ERB consultation	24
3	Conditions Precedent	26
3.1	Conditions Precedent	26
3.2	Reasonable endeavours	29
3.3	Regulatory matters	29
3.4	Waiver of Conditions Precedent	30
3.5	Notices in relation to Conditions Precedent	31
3.6	Consultation on failure of Condition Precedent	31
3.7	Failure to agree	32
4	Outline of Schemes	32
4.1	Outline of Schemes	32
4.2	Relationship between Share Scheme and Trust Schemes,	
	and the Schemes and the Destapling	32
4.3	Scheme Consideration	32
4.4	Provision of Westfield Register	33
4.5	Employee incentives	33
4.6	No amendment to the Schemes without consent	33
5	Implementation	33
5.1	General obligations	33
5.2	Westfield's preliminary obligations	33
5.3	Westfield's post-Consideration Period obligations	36
5.4	Unibail's preliminary obligations	38
5.5	Unibail's post-Consideration Period obligations	39
5.6	Transaction Steps	41
5.7	Responsibility statement	42
5.8	Disagreement on disclosure	42
5.9	Verification	43
5.10	Conduct of Scheme Court proceeding	43
5.11	Appeal process	43
5.12	No partnership or joint venture	43
6	Recommendations	44
6.1	Westfield Board recommendation	44
6.2	Westfield Board Voting Intention	44
6.3	Westfield Directors	44
6.4	Notification obligation	44

© King & Wood Mallesons Implementation Agreement

6.5	Unibail Board recommendation	44
7	Directors and employees	45
7.1 7.2	Release of Westfield and Westfield directors and officers Benefit for Westfield Indemnified Parties	45 45
7.3	Release of Unibail and Unibail directors and officers	45
7.4	Benefit for Unibail Indemnified Parties	45
7.5	Appointment/retirement of directors	45
7.6	Directors' and officers' insurance	46
7.7	Run-off	46
7.8	Period of undertaking	47
7.9	Benefit of undertaking for Westfield Group	47
8	Westfield conduct of business	47
8.1	Westfield conduct	47
8.2	Prohibited actions	48
8.3	Exceptions to conduct of business provisions	51
8.4	Access to people and Westfield Information	51
8.5	OneMarket Demerger	51
8.6	Integration planning	52
8.7	Change of control consents	53
9	Unibail conduct of business	54
9.1	Unibail conduct	54
9.2	Exceptions to conduct of business provisions	55
10	Exclusivity	55
10.1	No existing discussions	55
10.2	No-shop	55
10.3	No-talk	55
10.4	Due diligence information	56
10.5	Exceptions	56
10.6	Further exceptions	56
10.7	Notice of approaches	56
10.8	Matching right	57
10.9	Unibail counterproposal	57
10.10 10.11	Provision of Information Legal advice	58 58
11	Break Fee	58
11.1	Background	58
11.2	Payment by Westfield to Unibail	59
11.3	No amount payable if Schemes become Effective	59
11.4 11.5	Timing of payment Nature of payment	59 60
11.6	Break Fee does not limit liability	60
	•	
12	Reverse Break Fee	60
12.1	Background	60
12.2	Payment by Unibail to Westfield	61
12.3 12.4	No amount payable if Schemes become Effective	61
12.4	Timing of payment	61 62
12.5	Nature of payment Reverse Break Fee does not limit liability	62 62
12.0	Westerge Dieav i de noes hot illilli liabilità	02

© King & Wood Mallesons 35219573_1

13	Representations and warranties	62
13.1	Westfield's representations and warranties	62
13.2	Timing of Westfield's representations and warranties	70
13.3	Unibail's representations and warranties	71
13.4	Timing of Unibail's representations and warranties	76
13.5	Notification	76
14	Termination	76
14.1	Termination events	76
14.2	Termination	77
14.3	Effect of Termination	77
14.4	Damages	77
14.5	Non-application to ERB termination	77
15	Limitation of Liability	77
16	Public announcements	78
16.1	Public announcement of Schemes	78
16.2	Required disclosure	78
16.3	Other announcements	78
17	Confidential Information	78
18	Notices and other communications	79
18.1	Form	79
18.2	Delivery	79
18.3	When effective	79
18.4	When taken to be received	79
18.5	Receipt outside business hours	79
19	GST and VAT	80
19.1	Definitions and interpretation	80
19.2	GST exclusive and VAT exclusive	80
19.3	Payment of GST and VAT	80
19.4	Adjustment events	80
19.5	Reimbursements	80
20	Costs	81
20.1	Costs	81
20.2	Stamp duty and registration fees	81
21	General	81
21.1	Variation and waiver	81
21.2	Consents, approvals or waivers	81
21.3	Discretion in exercising rights	81
21.4	Partial exercising of rights	81
21.5	Conflict of interest	82
21.6	Remedies cumulative	82
21.7	Indemnities and reimbursement obligations	82
21.8	Inconsistent law	82
21.9	Supervening law	82
21.10	Counterparts	82
21.11	Entire agreement	82

21.12	Furthe	82	
21.13	No lia	83	
21.14	Sever	rability	83
21.15	Rules	of construction	83
21.16	Assig	nment	83
21.17	Enfor	ceability	83
21.18	No re	presentation or reliance	83
22	Gove	rning law	84
22.1	Gove	rning law and jurisdiction	84
22.2	Servir	ng documents	84
22.3	Servi	ce of process	84
Sched	lule 1	Timetable (clause 5.1)	85
Sched	lule 2	Transaction Steps	86
Sched	lule 3	Newco BI	87
Signin	ıg page	•	88
Annex	ure A	Public announcement	90
Annexure B		Scheme of Arrangement	91
Annexure C		Deed Poll	92
			92
Annex	ure D	Westfield Constitutional Amendments	93

Implementation Agreement

Details

Parties

Parties				
Unibail Name			Unibail-Rodamco SE	
	Regist numbe		682 024 096	
	Forme	d in	France	
	Addres	SS	7, place du Chancelier Adenauer, 75016 Paris France	
	Email		David.Zeitoun@unibail-rodamco.com	
	Attenti	on	David Zeitoun	
Westfield	Name		Westfield Corporation Limited (ABN 12 166 995 197) (WCL)	
			Westfield America Management Limited (ABN 66 072 780 619) (WAML) as responsible entity of Westfield America Trust (ABN 27 374 714 905) (WAT) and as responsible entity of WFD Trust (ABN 50 598 857 938) (WFDT)	
	Forme	d in	Australia	
	Addres	SS	Level 29, 85 Castlereagh Street, Sydney NSW 2000, Australia	
	Email		STuxen@westfield.com	
	Attenti	on	Simon Tuxen	
Governing law	New S	South Wales, Australia		
Recitals	Α	Unibail and	d Westfield propose to implement the Transaction.	
	В	To implement the Transaction, Westfield intends to propose the Schemes.		
	С	To implement the Transaction, Unibail intends to seek the Unibail Shareholder Approval.		
	D	Westfield and Unibail have agreed to implement the Schemes on the terms and conditions of this document.		

Implementation Agreement

General terms

1 Definitions and interpretation

1.1 Definitions

Unless the contrary intention appears, these meanings apply:

Additional Issuance Period has the meaning set out in clause 2.2(c)(iii).

AFM means the Dutch *Autoriteit financiële markten*.

AMF means the French Autorité des marchés financiers.

ASIC means the Australian Securities & Investments Commission.

Associate has the meaning set out in section 12 of the Corporations Act, and, in the case of Westfield, as if subsection 12(1) of the Corporations Act also included a reference to this document and Westfield was the designated body.

ASX means ASX Limited or the market operated by it, as the context requires.

ASX Listing Rules means the listing rules of ASX and any other applicable rules of ASX modified to the extent of any express written waiver by ASX.

Authorised Officer means a director or secretary of a party or any other person nominated by a party to act as an Authorised Officer for the purposes of this document.

Break Fee means an amount equal to US\$150,000,000.

Business Day means a business day as defined in the ASX Listing Rules and on which trading will take place on Euronext Paris and Euronext Amsterdam and the New York Stock Exchange.

Claim means, in relation to a party, a demand, claim, action or proceeding made or brought against the party, however arising and whether present unascertained, immediate, future or contingent.

Competing Proposal means a proposal, offer transaction or arrangement (whether by way of takeover bid, scheme of arrangement, capital reduction, sale of assets, sale or issue of securities, joint venture or otherwise) which, if completed substantially in accordance with its terms, would:

- (a) mean a person (other than Unibail or its Related Bodies Corporate) whether alone or together with its Associates would:
 - (i) directly or indirectly, acquire, or have a right to acquire, a legal, beneficial or economic interest in, or Relevant Interests in, or Control of, 20% or more of the Westfield Stapled Securities (other than as custodian, nominee or bare trustee);
 - (ii) acquire Control of Westfield;

© King & Wood Mallesons | Implementation Agreement 2

- directly or indirectly acquire, obtain a right to acquire, or (iii) otherwise obtain an economic interest in, or control of, all or a substantial part or a material part of the assets of or business conducted by the Westfield Group; or
- (iv) otherwise directly or indirectly acquire or merge (including by a reverse takeover bid or dual listed company structure) with Westfield: or
- (b) require, or be reasonably likely to require, Unibail to abandon, or otherwise fail to proceed with, the Transaction or any part of the Transaction, by whatever means,
- (c) and, in each case, includes any variation or modification of an earlier Competing Proposal.

Conditions Precedent means the conditions precedent set out in clause 3.1.

Confidentiality Deed means the Confidentiality Deed between the parties dated 6 November 2017.

Consideration Period means the Consultation Period plus a period of 5 Business Days immediately following the date of completion or deemed completion of the Consultation Period in accordance with clause 2.2(d).

Consultation Period means the period from the time the Consultation Process is initiated in accordance with clause 2.2(b) until the date the Consultation Process is completed or deemed completed in accordance with clause 2.2(d).

Consultation Process means the consultation with the EEC pursuant to Article 10 of the ECC Agreement and with Unibail UES Works Council in accordance with the relevant agreement with that body.

Contemplated Transaction means the transactions contemplated by clause 2.1.

Control has the meaning given in section 50AA of the Corporations Act.

Controlled Entities means, in relation to an entity, another entity which is a Subsidiary of it, or which is Controlled by it (as contemplated by section 50AA of the Corporations Act).

Controller has the meaning it has in the Corporations Act or other applicable

Corporations Act means the *Corporations Act 2001* (Cth) (Australia).

Corporations Regulations means the *Corporations Regulations 2001* (Cth) (Australia).

Costs includes costs, charges and expenses, including those incurred in connection with advisers and any legal costs on a full indemnity basis.

Court means the Federal Court of Australia, the Supreme Court of an Australian state, the French Court of Appeals (Cour d'appel) and Court of Cassation (Cour de cassation), a Dutch district court (rechtbank), a Dutch court of appeals (gerechtshof), the Dutch Supreme Court (Hoge Raad), the European Union Court of Justice or such other court of competent jurisdiction agreed by the parties.

Deed Poll means a deed poll substantially in the form of Annexure C to this document.

Destapling means the destapling of WCL, WAT and WFDT as described in the Transaction Steps.

Details means the section of this document headed "Details".

Disclosed means Fairly Disclosed:

- by Unibail to Westfield (including in the electronic data room made (a) available to Westfield) (as applicable), in writing prior to the date of this document;
- (b) in the Westfield Disclosure Materials;
- in any announcement made by Westfield on ASX or by Unibail on (c) Euronext Paris and/or Euronext Amsterdam (as applicable) or through a press release that is published in accordance with relevant laws and regulations and/or filed with the relevant regulator, prior to the date of this document; or
- by Unibail in the Unibail Reports. (d)

EEC means the European Employees' Committee of Unibail.

EEC Agreement means the agreement relating to the involvement of the employees of Unibail entered into on 27 February 2009 between Unibail and the members of the special negotiating body (as amended from time to time).

Effective, when used in relation to the Transaction and the Schemes, is all of the following events taking place:

- (a) the coming into effect, pursuant to section 411(10) of the Corporations Act, of the order of the Scheme Court made under section 411(4)(b) in relation to the Share Scheme, but in any event at no time before an office copy of the order of the Scheme Court is lodged with ASIC; and
- the Westfield Constitutional Amendments taking effect pursuant to (b) section 601GC(2) of the Corporations Act.

Effective Date means the date on which the Schemes become Effective.

Employee Right means any rights to Westfield Stapled Securities issued under employee incentive arrangements by the Westfield Group.

Encumbrance means any security for the payment of money or performance of obligations, including a mortgage, charge, lien, pledge, trust, power or title retention or flawed deposit arrangement and any "security interest" as defined in sections 12(1) or 12(2) of the PPSA or any agreement to create any of them or allow them to exist.

End Date means 30 September 2018, or such other date as is agreed by the parties.

ERB means the EEC and the Unibail UES Works Council.

ERB Opinion has the meaning set out in clause 2.2(d).

Euronext Approval means the approval, by the relevant competent body of each of Euronext Paris and Euronext Amsterdam, as applicable, of the application of Unibail and Newco BI to an admission to listing and trading of the New Unibail Stapled Securities on Euronext Paris and Euronext Amsterdam.

Euronext Amsterdam means the regulated market operated by Euronext Amsterdam NV.

Euronext Paris means the regulated market operated by Euronext Paris S.A.

Euronext Rules means the harmonized Euronext rules as set out in Euronext Rulebook I and the specific rules applicable to Euronext Amsterdam and Euronext Paris as set out in Euronext Rulebook II and any Euronext notice or other publication setting out rules applicable to either Euronext Amsterdam or Euronext Paris.

Exclusivity Period means the period from and including the date of this document to the earlier of:

- the termination of this document in accordance with its terms; and (a)
- the End Date. (b)

FCPA means the Foreign Corrupt Practices Act of 1977 (15 U.S.C. §§78dd-1, et seg.), as amended, and any rules, regulations and guidance promulgated thereunder.

FIRB means the Foreign Investment Review Board.

FIRB Act means the Foreign Acquisitions and Takeovers Act 1975 (Cth) (Australia).

Fairly Disclosed means fully, fairly and accurately disclosed, with sufficient detail so as to enable a reasonable and sophisticated investor to identify the nature and scope (where applicable) of the relevant matter, event or circumstance.

First Court Date means the first day on which an application made to the Scheme Court, in accordance with clause 5.3(a), is heard for orders:

- under section 411(1) of the Corporations Act convening the Share (a) Scheme Meeting to consider the Share Scheme; and
- (b) for the First Judicial Advice.

First Judicial Advice means confirmation from the Scheme Court under section 63 of the Trustee Act 1919 (NSW) that WAML would be justified in convening the Trust Scheme Meetings and proceeding on the basis that amending the Trust Constitutions as set out in the Westfield Constitutional Amendments would be within the powers of alteration conferred by the Trust Constitutions and section 601GC of the Corporations Act.

Foreign Nominee has the meaning it is given in the Share Scheme.

Guidance Note 15 means the current document entitled "Guidance Note 15: Trust Scheme Mergers" issued by the Takeovers Panel.

Implementation Date means the 5th Business Day following the Record Date or such other date after the Record Date as Unibail and Westfield agree in writing.

Incoming Unibail Directors means one of Peter Lowy and Stephen Lowy as nominated by Westfield.

Incoming Westfield Directors means such persons as nominated by Unibail.

Independent Expert means the independent expert appointed by Westfield under clause 5.2(c).

Independent Expert's Report means the report from the Independent Expert for inclusion in the Scheme Booklet, including any update or supplementary report, stating whether or not in the Independent Expert's opinion the Schemes are in the best interests of Westfield Securityholders.

Ineligible Foreign Holders has the meaning it is given in the Share Scheme.

Insolvent means, if:

- (a) in relation to any person that is an entity:
 - (i) it is (or states that it is) an insolvent under administration or insolvent (each as defined in the Corporations Act or other applicable law); or
 - (ii) it is in liquidation, in provisional liquidation, under administration or wound up or has had a Controller appointed to any part of its property; or
 - (iii) it is subject to any arrangement, assignment, moratorium or composition, protected from creditors under any statute or other applicable law or dissolved (in each case, other than to carry out a reconstruction or amalgamation while solvent on terms approved by the other parties to this document); or
 - (iv) an application or order has been made (and in the case of an application, it is not stayed, withdrawn or dismissed within 14 days), resolution passed or any other action taken, in each case in connection with that person, in respect of any of the things described in paragraphs (a)(i), (a)(ii) or (a)(iii) above;
 - it is taken (under section 459F(1) of the Corporations Act or (v) other applicable law) to have failed to comply with a statutory demand: or
 - it is the subject of an event described in section 459C(2)(b) or (vi) section 585 of the Corporations Act or other applicable law (or it makes a statement from which another party to this document reasonably deduces it is so subject); or
 - (vii) it is otherwise unable to pay its debts when they fall due; or
 - (viii) something having a substantially similar effect to (a)(i) to (a)(vii) happens in connection with that person under the law of any iurisdiction: and
- (b) in relation to a trust only:
 - (i) the trustee, relying on its right of indemnity out of trust assets, is not able to pay all the debts of the trust as and when they fall due:
 - (ii) the trustee, a court of competent jurisdiction or another authorised person determines or orders that the trust is wound up; or
 - something having a substantially similar effect to any of the (iii) things described in (b)(i) or (b)(ii) above.

Integration Committee has the meaning given in clause 8.6(b).

Judicial Advice means the First Judicial Advice and the Second Judicial Advice.

Key Management Personnel has the meaning given in section 9 of the Corporations Act.

Losses means all claims, demands, damages, losses, costs, expenses and liabilities.

Newco BI means a public company with limited liability (naamloze vennootschap) incorporated or to be incorporated in the Netherlands in accordance with step 1 of the Transaction Steps with shares as described in Schedule 3.

Newco BI Share means a class A ordinary share in the capital of Newco BI as described in Schedule 3.

New Unibail means Unibail and Newco BI (a stapled group).

New Unibail CDI means CHESS Depositary Interests over New Unibail Stapled Securities issued by CHESS Depositary Nominees Ltd pursuant to the clearing and settlement facility operated by ASX.

New Unibail Shares means the Unibail Shares to be issued in the context of the Share Scheme and the Transaction.

New Unibail Shares Listing Prospectus means the prospectus (*Document E*) to be prepared by Unibail, approved by the AMF and notified to the AFM, in order for the New Unibail Shares to be admitted to trading on Euronext Paris and Euronext Amsterdam.

New Unibail Shares Listing Prospectus Approval means the approval by the AMF of the New Unibail Shares Listing Prospectus.

New Unibail Stapled Securities means a Unibail Share and a Newco BI Share stapled together in accordance with the articles of association of Unibail and Newco BI and in compliance with applicable French and Dutch laws.

New Unibail Stapled Securities Listing Prospectus means the prospectus to be prepared by Unibail and approved by the AMF and/or the AFM, as applicable, in order for the New Unibail Stapled Securities to be admitted to trading on Euronext Paris and Euronext Amsterdam.

New Unibail Stapled Securities Listing Prospectus Approval means the approval, by the AMF and/or the AFM, as applicable, of the New Unibail Stapled Securities Listing Prospectus.

OneMarket means OneMarket Limited.

OneMarket Demerger means the demerger of OneMarket from WCL as provided for in clause 8.5.

OneMarket Demerger Deeds means the demerger implementation deed and demerger deed referred to in clause 8.5(b).

Outgoing Westfield Directors means each of the existing Westfield Directors.

PPSA means the Personal Property Securities Act 2009 (Cth) (Australia).

Preliminary Provisions means clauses 3, 5.1 (in relation to the obligations referred to in this definition other than this reference), 5.2, 5.4, 6 and 8 to 22 of this document.

Preserved Clauses means clauses 11, 12 and 15 to 22.

Record Date means 5.00pm on the 4th Business Day following the Effective Date (or such other date as required or approved by ASX) or such other date as the parties may agree in writing.

Regulator's Draft means the draft of the Scheme Booklet in a form acceptable to all of the parties which is provided to ASIC for approval pursuant to section 411(2) of the Corporations Act and for the purposes of Regulatory Guide 74 and to ASX under ASX Listing Rules 15.1, 15.1.1 and 15.1.7.

Regulatory Approval means any approval of a Regulatory Authority to the Schemes or any aspect of it which is necessary or desirable to implement the Schemes.

Regulatory Authority includes:

- (a) ASX, Euronext Paris and Euronext Amsterdam;
- Australian Competition and Consumer Commission, ASIC, the (b) Takeovers Panel, FIRB and the Australian Taxation Office;
- (c) AMF, AFM, the European Commission, Confédération Fiscale Européenne and Belastingdienst;
- (d) a government or governmental, semi-governmental or judicial entity or authority:
- a minister, department, office, commission, delegate, instrumentality, (e) agency, board, authority or organisation of any government; and
- (f) any regulatory organisation established under statute.

Regulatory Guide 60 means the current "Regulatory Guide 60 (Schemes of arrangement)" issued by ASIC.

Regulatory Guide 74 means the current "Regulatory Guide 74 (Acquisitions agreed to by shareholders)" issued by ASIC.

Related Body Corporate has the meaning it has in the Corporations Act, except that the term "subsidiary" used in the Corporations Act shall have the meaning ascribed to "Subsidiary" in this document.

Related Party has the meaning it has in the Corporations Act.

Relevant Interest has the meaning it has in sections 608 and 609 of the Corporations Act.

Representative means, in relation to a party:

- (a) a Related Body Corporate;
- (b) a director, officer or employee of the party or any of the party's Related Bodies Corporate; or
- an adviser to the party or any of the party's Related Bodies Corporate. (c) where an "adviser" means, in relation to an entity, a financier, financial

© King & Wood Mallesons 8 Implementation Agreement

adviser, corporate adviser, legal adviser, or technical or other expert adviser or consultant who provides advisory services in a professional capacity and who has been engaged by that entity.

Reverse Break Fee means an amount equal to US\$150,000,000.

Schemes mean the Share Scheme and the Trust Schemes.

Scheme Booklet means, in respect of the Transaction and the Schemes and the Destapling, the information booklet to be approved by the Scheme Court and despatched to Westfield Securityholders which must:

- (a) include the Share Scheme and an explanatory statement complying with the requirements of the Corporations Act and notice(s) of meeting and proxy form(s):
- include the notices of meeting to be prepared by WAML for the Trust (b) Scheme Meetings;
- include the notice of meeting to be prepared by WCL for the (c) Westfield General Meeting; and
- (d) comply with the Corporations Act, Corporations Regulations, Regulatory Guide 60, Regulatory Guide 74, Guidance Note 15, the ASX Listing Rules and any other applicable law.

Scheme Consideration means the WAT Scheme Consideration, the Westfield Company Scheme Consideration and the WFDT Scheme Consideration.

Scheme Court means the Federal Court of Australia, Sydney Registry, unless the parties determine otherwise.

Scheme Meetings mean the Share Scheme Meeting and Trust Scheme Meetings.

Scheme Participants means each person who is a Westfield Securityholder at the Record Date.

Second Court Date means the day on which the Scheme Court:

- makes an order pursuant to section 411(4)(b) of the Corporations Act (a) approving the Share Scheme and also grants the Second Judicial Advice; or
- (b) declines to approve the Share Scheme and/or grant the Second Judicial Advice.

Second Judicial Advice means confirmation from the Scheme Court under section 63 of the Trustee Act 1919 (NSW) (Australia) that WAT and WFDT would be justified in acting upon the Trust Scheme Resolutions in doing all things and taking all necessary steps to put the Trust Schemes into effect.

Share Scheme means the scheme of arrangement under part 5.1 of the Corporations Act under which Unibail acquires all of the TH Newco Shares and WCL Shares from each Westfield Securityholder to give effect to the Transaction substantially in the form of Annexure B, together with any amendment or modification made pursuant to section 411(6) of the Corporations Act and agreed to by Unibail and Westfield.

Share Scheme Meeting means the meeting to be convened by the Scheme Court at which Westfield Securityholders will vote on the Share Scheme pursuant

to section 411(1) of the Corporations Act, and includes any adjournment of such meetina.

Share Scheme Resolution means the resolution of Westfield Securityholders to approve the Share Scheme for the purposes of section 411(4) of the Corporations Act.

Specified Event means an event, occurrence or matter that:

- (a) occurs after the date of this document;
- (b) occurs before the date of this document but is only announced or publicly disclosed after the date of this document; or
- will or is likely to occur after the date of this document and which has not (c) been publicly announced prior to the date of this document.

Stapling Deed means the stapling deed between WCL and WAML (as responsible entity for WAT and for WFDT) dated 30 June 2014.

Subsidiary has the meaning given in the Corporations Act, provided that an entity will also be taken to be a Subsidiary of another entity if it is Controlled by that entity and, without limitation:

- a trust may be a Subsidiary, for the purposes of which a unit or other (a) beneficial interest will be regarded as a share;
- an entity may be a Subsidiary of a trust if it would have been a (b) Subsidiary if that trust were a corporation; and
- an entity will also be deemed to be a Subsidiary of an entity if that entity (c) is required by the accounting standards to be consolidated with that entity.

Superior Proposal means a bona fide Competing Proposal received by Westfield and which the Westfield Board, acting in good faith, and after taking advice from its legal and financial advisers, determines is:

- (a) reasonably capable of being completed taking into account all aspects of the Competing Proposal, including its conditions; and
- of a higher financial value and is more favourable to the Westfield (b) Securityholders than the Schemes,

in each case, taking into account all aspects of the Competing Proposal. including the terms of the Competing Proposal, the price and/or value of the Competing Proposal, any conditions, timing considerations and any other matters affecting the probability of the Competing Proposal being completed in accordance with its terms, the identity, expertise, reputation and financial condition of the person making such proposal and legal, regulatory and financial matters.

Tax Act means the Income Tax Assessment Act 1936 (Cth) (Australia) or the Income Tax Assessment Act 1997 (Cth) (Australia), or both as the context requires.

Taxes means taxes, levies, imposts, charges and duties (including stamp and transaction duties) paid, payable or assessed as being payable by any authority together with any fines, penalties and interest in connection with them.

Taxing Authority means any Regulatory Authority responsible for the administration of any Taxes.

Tax Return means any returns, declarations, statements, claim for refund, election, estimate, reports, forms and information returns and any schedules or amendments thereto relating to Taxes.

TH Newco means a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) incorporated or to be incorporated in the Netherlands.

TH Newco Scheme Shares means the TH Newco Shares which are issued on the Implementation Date under the WFDT Trust Scheme.

TH Newco Share means an ordinary share in the share capital of TH Newco.

Third Party means a person other than Unibail and its Associates.

Third Party Nominee means an entity appointed by Unibail that will be issued TH Newco Scheme Shares.

Timetable means the timetable set out in Schedule 1.

Transaction means the combination of Unibail and Westfield through the acquisition of Westfield by Unibail as contemplated by this document, including the implementation of all of the Transaction Steps in accordance with this document, the Share Scheme, the Trust Schemes and the Westfield Constitutional Amendments (but excluding the OneMarket Demerger).

Transaction Steps means each of the steps set out in Schedule 2.

Treasurer means the Treasurer of the Commonwealth of Australia.

Trust Constitutions means the WAT Constitution and the WFDT Constitution.

Trust Property means the WAT Property and WFDT Property.

Trust Schemes means the arrangements under which TH Newco acquires all of the WAT Units and US NewCo acquires all of the WFDT Units from each Westfield Securityholder to give effect to the Transaction, facilitated by amendments to the WAT Constitution and the WFDT Constitution as set out in the Westfield Constitutional Amendments.

Trust Scheme Meetings means the meetings of Westfield Securityholders convened pursuant to the WAT Constitution and the WFDT Constitution to consider the Trust Scheme Resolutions, and includes any adjournment of those meetings.

Trust Scheme Resolutions means the resolutions of Westfield Securityholders to approve the Trust Schemes (including resolutions required under section 611 item 7 and section 601GC(1) of the Corporations Act) and the Destapling.

UK Bribery Act means the UK Bribery Act of 2010 (c.23), as amended, and any rules, regulations and guidance promulgated thereunder.

Unibail has the meaning given to it in the Details.

Unibail Board means the supervisory board (conseil de surveillance) of Unibail.

Unibail Change of Control Proposal means a bone fide proposal or offer (whether by way of public tender offer (offre publique d'acquisition), merger,

acquisition, sale or issue of securities, joint venture or otherwise) received by the Unibail Board from any person or group of persons which, if completed substantially in accordance with its terms, would:

- result in a person, whether alone or in concert (*de concert*) with other persons:
 - (i) directly or indirectly owning 30% or more of the share capital or voting rights of Unibail; or
 - directly or indirectly acquiring, obtaining a right to acquire, or otherwise obtaining an economic interest in, or control of, all or a substantial part of the assets of or business conducted by the Unibail Group; and
- (b) be determined by the Unibail Board to be consistent with its corporate interest (*intérêt social*) and in the best interest of Unibail, its shareholders, employees and other stakeholders.

Unibail Contribution Auditor Court Application means the application to be filed by Unibail to request the appointment of contribution auditors (*commissaire aux apports*) for the contribution of TH Newco Shares and WCL Shares, in accordance with Article L. 225-147 of the French commercial code (*code de commerce*).

Unibail Convening Notice means the convening notice (*avis de convocation*) of Unibail for purposes of convening the Unibail Shareholder Meeting.

Unibail Counterproposal has the meaning it is given in clause 10.9.

Unibail Debt Facility means a debt facility arrangement Disclosed to Westfield prior to the date of this document.

Unibail Director means a member of the Unibail Board.

Unibail Group means Unibail and its Subsidiaries.

Unibail Indemnified Parties means Unibail, its officers, employees and advisers, its Related Bodies Corporate and the officers, employees and advisers of each of its Related Bodies Corporate.

Unibail Information means:

- (a) the information regarding Unibail (other than information concerning or referrable to Westfield in relation to the combined Westfield Group and Unibail Group, but only to the extent prepared or provided by Westfield) as is required to be included in the Scheme Booklet under the Corporations Act, Corporations Regulations, Regulatory Guide 60, Regulatory Guide 74 and Guidance Note 15. Unibail Information does not include information about the Westfield Group (except to the extent it relates to any statement of intention relating to the Westfield Group following the Effective Date); and
- (b) with respect to the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting, all information contained in the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting, other than the Westfield Information.

& Wood Mallesons Implementation Agreement 12

Unibail Material Adverse Effect means a Specified Event which has, has had, or is reasonably likely to have, either individually or when aggregated with any Specified Events of a similar kind or category, the effect of:

- (a) the consolidated recurring earnings of the Unibail Group calculated using the same principles as were used to calculate consolidated earnings in the financial statements for the six months ended 30 June 2017 being reduced on a recurring basis in successive financial years by at least €70,000,000; or
- (b) the consolidated net asset value of the Unibail Group (taken as a whole) calculated using the same principles as were used to calculate consolidated net asset value in the financial statements for the six months ended 30 June 2017 being reduced by at least €900,000,000,

but does not include:

- any matter Disclosed (or which ought reasonably have been expected to (c) arise from a matter, event or circumstance which was so Disclosed);
- (d) any Specified Event arising from changes in general economic or political conditions, the securities market in general or law applying to entities generally, in the jurisdictions in which the Unibail Group operates;
- any change in interest rates or general economic conditions which (e) impact on Westfield and Unibail in a similar manner;
- any change in taxation or tax laws applicable to companies, trusts and (f) securityholders generally, in the jurisdictions in which the Unibail Group operates:
- (g) any change in generally accepted accounting principles applicable to Unibail or the interpretation of them;
- (h) any change occurring directly or indirectly as a result of any Specified Event required by this document, the Transaction, the Schemes or the transactions contemplated by them;
- the effect of any mark to market movements relating to financial (i) derivatives entered into in respect of interest bearing liabilities and foreign exchange rates (including any impacts of those movements on funds from operations);
- (j) the effect of fair value movements in derivative financial instruments used to manage interest and currency risks (including the impacts of those movements on consolidated earnings); or
- (k) any change occurring with the written consent of Westfield,

Regard must also be had to any other matter agreed by the parties in writing prior to the date of this document for the purpose of this definition.

Unibail Notice of Meeting means the notice of meeting (avis de réunion) of Unibail for purposes of convening the Unibail Shareholder Meeting.

Unibail Prescribed Event means any of the following events:

(conversion) Unibail converts all or any of its Unibail Shares into a (a) larger or smaller number of shares or securities (as applicable) or a resolution is passed to do so;

- (b) (reduction of capital) Unibail reduces (or resolves to reduce) its capital in any way, or reclassifies, combines, splits or redeems or repurchases directly or indirectly any of its securities:
- (repurchase) Unibail repurchases or buys-back, or agrees to or (c) approves the repurchase or buy-back, any of its Unibail Shares:
- (distribution) Unibail or any Unibail Group entity agrees to pay, makes (d) or declares, or announces an intention to make or declare or pay, any distribution (whether by way of dividend, capital reduction or otherwise and whether in cash or in specie) except:
 - (i) a distribution for the financial year ending 31 December 2017 (provided it is no more than €11 per Unibail Share);
 - (ii) a distribution by any Unibail Group entity where the recipient of that dividend or distribution is Unibail or a Wholly Owned Subsidiary of Unibail; or
 - (iii) distributions made to any joint venture partners in the ordinary course under a joint venture or partnership agreement to which a Unibail Group entity is a party and which is Disclosed or otherwise agreed to in writing by Westfield;
- (issuing or granting shares or options) any Unibail Group entity: (e)
 - (i) issues shares or securities (including interests in a trust);
 - (ii) grants an option over or to subscribe for its shares or securities (including interests in a trust); or
 - (iii) agrees to make such an issue or grant such an option,

in each case to a person that is not Unibail or a Wholly Owned Subsidiary of Unibail other than securities issued:

- (iv) pursuant to the vesting (including by way of any acceleration of vesting) of any rights under a director, officer or employee share or other incentive plan on issue as at the date of this document in accordance with the relevant terms of issue and Disclosed to Westfield:
- (v) the grant or issue of any employee options or rights under employee incentive arrangements, where such issue or grant is consistent with past practice, or pursuant to the vesting of such options or rights (including by way of any acceleration of vesting)
- (vi) to the extent required to comply with, and in accordance with, the terms of a joint venture agreement to which a Unibail Group entity is a party and which is Disclosed or otherwise agreed to in writing by Westfield or in the ordinary course of business for any other joint venture agreement, partnership agreement or shareholder's agreement to which Unibail is a party; or
- (vii) pursuant to any other agreement that is Disclosed or agreed to in writing by Westfield;
- (f) (suspension) Unibail Shares are delisted or are subject to suspension from quotation for 5 or more trading days, other than due to or as a result of, an action taken by Westfield or one of its Associates;

- (ceases business) Unibail or any of its material Subsidiaries ceases, or (g) threatens to cease, to carry on business:
- (h) (breach of financing arrangements) in respect of any financing arrangement, agreement or instrument a Unibail Group entity has with a person other than a Unibail Group entity:
 - an event of default, or potential event of default, occurs that is (i) not remedied in accordance with the relevant cure rights under the arrangement, agreement or instrument;
 - a Unibail Group entity allows an obligation to pay an amount to (ii) be accelerated other than to prevent an event referred to in paragraph (i) above from occurring; or
 - (iii) a Unibail Group entity takes any action (including incurring any new indebtedness or drawing down moneys) which leads to the credit rating of any Unibail Group entity being downgraded or placed on negative watch,

but only where such an action would have a material adverse impact on the Unibail Group (taken as a whole);

- (i) (securities or other instruments) any Unibail Group entity issues or agrees to issue securities or other instruments convertible into shares or securities (including interests in a trust) in each case to a person outside the Unibail Group other than as Disclosed;
- (constituent documents) Unibail adopts new constituent documents or (j) modifies or repeals its constituent documents or a provision of them;
- (Insolvency) any material Unibail Group entity becomes Insolvent, is (k) wound up or terminated: or
- (Deregistration) any material Unibail Group entity being deregistered as (I) a company, or being otherwise dissolved,

except, in the case of a Unibail Prescribed Event listed in items (a) to (g) and (i) to (j), as:

- (m) required or expressly permitted by this document, the Transaction or the Schemes:
- required by any applicable law (except where that requirement arises as (n) a result of an action by a Unibail Group entity between the date of this document and the Implementation Date);
- (o) Disclosed: or
- (p) agreed to in writing by Westfield.

Unibail Reports means, in relation to Unibail, any reference document (document de référence), release, report, schedule, form, statement or other documents filed with the AMF or the French Trade and Company Registry (Registre du commerce et des sociétés) including the Unibail Annual and Sustainable Development Report for the year ended 31 December 2016 and from the time of its release (except for the purpose of the definition of Disclosed), the Unibail Annual and Sustainable Development Report for the year ended 31 December 2017.

Unibail Shareholder Approval means the approval by the Unibail Shareholders. by the requisite affirmative vote as required under French law, of the transactions involving Unibail contemplated by the Transaction Steps, including the decisions in relation to:

- (a) the distribution of Newco BI Shares to Unibail Shareholders:
- (b) the:
 - (i) increase of the Unibail share capital without preferential subscription right by a contribution in kind of the Westfield Securityholders of their TH Newco Shares and WCL Shares;
 - (ii) the issuance of New Unibail Shares to the Westfield Securityholders; and
- (c) the stapling of the Newco BI Shares to the Unibail Shares,

each of these decisions subject to the condition that the Transaction Steps that should have been implemented prior to any of them being implemented.

Unibail Shareholder Meeting means the meeting of Unibail Shareholders to obtain the Unibail Shareholder Approval by a single majority vote or by a two third majority vote, as applicable, as required under French law depending on the nature of the decisions to be taken by the Unibail Shareholders, and includes any adjournment of that meeting.

Unibail Shareholders means the shareholders of Unibail.

Unibail Shares means a fully paid ordinary share in the share capital (capital social) of Unibail, with a nominal value of €5.

Unibail Specified Joint Venture Agreement means certain joint venture agreements Disclosed to Westfield prior to the date of this document (as agreed between the parties).

Unibail UES Works Council means the works council of the Unibail UES (unité économique et sociale) composed of Unibail Management, Espace Expansion and Unibail Marketing et Multimédia - U2M.

US Newco means a corporation incorporated or to be incorporated in the United States in accordance with the Transaction Steps.

VAT means (i) any tax imposed in compliance with the Council Directive of 28 November 2006 on the common system of value added tax (Directive 2006/112/EC) and (ii) any other tax of a similar nature, whether imposed in a member state of the European Union in substitution for, or levied in addition to. such tax referred to in paragraph (i) above, or imposed elsewhere (other than GST).

VAT Amount has the meaning given in clause 19.3(a).

WAML has the meaning given to it in the Details.

WAT has the meaning given to it in the Details.

WAT Constitution means the trust deed establishing WAT dated 28 March 1996, as amended.

WAT Property means all of the scheme property of WAT, including all WAML's rights, property and undertaking which are the subject of WAT:

- (a) of whatever kind and wherever situated; and
- whether present or future. (b)

WAT Scheme Consideration means the consideration payable for WAT Units in accordance with the WAT Trust Scheme.

WAT Scheme Unit means a WAT Unit on issue as at the Record Date.

WAT Trust Scheme means the arrangement under which US Newco acquires all of the WAT Scheme Units, facilitated by amendments to the WAT Constitution as set out in the Westfield Constitutional Amendments.

WAT Unit means a fully paid ordinary unit in WAT.

WCL has the meaning given to it in the Details.

WCL Scheme Share means a WCL Share on issue as at the Record Date.

WCL Share means a fully paid ordinary share in the capital of WCL.

Westfield means each of WCL, WAT and WFDT (and where applicable includes WAML in its capacity as responsible entity of WAT or WFDT, as applicable), or all of them as the context requires.

Westfield Board means the board of directors of each of WCL and WAML.

Westfield Budgets means the financial budgets for the financial year ending 31 December 2018, development project budgets and capital expenditure budgets for the Westfield Group as Disclosed and agreed in writing by Unibail and Westfield.

Westfield Company Scheme Consideration means the consideration payable for WCL Shares and TH Newco Shares in accordance with the Share Scheme

Westfield Constitutional Amendments means the supplemental deeds poll under which WAML will amend the:

- (a) WAT Constitution; and
- (b) WFDT Constitution.

for the purposes of implementing the Trust Schemes, substantially in the form attached as Annexure D (unless the parties agree otherwise).

Westfield Constitutions means the constitutions of each entity comprising Westfield.

Westfield Debt Facility means a debt facility arrangement Disclosed to Unibail prior to the date of this document.

Westfield Director means a director of WCL or WAML.

Westfield Disclosure Materials means:

the documents and information contained in the data room made (a) available by Westfield to Unibail and its Representatives, the index of

- which has been initialled by, or on behalf of, the parties for identification prior to the date of this document:
- (b) written responses from Westfield and its Representatives to requests for further information made by Unibail and its Representatives, a copy of which has been initialled by, or on behalf of, the parties for identification prior to the date of this document; and
- any other documents, information, responses or disclosures agreed in (c) writing by the parties to comprise the Westfield Disclosure Materials.

Westfield General Meeting means a meeting of the members of WCL to approve:

- the implementation of the Transaction for all purposes by ordinary (a) resolution; and
- the Destapling by way of a special resolution under and in accordance (b) with each of the Westfield Constitutions and the Stapling Deed,
- (c) and includes any adjournment of that meeting.

Westfield Group means Westfield and its Subsidiaries.

Westfield Indemnified Parties means Westfield, its officers, employees, and advisers and its Related Bodies Corporate and the officers, employees and advisers of each of its Related Bodies Corporate.

Westfield Information means:

- (a) with respect to the Scheme Booklet, all information contained in the Scheme Booklet other than the Unibail Information and the Independent Expert's Report; and
- (b) with respect to the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting, the information regarding Westfield as is required to be included in the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting (as the case may be) in all respects as required by applicable law.

Westfield Material Adverse Effect means a Specified Event which has, has had, or is reasonably likely to have, either individually or when aggregated with any Specified Events of a similar kind or category, the effect of:

- (a) the consolidated funds from operations of Westfield Group on a consolidated basis calculated using the same principles as were used to calculate funds from operations in the financial statements for the six months ended 30 June 2017 being reduced on a recurring basis in successive financial years by at least US\$35,000,000; or
- the consolidated net assets of the Westfield Group (taken as a whole) (b) calculated using the same principles as were used to calculate consolidated net assets in the financial statements for the six months ended 30 June 2017 being reduced by at least US\$500,000,000

but does not include:

(c) any matter Disclosed (or which ought reasonably have been expected to arise from a matter, event or circumstance which was so Disclosed);

- any Specified Event arising from changes in general economic or (d) political conditions, the securities market in general or law applying to entities generally, in the jurisdictions in which the Westfield Group operates:
- (e) any change in interest rates or general economic conditions which impact on Westfield and Unibail in a similar manner;
- any change in taxation or tax laws which apply to companies, trusts and (f) securityholders generally, in the jurisdictions in which the Westfield Group operates;
- (g) any change in generally accepted accounting principles applicable to Westfield or the interpretation of them;
- (h) any change occurring directly or indirectly as a result of any Specified Event required by this document, the Transaction, the Schemes or the transactions contemplated by them;
- (i) the effect of any mark to market movements relating to financial derivatives entered into in respect of interest bearing liabilities and foreign exchange rates (including any impacts of those movements on funds from operations);
- the effect of fair value movements in derivative financial instruments (j) used to manage interest and currency risks (including the impacts of those movements on funds from operations); or
- (k) any change occurring with the written consent of Unibail.

A fixed US\$/UK£ exchange rate of 1.3004 (spot) and 1.3256 (average) will be used (as applicable) and no regard will be had to movements in US\$/UK£ exchange rates or in the foreign currency translation reserve,

Regard must also be had to any other matter agreed by the parties in writing prior to the date of this document for the purpose of this definition.

Westfield Prescribed Event means any of the following events:

- (a) (conversion) Westfield converts all or any of its Westfield Stapled Securities into a larger or smaller number of shares or units (as applicable) or a resolution is passed to do so;
- (reduction of capital) Westfield reduces (or resolves to reduce) its (b) capital in any way, or reclassifies, combines, splits or redeems or repurchases directly or indirectly any of its securities;
- (buy-back) Westfield: (c)
 - (i) enters into a buy-back agreement;
 - (ii) resolves to approve the terms of a buy-back agreement under the Corporations Act; or
 - buys back Westfield Stapled Securities on-market; (iii)
- (d) (distribution) Westfield or any Westfield Group entity agrees to pay, makes or declares, or announces an intention to make or declare or pay, any distribution (whether by way of trust distribution, dividend, capital reduction or otherwise and whether in cash or in specie) other than:

- a distribution of no more than \$0.255 per Westfield Stapled (i) Securities for the financial year ending 31 December 2017;
- a distribution by any Westfield Group entity where the recipient (ii) of that distribution is Westfield or a Wholly Owned Subsidiary of Westfield; or
- (iii) distributions made to any joint venture partners in the ordinary course under a joint venture or partnership agreement to which a Westfield Group entity is a party and which is Disclosed to Unibail prior to the date of this document or otherwise agreed to in writing by Unibail.
- (e) (issuing or granting shares, units or options) any Westfield Group entity:
 - (i) issues shares or securities (including interests in trusts);
 - (ii) grants an option over or to subscribe for its shares or securities (including interests in trusts); or
 - (iii) agrees to make such an issue or grant such an option,

in each case to a person that is not Westfield or a Wholly Owned Subsidiary of Westfield other than:

- pursuant to the vesting of any Employee Rights: (A) Disclosed to (iv) Unibail and on issue as at the date of this document in accordance with the terms of issue or clause 4.5 of this document; or (B) otherwise agreed to in writing by Unibail;
- (v) pursuant to the exercise of options in Westfield Inc or the conversion of convertible securities in Westfield L.P. in accordance with the terms of issue which have been Disclosed to Unibail prior to the date of this document or otherwise agreed to in writing by Unibail;
- (vi) to the extent required to comply with, and in accordance with, the terms of a Westfield Specified Joint Venture Agreement or in the ordinary course of business for any other joint venture agreement, partnership agreement or shareholder's agreement to which Westfield is a party; or
- pursuant to any other agreement Disclosed to Unibail prior to the (vii) date of this document and agreed to in writing by Unibail.
- (f) (securities or other instruments) any Westfield Group entity issues or agrees to issue securities or other instruments convertible into shares or securities (including interests in trusts) in each case to a person that is not Westfield or a Wholly Owned Subsidiary of Westfield:
- (g) (constitution) Westfield adopts a new constitution or modifies or repeals its constitution or a provision of it;
- (h) (Trust) any of the following occurs:
 - Westfield effects or facilitates the retirement, removal or (i) replacement of WAML as trustee or responsible entity of WAT and/or WFDT, or a meeting is convened in response to a notice (other than a notice signed by or on behalf of Unibail) received by WAML requisitioning a meeting of Westfield Securityholders

- to replace WAML as trustee or responsible entity of WAT and/or WFDT:
- (ii) an application is made in any court for the appointment of a temporary responsible entity of WAT and/or WFDT in accordance with the Corporations Act:
- Westfield does or omits to do anything that restricts WAML's (iii) rights of indemnity from the Trust Property in respect of obligations incurred by WAML; or
- Westfield effects or facilitates the resettlement of the Trust (iv) Property;
- (i) (breach of financing arrangements) in respect of any financing arrangement, agreement or instrument a Westfield Group entity has with a person other than a Westfield Group entity:
 - an event of default, or potential event of default, occurs that is (i) not remedied in accordance with the relevant cure rights under the arrangement, agreement or instrument;
 - (ii) a Westfield Group entity allows an obligation to pay an amount to be accelerated other than to prevent an event referred to in paragraph (i) above from occurring; or
 - (iii) a Westfield Group entity takes any action (including incurring any new indebtedness or drawing down moneys) which leads to the credit rating of any Westfield Group entity being downgraded or placed on negative watch,

but only where such an action would have a material adverse impact on the Westfield Group (taken as a whole) excluding any circumstances relating to possible change of control matters that have been Disclosed;

- (suspension) Westfield Securities are delisted or are subject to (j) suspension from quotation for 5 or more trading days, other than due to, or as a result of, an action taken by Unibail or one of its Associates;
- (k) (No Unibail securities) Westfield or any of its Associates acquires a Relevant Interest in Unibail Shares or becomes a party to any agreement, arrangement or understanding the economic effect of which is to confer rights equivalent to, or substantially equivalent to, the acquisition, holding or disposal of Unibail Shares or a Relevant Interest in Unibail Shares (including any cash settled equity swap or similar derivative relating to any Unibail Shares);
- (I) (Division 6C) WAML approves or takes any action or makes any investment that could reasonably result in WAT and/or WFDT commencing to carry on a trading business within the meaning of Division 6C of the Tax Act or controlling or having the ability to control. directly or indirectly the affairs or operations of another person in respect of the carrying on by that person of a trading business within the meaning of that Division;
- (m) (ceases business) Westfield or any of its material Subsidiaries ceases, or threatens to cease, to carry on business;
- (Insolvency) any material Westfield Group entity becomes Insolvent, is (n) wound up or terminated;

© King & Wood Mallesons 21 Implementation Agreement

(Deregistration) any material Westfield Group entity becomes, or takes (o) steps to become, deregistered as a company, registered managed investment scheme, or being otherwise dissolved,

except in the case of a Westfield Prescribed Event listed in items (a) to (h) and (j) to (m) as:

- (p) required or expressly permitted by this document, the Transaction or the Schemes:
- (q) required by any applicable law (except where that requirement arises as a result of an action by a Westfield Group entity between the date of this document and the Implementation Date);
- required for purposes of implementing the OneMarket Demerger and (r) agreed to in writing by Unibail;
- Disclosed; or (s)
- (t) agreed to in writing by Unibail.

Westfield Register means the register of Westfield Stapled Securities and Registry has a corresponding meaning.

Westfield Securityholder means each person registered in the Westfield Register as the holder of Westfield Stapled Securities.

Westfield Significant Contract means a contract or commitment (excluding a contract or commitment contemplated by clause 8.2(b)):

- (a) the consideration under, or the value of which, exceeds \$10,000,000 (or \$20,000,000 in the case of a building or construction contract);
- (b) that increases the aggregate liabilities or off balance sheet commitments of the Westfield Group by an amount exceeding \$10,000,000 (or \$20,000,000 in the case of a building or construction contract);
- (c) that generates, or is expected to generate, annual revenues or gross annual expenditure in excess of \$10,000,000 per annum (or \$20,000,000 in the case of a building or construction contract);
- (d) Westfield Specified Joint Venture Agreement; or
- (e) any joint venture agreement, partnership agreement or shareholders' agreement entered into after the date of this document that is material to Westfield Group's business as a whole.

other than entry of leases in the ordinary course of business, where the net annual revenue associated with the lease is less than \$10,000,000 and where associated leasing capital incentives are less than \$5,000,000.

Westfield Specified Joint Venture Agreement means certain joint venture agreements Disclosed to Unibail prior to the date of this document.

Westfield Stapled Security means a WCL Share, a WAT Unit and a WFDT Unit stapled to each other in accordance with the provisions of the Stapling Deed and the Westfield Constitutions.

WFDT has the meaning given to it in the Details.

WFDT Constitution means the trust deed establishing WFDT dated 26 March 2014, as amended.

WFDT Property means all of the scheme property of WFDT, including all WAML's rights, property and undertaking which are the subject of WFDT:

- (a) of whatever kind and wherever situated; and
- (b) whether present or future.

WFDT Scheme Consideration means the consideration payable for TH Newco Shares in accordance with the Trust Scheme in respect of WFDT.

WFDT Scheme Unit means a WFDT Unit on issue as at the Record Date.

WFDT Trust Scheme means the arrangement under which TH Newco acquires all of the WFDT Scheme Units, facilitated by amendments to the WFDT Constitution as set out in the Westfield Constitutional Amendments.

WFDT Unit means a fully paid ordinary unit in WFDT.

Wholly Owned-Subsidiary means, in relation to a party, a Subsidiary, all of the issued shares or trust units of which are or will be directly or indirectly owned by that party.

1.2 **General interpretation**

Headings and labels used for definitions are for convenience only and do not affect interpretation. Unless the contrary intention appears, in this document:

- the singular includes the plural and vice versa; (a)
- (b) a reference to a document includes any agreement or other legally enforceable arrangement created by it (whether the document is in the form of an agreement, deed or otherwise);
- a reference to a document also includes any variation, replacement or (c) novation of it;
- (d) the meaning of general words is not limited by specific examples introduced by "including", "for example", "such as" or similar expressions;
- (e) a reference to "person" includes an individual, a body corporate, a partnership, a joint venture, an unincorporated association and an authority or any other entity or organisation;
- a reference to a particular person includes the person's executors. (f) administrators, successors, substitutes (including persons taking by novation) and assigns:
- a reference to a time of day is a reference to Sydney, Australia time; (g)
- (h) where an act must take place on a day or a Business Day it must occur at a time of day that is both part of a day of the same date in Sydney, Australia and part of the same day of the same date in Paris, France;
- (i) a reference to:
 - (i) \$, US\$ or dollars is a reference to the currency of the United States of America;

- (ii) A\$ is a reference to the currency of Australia; and
- euro or € is a reference to the currency of the European Union; (iii)
- a reference to "law" includes common law, civil law, principles of equity (j) and legislation (including regulations):
- a reference to any legislation includes regulations under it and any (k) consolidations, amendments, re-enactments or replacements of any of them:
- (I) a reference to "regulations" includes instruments of a legislative character under legislation (such as regulations, rules, by-laws, ordinances and proclamations);
- (m) a reference to a group of persons is a reference to any 2 or more of them jointly and to each of them individually; and
- (n) a reference to anything (including an amount) is a reference to the whole and each part of it.

2 Agreement to propose Transaction

2.1 **Contemplated Transaction**

Subject to the satisfaction of the condition precedent to formation set out in clause 2.2, the parties agree to implement the Transaction on the terms and subject to the conditions of this document.

ERB consultation 2.2

- Except for the Preliminary Provisions which are binding on the parties (a) from execution of this document, all other provisions of this document are conditional and will not become binding on the parties until the Consultation Process in relation to the Contemplated Transaction has been completed or deemed to have been completed and the Consideration Period has concluded without this document having been terminated in accordance with clause 2.2(g). If this document has not been terminated in accordance with clause 2.2(g) all provisions of this document will become binding on the parties on the Business Day following the end of the Consideration Period.
- (b) Unibail must formally initiate the Consultation Process no later than 2 Business Days following the date of this document.
- Until the end of the Additional Issuance Period: (c)
 - (i) Each party must use its best endeavours to progress and complete the Consultation Process as soon as practicable following the initiation of the Consultation Process.
 - (ii) The objective of the parties is that the Unibail UES Works Council opinion shall be issued by 15 January 2018 and the parties must use their best endeavours to complete that Consultation Process as soon as practicable having regard to that timeframe.
 - If the EEC's ERB Opinion has not been issued within 2 months (iii) of the date for the initiation of the Consultation Process in accordance with clause 2.2(b), Unibail must notify the EEC and

- give the EEC a further period of 7 days to issue its opinion pursuant to and in accordance with the last paragraph of Article 10 of the EEC Agreement ("Additional Issuance Period").
- (iv) Westfield agrees to cooperate, as may be required, with Unibail, including by providing in a timely manner, within reasonable limits and according to standard practice in similar transactions. any documents and information relating to Westfield that may be reasonably requested by Unibail or by any ERB in connection with the Consultation Process, to the extent such documents and information are reasonably available. Westfield must use reasonable efforts to provide answers to any questions raised by any ERB as part of the Consultation Process.
- (v) Unibail must keep Westfield regularly informed of the progress of and any relevant developments in the Consultation Process. Unibail must provide Westfield with details and copies of any correspondence or communications in relation to the Consultation Process provided by Unibail to an ERB or any expert appointed by an ERB (or, conversely, of any correspondence or communications in relation to the Consultation Process provided to Unibail by an ERB or any expert appointed by an ERB) and consult with Westfield as to any such correspondence or communications prior to responding to such correspondence or communications.
- (vi) The parties acknowledge that experts may be appointed by an ERB and each party must use its reasonable efforts to cooperate with such expert.
- (vii) Unibail must not and must cause its Representatives not to make any commitment to an ERB, or any member of any ERB with respect to modifications to any term or condition of the Contemplated Transaction without the prior written consent of Westfield.
- (d) Subject to any court decision resulting in the extension of the Consultation Period, the Consultation Process will be completed or deemed to have been completed in relation to the Contemplated Transaction on the date on which:
 - (i) an ERB has issued its respective opinion (whether positive, neutral or negative) with respect to the Contemplated Transaction which may be rendered orally or in writing or by any other action or omission (including the passage of time) which, under applicable law or regulation, would cause such ERB to be deemed to have rendered such opinion (the "ERB Opinion"); or
 - (ii) if the EEC has not issued its opinion, the Additional Issuance Period has expired.
- (e) Unibail must promptly provide Westfield with full details of any ERB Opinion including a copy of any written ERB Opinion, and in the case of an oral ERB Opinion a formal declaration (déclaration sur l'honneur) detailing the full contents of the oral ERB Opinion.
- (f) During the Consideration Period, the parties will give due consideration to any recommendations made and the opinions rendered by each ERB concerning the Contemplated Transaction, if any, and, as the case may be:

- (i) adopt changes to this document; or
- (ii) make any commitment,

in each case, as the parties may mutually agree in writing in response to such recommendations and opinions. Unibail and Westfield acknowledge and agree that neither of them is under any obligation to adopt any change to this document or make any commitment in order to address any recommendations and opinions made by any ERB concerning the Contemplated Transaction.

If Unibail has received, within the Consultation Period, an ERB Opinion (g) in opposition to the Contemplated Transaction, Unibail will have the right to terminate this document by notifying Westfield in writing of that termination prior to the end of the Consideration Period, in which case this document (other than the Preserved Clauses) will be void ab initio and of no legal effect.

3 **Conditions Precedent**

3.1 **Conditions Precedent**

Subject to this clause 3, the Schemes and the Transaction will not become Effective until each of the following Conditions Precedent are satisfied or waived to the extent and in the manner set out in this clause.

Con	dition Precedent	Party entitled to benefit	Party responsible
(a)	(FIRB approval) before 8.00am on the Second Court Date either:	Cannot be waived	Unibail
	 (i) the Treasurer (or the Treasurer's delegate) has provided a written no objection notification to the Transaction either without conditions or with conditions accepted by Unibail acting reasonably (and it would not be reasonable for Unibail to not accept standard tax conditions that are in the form of Attachment A of FIRB Guidance Note 47 as at 24 November 2016); or (ii) following notice of the proposed Transaction having been given by Unibail to the Treasurer under the FIRB Act, the Treasurer has ceased to be empowered to make any order under Part 3 of the FIRB Act because the applicable time limit on making orders and decisions under the FIRB Act has expired. 		
(b)	(ASIC and ASX) before 8.00am on the Second Court Date, ASIC and ASX issue or provide any consents or approvals, or have done any other acts, which the parties agree are reasonably necessary or desirable to	Both	Both

Con	ndition F	Precedent	Party entitled to benefit	Party responsible
	implem	nent the Schemes, including:		
	(i)	ASIC having granted a modification of item 7 of section 611 of the Corporations Act, allowing each Scheme Participant (other than those excluded from voting) to vote in favour of the Trust Scheme Resolutions for the purpose of item 7 of section 611 or indicated in writing that modification will not be required;		
	(ii)	ASIC has granted an exemption in favour of Westfield and Unibail (and their Controlled Entities) under Division 2 of Part 7.7 of the Corporations Act from the requirement to provide a financial services guide in connection with the Scheme Booklet; and		
	(iii)	ASX has confirmed that it does not object to the proposed amendments to the WAT Constitution, the WFDT Constitution or the draft Scheme Booklet under ASX Listing Rule 15.1,		
		ose consents, approvals or other acts ot been withdrawn or revoked at that		
(c)	Westfie Schem Meetin the Tru (at the requisi	rield Securityholder approval) the eld Securityholders approve the Share are Resolution (at the Share Scheme g), the Trust Scheme Resolutions (at last Scheme Meeting) and the Destapling Westfield General Meeting) by the te majorities in accordance with the reations Act.	Cannot be waived	Westfield
(d)	5.00pm	nil Shareholder Approval) on or before in on the day prior to the Scheme gs, the Unibail Shareholder Approval is ed.	Cannot be waived	Unibail
(e)	Approprior to	Unibail Shares Listing Prospectus val) on or before 5.00pm on the day of the Scheme Meetings, the AMF has red the New Unibail Shares Listing ectus.	Cannot be waived	Unibail

Condition Precedent		Party entitled to benefit	Party responsible
(f)	(New Unibail Stapled Securities Listing Prospectus Approval) on or before 5.00pm on the day prior to the Scheme Meetings, the AMF and/or the AFM, as applicable, have approved the New Unibail Stapled Securities Listing Prospectus.	Cannot be waived	Unibail
(g)	(Court approval) the Scheme Court approves the Share Scheme in accordance with section 411(4)(b) of the Corporations Act and provides the First Judicial Advice and Second Judicial Advice in respect of the Trust Schemes.	Cannot be waived	Westfield
(h)	(Regulatory intervention) no Court or Regulatory Authority has issued or taken steps to issue an order, temporary restraining order, preliminary or permanent injunction, decree or ruling or taken any action enjoining, restraining or otherwise imposing a legal restraint or prohibition preventing the Schemes and no such order, decree, ruling, other action or refusal is in effect as at 8.00am on the Second Court Date.	Both	Both
(i)	(Independent Expert) the Independent Expert issues a report which concludes that the Schemes are in the best interests of, Scheme Participants before the date on which the Scheme Booklet is lodged with ASIC and that expert does not change its opinion or otherwise withdraw the report before 8:00am on the Second Court Date.	Westfield	Westfield
(j)	(No Westfield Prescribed Event) no Westfield Prescribed Event occurs between the date of this document and 8.00am on the Second Court Date.	Unibail	Westfield
(k)	(No Westfield Material Adverse Effect) no Westfield Material Adverse Effect occurs between the date of this document and 8.00am on the Second Court Date.	Unibail	Westfield
(1)	(No Unibail Prescribed Event) no Unibail Prescribed Event occurs between the date of this document and 8.00am on the Second Court Date.	Westfield	Unibail
(m)	(No Unibail Material Adverse Effect) no Unibail Material Adverse Effect occurs between the date of this document and 8.00am on the Second Court Date.	Westfield	Unibail
(n)	(New Unibail CDIs and New Unibail Stapled Securities) the following occurs:	Both	Unibail
	(i) New Unibail Shares to be issued		

28

Condition Precedent	Party entitled to benefit	Party responsible
pursuant to the Schemes and the Transactions have been approved by Euronext Paris and Euronext Amsterdam, as applicable and New Unibail Stapled Securities have obtained listing and trading approval from each of Euronext Paris and Euronext Amsterdam subject to customary conditions by, and such approvals remain in full force and effect in all respects at, 8:00am on the Second Court Date; and (ii) Unibail and Newco BI are approved for admission to the official list of ASX (as a foreign exempt listing) and the New Unibail CDIs to be issued or transferred pursuant to the Schemes are approved for official quotation by ASX, subject to customary conditions by, and such approvals and consents have not been withdrawn, suspended or revoked before 8:00am on the		

3.2 Reasonable endeavours

Each of Westfield and Unibail agree to use reasonable endeavours to procure that:

- (a) each of the Conditions Precedent for which it is a party responsible (as noted in clause 3.1):
 - (i) is satisfied as soon as practicable after the date of this document (including by offering any reasonable undertakings or commitments necessary to satisfy the relevant Condition Precedent); and
 - continues to be satisfied at all times until the last time it is to be (ii) satisfied (as the case may require); and
- there is no occurrence that would prevent the Condition Precedent for (b) which it is a party responsible being satisfied.

3.3 Regulatory matters

Without limiting clause 3.2, each party:

- (applying for Regulatory Approvals) must promptly apply for all (a) relevant Regulatory Approvals and provide each other party with a copy of those applications;
- (Regulatory Approvals process) must take all steps it is responsible (b) for as part of the Regulatory Approval process, including responding to requests for information at the earliest practicable time;

- (representation) has the right to be represented and make submissions (c) at any meeting with any Regulatory Authority relating to a Regulatory Approval (but only to the extent permitted by the Regulatory Authority):
- (consultation) must consult with the other party in advance in relation to (d) all communications (whether written or oral, and whether direct or via a Representative) with any Regulatory Authority relating to any Regulatory Approval and:
 - (i) provide the other party with drafts of any material written communications to be sent to a Regulatory Authority and make any amendments as the other party reasonably requires; and
 - (ii) provide copies of any material written communications sent to or received from a Regulatory Authority to the other party promptly upon despatch or receipt (as the case may be),

in each case to the extent it is reasonable to do so and provided that:

- (iii) a party may withhold or redact information or documents from the other parties if and to the extent that they are either confidential to a third party, or commercially sensitive and confidential to the applicant or subject to legal professional privilege in favour of that party;
- (iv) no party is required to disclose materially commercially sensitive information to any other party; and
- the party applying for a Regulatory Approval is not prevented (v) from taking any step (including communicating with a Regulatory Authority) in respect of a Regulatory Approval if the other party has unduly delayed responding under clause 3.3(d) and has been notified of same.

3.4 **Waiver of Conditions Precedent**

- (a) A Condition Precedent may only be waived in writing by the party or parties entitled to the benefit of that Condition Precedent as noted in clause 3.1 and will be effective only to the extent specifically set out in that waiver.
- (b) A party entitled to waive the breach or non-fulfilment of a Condition Precedent under this clause 3.4 may do so in its absolute discretion.
- (c) If either Westfield or Unibail waives the breach or non-fulfilment of a Condition Precedent in accordance with this clause 3.4. then:
 - (i) subject to clause 3.4(c)(ii), that waiver precludes that party from suing the other for any breach of this document arising as a result of the breach or non-fulfilment of that Condition Precedent or arising from the same event which gave rise to the breach or non-fulfilment of that Condition Precedent; but
 - if the waiver of the Condition Precedent is itself conditional and (ii) the other party:
 - (A) accepts the condition, the terms of that condition apply notwithstanding any inconsistency with clause 3.4(c)(i); or

© King & Wood Mallesons 30 Implementation Agreement

- (B) does not accept the condition, the Condition Precedent has not been waived.
- (d) A waiver of a breach or non-fulfilment in respect of a Condition Precedent does not constitute:
 - a waiver of a breach or non-fulfilment of any other Condition (i) Precedent arising from the same event; or
 - a waiver of a breach or non-fulfilment of that Condition (ii) Precedent resulting from any other event.

Notices in relation to Conditions Precedent 3.5

Each party must:

- (a) (notice of satisfaction) promptly notify the other of satisfaction of a Condition Precedent and must keep the other informed of any material development of which it becomes aware that may lead to the breach or non-fulfilment of a Condition Precedent;
- (notice of failure) immediately give written notice to the other of a (b) breach or non-fulfilment of a Condition Precedent, or of any event which will prevent a Condition Precedent being satisfied; and
- (notice of waiver) upon receipt of a notice given under clause 3.5(b), (c) give written notice to the other party as soon as possible (and in any event before 5.00pm on the Business Day before the Second Court Date) as to whether or not it waives the breach or non-fulfilment of any Condition Precedent resulting from the occurrence of that event (if such waiver is possible from that party), specifying the Condition Precedent in question.

Consultation on failure of Condition Precedent 3.6

If:

- there is a breach or non-fulfilment of a Condition Precedent which is not (a) waived in accordance with this document by the time or date specified in this document for the satisfaction of the Condition Precedent;
- there is an act, failure to act or occurrence which will prevent a Condition (b) Precedent being satisfied by the time or date specified in this document for the satisfaction of the Condition Precedent (and the breach or nonfulfilment which would otherwise occur has not already been waived in accordance with this document); or
- (c) the Schemes have not become Effective or the Unibail Shareholder Approval has not been obtained by the End Date,

the parties must consult in good faith with a view to determine whether:

- (d) the Transaction can proceed by way of alternative means or methods;
- (e) to extend the relevant time for satisfaction of the Condition Precedent or to adjourn or change the date of an application to the Scheme Court; or
- (f) to extend the End Date.

3.7 Failure to agree

- If the parties are unable to reach agreement under clause 3.6 within 10 **Business Days:**
 - (i) subject to clause 3.7(a)(ii), either party may terminate this document (and that termination will be in accordance with clause 14.1(g)(i)); or
 - if a Condition Precedent may be waived and exists for the (ii) benefit of one party only, that party only may waive that Condition Precedent or terminate this document (and that termination will be in accordance with clause 14.1(g)(ii)).
- (b) A party will not be entitled to terminate this document under this clause if the relevant Condition Precedent has not been satisfied or agreement cannot be reached as a result of a breach of this document by that party or a deliberate act or omission of that party.

4 **Outline of Schemes**

4.1 **Outline of Schemes**

Subject to the satisfaction or waiver of the Conditions Precedent on or before the Implementation Date:

- the Transaction Steps will be implemented, which will result in all of the (a) components of the Westfield Stapled Securities held by the Westfield Securityholders being transferred to New Unibail or its Subsidiaries (including US Newco); and
- (b) each Scheme Participant will be entitled to receive the Scheme Consideration in accordance with the terms of this document, the Share Scheme and the Westfield Constitutional Amendments.

Relationship between Share Scheme and Trust Schemes, and the 4.2 Schemes and the Destapling

- (a) The Share Scheme and Trust Schemes are interconditional so that the Share Scheme and each of the Trust Schemes must become Effective for the Schemes to proceed.
- (b) The Destapling and the Schemes are interconditional such that the Destapling may only occur if the Schemes are implemented and the Schemes may only be implemented if the Destapling occurs.

4.3 **Scheme Consideration**

- The Scheme Consideration to be provided in respect of each Westfield (a) Stapled Security is as described in the Share Scheme and as will be described in the Westfield Constitutional Amendments.
- Subject to clause 3.1, each Scheme Participant is entitled to receive the (b) Scheme Consideration in respect of each Westfield Stapled Security held by that Scheme Participant, in accordance with the terms of this document, the Share Scheme and the Westfield Constitutional Amendments.

4.4 **Provision of Westfield Register**

In order to facilitate the provision of the Scheme Consideration, Westfield must provide, or procure the provision of, to Unibail or a nominee of Unibail a complete copy of the Westfield Register as at the Record Date (which must include the name, registered address and registered holding of each Scheme Participant as at the Record Date), within one Business Day after the Record Date.

4.5 **Employee incentives**

- (a) Westfield must ensure that, by no later than the day following the Effective Date, there are no outstanding Employee Rights.
- (b) In order to comply with its obligation under clause 4.5(a) and except as agreed in writing between the parties, Westfield must:
 - (i) cause some or all of the outstanding Employee Rights to vest on or after the Effective Date and, following such vesting and to the extent that the Employee Rights will not be cash settled, cause the relevant number of Westfield Stapled Securities to be transferred or issued (as applicable) to the relevant holders of Employee Rights in sufficient time to allow the relevant holders of the Employee Rights to participate in the Schemes; and
 - (ii) take such action as may be necessary to cancel any outstanding Employee Rights which it does not cause to vest in accordance with clause 4.5(b)(i) (if any).

No amendment to the Schemes without consent 4.6

Westfield must not consent to any modification of, or amendment to, or the making or imposition by the Scheme Court of any condition in respect of, the Schemes without the prior written consent of Unibail (such consent not to be unreasonably withheld or delayed except to the extent such modification, amendment or condition results in a modification of the Transaction Steps or the terms of the Transaction).

5 **Implementation**

5.1 **General obligations**

Westfield and Unibail must each:

- (a) use all reasonable endeavours and commit necessary resources (including management and corporate relations resources and the resources of external advisers); and
- (b) procure that its officers and advisers work in good faith and in a timely and co-operative fashion with the other party (including by attending meetings and by providing information),

to implement the Transaction and the Schemes as soon as reasonably practicable and in accordance with the Timetable.

5.2 Westfield's preliminary obligations

Westfield must:

- (announce directors' recommendation) following execution of this (a) document, announce, in the form contained in Annexure A (on the basis of statements made to Westfield by each Westfield Director) that:
 - the Westfield Directors intend to unanimously recommend to (i) Scheme Participants that the Transaction and the Schemes be approved; and
 - (ii) each Westfield Director who holds or controls the voting rights attached to Westfield Stapled Securities, intends to vote his or her Westfield Stapled Securities in favour of the Transaction and the Schemes,

subject to:

- the Independent Expert concluding, and continuing to conclude, (iii) that the Transaction and the Schemes are in the best interests of the Westfield Securityholders; and
- (iv) there being no Superior Proposal,

and Westfield must procure that any other material public statements or documents made or released by Westfield in relation to the Schemes or the Transaction includes a statement by Westfield Directors to that effect;

- (b) (preparation of Scheme Booklet) as soon as practicable after the date of this document, prepare the Scheme Booklet:
 - (i) in accordance with all applicable laws, including the Corporations Act, Corporations Regulations, Regulatory Guide 60, Regulatory Guide 74, Guidance Note 15 and the ASX Listing Rules: and
 - which includes a statement by the Westfield Directors: (ii)
 - (A) unanimously recommending that the Westfield Securityholders vote in favour of the Transaction and the Schemes subject to the Independent Expert continuing to conclude that the Transaction and the Schemes are in the best interests of the Westfield Securityholders and there being no Superior Proposal;
 - (B) that each Westfield Director who holds or controls the voting rights attached to Westfield Stapled Securities intends to vote his or her Westfield Stapled Securities in favour of the Transaction and the Schemes subject to the Independent Expert continuing to conclude that the Transaction and the Schemes are in the best interests of the Westfield Securityholders and there being no Superior Proposal:
- (c) (Independent Expert) promptly appoint the Independent Expert and provide any assistance and information reasonably requested by the Independent Expert to enable the Independent Expert to prepare its report for the Scheme Booklet as soon as practicable;
- (consultation with Unibail) consult with Unibail as to the content and (d) presentation of:

© King & Wood Mallesons 34 Implementation Agreement

- (i) the Scheme Booklet, which includes:
 - (A) allowing Unibail a reasonable opportunity to review and make comments on successive drafts of the Scheme Booklet (accepting that any review of the Independent Expert's Report is limited to review for factual accuracy of those parts that include information relating to Unibail):
 - (B) taking any reasonable comments made by Unibail into account in good faith when producing a revised draft of the Scheme Booklet;
 - (C) providing to Unibail a revised draft of the Scheme Booklet within a reasonable time before the Regulator's Draft is provided to ASIC for approval pursuant to section 411(2) of the Corporations Act is finalised; and
 - (D) obtaining Unibail's consent to the inclusion of the Unibail Information (including in respect of the form and context in which the Unibail Information appears in the Scheme Booklet); and
- (ii) documents required for the purposes of the Scheme Court hearings to be held in relation to the Schemes (including originating processes, affidavits, submissions and draft minutes of Scheme Court orders), and consider in good faith any comments on, or suggested amendments to, those documents from Unibail prior to filing those documents with the Scheme Court; and

(lodgement of Regulator's Draft) (e)

- (i) no later than 14 days before the First Court Date, provide the Regulator's Draft to ASIC for its review for the purposes of section 411(2) of the Corporations Act and to ASX for its review. and provide a copy of the Regulator's Draft to Unibail immediately thereafter; and
- keep Unibail fully informed of the progress of and any material (ii) issues raised by ASIC or ASX in relation to the Regulator's Draft and:
 - (A) provide Unibail with details and copies of any correspondence or communications provided by Unibail to ASIC or the ASX (or, conversely, of any correspondence or communications provided to Westfield by ASIC or the ASX) and consult with Unibail as to any such correspondence or communications prior to responding to such correspondence or communications; and
 - (B) where practical to do so, consult with Unibail in good faith prior to taking any steps or actions to address those material issues (provided that, where those issues relate to Unibail Information, Westfield must not take any steps to address them without Unibail's prior written consent, such consent not to be unreasonably withheld).

5.3 **Westfield's post-Consideration Period obligations**

Following the Consideration Period Westfield must:

- (a) (Court application) apply to the Scheme Court for:
 - an order under section 411(1) of the Corporations Act directing Westfield to convene the Share Scheme Meeting; and
 - (ii) the First Judicial Advice:
- (send Scheme Booklet) send the Scheme Booklet to the Westfield (b) Securityholders as soon as practicable after the Scheme Court orders Westfield to convene the Scheme Meetings:
- (supplementary disclosure) if, after despatch of the Scheme Booklet, (c) Westfield becomes aware:
 - that information included in the Scheme Booklet is or has (i) become misleading or deceptive in any material respect (whether by omission or otherwise); or
 - (ii) of information that is required to be disclosed to the Westfield Securityholders under any applicable law but was not included in the Scheme Booklet.

promptly consult with Unibail in good faith as to the need for, and the form of, any supplementary disclosure to the Westfield Securityholders, and make any disclosure that Westfield considers reasonably necessary in the circumstances, having regard to applicable laws and to ensure that there would be no breach of clause 13.1(I) if it applied as at the date that information arose:

- (d) (Scheme Meetings) convene the Scheme Meetings and the Westfield General Meeting in accordance with any orders made by the Scheme Court and in accordance with any applicable laws and regulations:
- (director's voting) use its reasonable endeavours to procure that each (e) Westfield Director votes any Westfield Stapled Securities in which they have a Relevant Interest in favour of the Transaction and the Schemes:
- (section 411(17)(b) statement) apply to ASIC for the production of a (f) statement pursuant to section 411(17)(b) of the Corporations Act stating that ASIC has no objection to the Share Scheme:
- (Court approval) subject to the Conditions Precedent, other than (g) paragraph (g) in clause 3.1 being satisfied or waived in accordance with this document, apply to the Scheme Court for:
 - an order approving the Share Scheme in accordance with (i) sections 411(4)(b) and 411(6) of the Corporations Act; and
 - (ii) the Second Judicial Advice:
- (h) (No Breach certificate) on the day prior to the Second Court Hearing, provide to Unibail a certificate signed by the Co-Chief Executive Officers of Westfield (on behalf of Westfield and without those individuals assuming liability of any kind) confirming to the best of the knowledge. belief or awareness of the Co-Chief Executive Officers, having made all reasonable enquiries of appropriate officers and employees of Westfield

who report to them and who could reasonably be expected to have information relevant to such matters, whether or not Westfield has breached any representation or warranty given by Westfield under this document or is reasonably expected to be in breach any of those representations or warranties as at 8:00am on the Second Court Date:

- (i) (Conditions Precedent certificate) at the hearing on the Second Court Date, provide to the Scheme Court (through its counsel):
 - a certificate confirming (in respect of matters within Westfield's (i) knowledge) whether or not the Conditions Precedent for which it is responsible, as noted in clause 3.1 (other than paragraph (g)), have been satisfied or waived in accordance with clause 3, a draft of which must be provided to Unibail by 5.00pm on the third Business Day prior to the Second Court Date and a copy of the final signed certificate must be provided to Unibail before 8.00am on the Second Court Date; and
 - (ii) any certificate provided to it by Unibail under clause 5.5(h);
- (j) (Westfield Constitutional Amendments) conditional on the passing of the Trust Scheme Resolutions, execute the Westfield Constitutional Amendments:
- (k) (lodge copy of Scheme Court order) lodge with ASIC:
 - (i) an office copy of the Scheme Court order approving the Share Scheme as approved by the Westfield Securityholders at the Share Scheme Meeting in accordance with section 411(10) of the Corporations Act; and
 - the duly executed Westfield Constitutional Amendments. (ii)
- **(I)** (Westfield Register) close the Westfield Register as at the Record Date to determine the identity of Scheme Participants and their entitlements to Scheme Consideration:
- (Suspension of trading) apply to ASX to suspend trading in Westfield (m) Stapled Securities with effect from the close of trading on the Effective Date;
- (n) (**listing**) take all reasonable steps to maintain Westfield's listing on ASX. notwithstanding any suspension of the quotation of the Westfield Stapled Securities, up to and including the Implementation Date, including making appropriate applications to ASX and ASIC;
- (Transaction) implement the Transaction in accordance with the Share (o) Scheme, the Westfield Constitutional Amendments and the Transaction Steps:
- (information) procure that Westfield's share registry delivers to Unibail: (p)
 - (i) promptly at Unibail's request, any information which Unibail reasonably requires for the purpose of understanding legal and beneficial ownership of Westfield Stapled Securities; and
 - on the date three weeks before the Scheme Meeting Date, on (ii) the date two weeks before the Scheme Meeting Date, and each Business Day in the week prior to the Scheme Meeting Date (inclusive) the following information in a form reasonably requested by Unibail: the total number of voting proxies in

© King & Wood Mallesons 37 Implementation Agreement

respect of the Share Scheme Resolutions and the Trust Scheme Resolutions delivered by Westfield Securityholders to Westfield, providing details of the aggregate number of proxies in favour of, against and abstaining from the relevant Share Scheme Resolutions and the Trust Scheme Resolutions and the aggregate number of Westfield Stapled Securities to which those proxies relate; and

(other steps) do all other things necessary to give effect to the (q) Transaction in accordance with this document, the Schemes and all applicable Scheme Court orders, laws and regulations.

5.4 Unibail's preliminary obligations

Unibail must:

- (a) (announce board support statements) following the execution of this document, announce, in the form contained in Annexure A (on the basis of statements made to Unibail by the Unibail Directors) that the Unibail Board has determined that the Contemplated Transaction is consistent with the corporate interest (intérêt social), business objectives and goals of Unibail, and is in the best interests of Unibail, its shareholders, employees and other stakeholders;
- (Unibail Information) prepare and promptly provide to Westfield for (b) inclusion in the Scheme Booklet the Unibail Information (in accordance with all applicable laws, including the Corporations Act, Corporations Regulations, Regulatory Guide 60, Regulatory Guide 74 and Guidance Note 15) and consent to the inclusion of that information in the Scheme Booklet:
- (c) (New Unibail Shares Listing Prospectus and New Unibail Stapled Securities Listing Prospectus) prepare and consult with Westfield as to the content of the New Unibail Shares Listing Prospectus and the New Unibail Stapled Securities Listing Prospectus, which includes:
 - (i) allowing Westfield a reasonable opportunity to review and make comments on successive drafts of the New Unibail Shares Listing Prospectus and the New Unibail Stapled Securities Listing Prospectus;
 - (ii) taking any reasonable comments made by Westfield into account in good faith when producing a revised draft of the New Unibail Shares Listing Prospectus and the New Unibail Stapled Securities Listing Prospectus; and
 - (iii) obtaining Westfield's consent to the inclusion of the Westfield Information (including in the form and context in which it appears) in the New Unibail Shares Listing Prospectus and the New Unibail Stapled Securities Listing Prospectus;
- (d) (Unibail Contribution Auditor Court Application) prepare and consult with Westfield as to the content of the Unibail Contribution Auditor Court Application, which includes:
 - (i) allowing Westfield a reasonable opportunity to review and make comments on the draft of the Unibail Contribution Auditor Court Application; and

38 © King & Wood Mallesons Implementation Agreement

- taking any reasonable comments made by Westfield into (ii) account in good faith when producing a revised draft of the Unibail Contribution Auditor Court Application;
- (Unibail Convening Notice and Unibail Notice of Meeting) prepare (e) and consult with Westfield as to the content and presentation of the Unibail Convening Notice and the Unibail Notice of Meeting, which include:
 - (i) allowing Westfield a reasonable opportunity to review and make comments on successive drafts of the Unibail Convening Notice and the Unibail Notice of Meeting; and
 - (ii) taking any reasonable comments made by Westfield into account in good faith when producing a revised draft of the Unibail Convening Notice and the Unibail Notice of Meeting:

For the avoidance of doubt, the obligation to prepare the Unibail Convening Notice and the Unibail Notice of Meeting does not in any way require Unibail to convene the Unibail Shareholders meeting until such time as this document has become binding in accordance with clause 2.2(a); and

(f) (Independent Expert information) provide any assistance or information reasonably requested by the Independent Expert in connection with the preparation of the Independent Expert's Report.

5.5 **Unibail's post-Consideration Period obligations**

Following the Consideration Period, Unibail must:

- (a) (post-ERB board recommendation) announce that the Unibail Board:
 - has determined that the Contemplated Transaction is consistent (i) with the corporate interest (intérêt social), business objectives and goals of Unibail, and is in the best interests of Unibail, its shareholders, employees and other stakeholders; and
 - (ii) recommends that all Unibail Shareholders vote in favour of each of the resolutions required for the Unibail Shareholder Approval;
- (b) (Unibail Contribution Auditor Court Application) request the appointment of a contribution auditor (commissaire aux apports) in the context of the contribution of the TH Newco Shares and the WCL Shares to Unibail, by filing the Unibail Contribution Auditor Court Application with the president of the commercial court (président du tribunal de commerce) of Nanterre;
- (c) (Convening Shareholders Meeting) convene the Unibail Shareholders Meeting by publishing the Unibail Convening Notice and the Unibail Notice of Meeting and hold the corresponding Unibail Shareholders Meeting in compliance with French law and Unibail's by-laws;
- (d) (**Deed Poll**) prior to the First Court Date, sign and deliver the Deed Poll;
- (further Unibail Information) promptly provide to Westfield any further (e) or new Unibail Information as may arise after the Scheme Booklet has been sent to the Westfield Securityholders and until the date of the Scheme Meetings, as may be necessary to ensure that the Unibail Information contained in the Scheme Booklet is not, having regard to applicable disclosure requirements, false, misleading or deceptive in any

material respect (including because of any material omission) and to ensure that there would be no breach of clause 13.3(h) if it applied as at the date on which such further or new Unibail Information arose;

- (supplementary disclosure New Unibail Shares Listing (f) **Prospectus and New Unibail Stapled Securities Listing Prospectus)** if, after despatch of the New Unibail Shares Listing Prospectus or the New Unibail Stapled Securities Listing Prospectus, Unibail becomes aware:
 - (i) that information included in the New Unibail Shares Listing Prospectus or the New Unibail Stapled Securities Listing Prospectus is or has become misleading or deceptive in any material respect (whether by omission or otherwise); or
 - (ii) of information that is required to be disclosed to the Unibail Shareholders under any applicable law but was not included in the New Unibail Shares Listing Prospectus or the New Unibail Stapled Securities Listing Prospectus,

Unibail must promptly consult with Westfield in good faith as to the need for, and the form of, any supplementary disclosure to the Unibail Shareholders, and make any disclosure that Unibail considers reasonably necessary in the circumstances, having regard to applicable laws:

- (supplementary disclosure Unibail Convening Notice and Unibail (g) Notice of Meeting) if, after despatch of the Unibail Convening Notice and the Unibail Notice of Meeting, Unibail becomes aware:
 - that information included in the Unibail Convening Notice and (i) the Unibail Notice of Meeting is or has become misleading or deceptive in any material respect (whether by omission or otherwise); or
 - (ii) of information that is required to be disclosed to the Unibail Shareholders under any applicable law but was not included in the Unibail Convening Notice and the Unibail Notice of Meeting,

Unibail must promptly consult with Westfield in good faith as to the need for, and the form of, any supplementary disclosure to the Unibail Shareholders, and make any disclosure that Unibail considers reasonably necessary in the circumstances, having regard to applicable laws;

- (Conditions Precedent certificate) before 8.00am on the Second Court (h) Date, provide to Westfield for provision to the Scheme Court at the hearing on that date a certificate confirming (in respect of matters within Unibail's knowledge) whether or not the Conditions Precedent for which Unibail is responsible, as noted in clause 3.1 (other than paragraph (g)). have been satisfied or waived in accordance with clause 3, a draft of which must be provided to Westfield by 5.00pm on the third Business Day prior to the Second Court Date:
- (i) (Listing) use all reasonable endeavours to:
 - obtain the New Unibail Shares Listing Prospectus Approval (i) (which includes filing the New Unibail Shares Listing Prospectus with the AMF and requesting the notification of the approved New Unibail Shares Listing Prospectus to the AFM);

- obtain the New Unibail Stapled Securities Listing Prospectus (ii) Approval (which includes filing the New Unibail Stapled Securities Listing Prospectus with the AMF and/or the AFM, as applicable);
- obtain the Euronext Approval (which includes, in particular, (iii) procuring that the New Unibail Stapled Securities to be issued pursuant to the Share Scheme and the Westfield Constitutional Amendments have been approved by Euronext Paris and Euronext Amsterdam, as applicable and that those New Unibail Stapled Securities have received listing and trading approval from each of Euronext Paris and Euronext Amsterdam, as applicable); and
- (iv) procure that Unibail and Newco BI are approved for admission to the official list of ASX (as a foreign exempt listing) and that the New Unibail CDIs to be issued as Scheme Consideration pursuant to the Share Scheme and the Westfield Constitutional Amendments are approved for official quotation by ASX, subject to any conditions which ASX may reasonable require and which are acceptable to Unibail, acting reasonably;
- (Transaction) implement the Transaction in accordance with the Share (j) Scheme, the Westfield Constitutional Amendments and the Transaction Steps;
- (k) (Scheme Consideration) if the Schemes become Effective, provide or procure the provision of the Scheme Consideration in the manner and amount contemplated by this document and the terms of the Share Scheme, Deed Poll and the Westfield Constitutional Amendments;
- (l) (No Breach certificate) on the day prior to the Second Court Hearing, provide to Westfield a certificate signed by the Chief Executive Officer of Unibail (on behalf of Unibail and without that individual assuming liability of any kind) confirming to the best of the knowledge, belief or awareness of the Chief Executive Officer, having made all reasonable enquiries of appropriate officers and employees of Unibail who report to the Chief Executive Officer and who could reasonably be expected to have information relevant to such matters, whether or not Unibail has breached any representation or warranty given by Unibail under this document or is reasonably expected to be in breach any of those representations or warranties as at 8:00am on the Second Court Date;
- (m) (representation) procure that it is represented by counsel at the Scheme Court hearings convened for the purposes of sections 411(1) and 411(4)(b) of the Corporations Act, at which, through its counsel, Unibail must undertake (if requested by the court) to do all things and take all steps within its power as may be necessary in order to ensure the fulfilment of its obligations under this document and the Schemes; and
- (n) (other steps) do all other things necessary to give effect to the Transaction in accordance with this document, the Schemes, Deed Poll and all applicable Scheme Court orders, laws and regulations.

5.6 **Transaction Steps**

On or prior to the Implementation Date, Unibail and Westfield must, or must procure that each of its Subsidiaries take all actions reasonably necessary or expedient to complete the Transaction Steps, including that each relevant party to the Transaction Steps takes the requisite action to attend, resolve, approve,

execute, deliver, procure, or otherwise complete each Transaction Step for which they are responsible and takes such other action as reasonably necessary to implement and complete the relevant Transaction Step including:

- (a) making filings with Regulatory Authorities;
- (b) obtaining consents, approvals, permits, authorisations and waivers from Regulatory Authorities; and
- (c) paying fees as necessary or appropriate.

5.7 Responsibility statement

- The responsibility statement to appear in the Scheme Booklet, in a form to be agreed by the parties, will contain words to the effect of:
 - (i) Westfield has prepared, and is responsible for, the content of the Scheme Booklet other than, to the maximum extent permitted by law, the Unibail Information, the Independent Expert's Report or any other report or letter issued to Westfield by a third party; and
 - (ii) Unibail has prepared, and is responsible for, the Unibail Information in the Scheme Booklet (and no other part of the Scheme Booklet).
- The responsibility statement to appear in the New Unibail Shares Listing (b) Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting, in a form to be agreed by the parties, will contain words to the effect of:
 - (i) Unibail has prepared, and is responsible for, the content of the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting (as the case may be) other than, to the maximum extent permitted by law, the Westfield Information or any other report or letter issued to Unibail by a Third Party;
 - (ii) Westfield has prepared, and is responsible for, the Westfield Information in the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting (as the case may be) and no other part of that document.

5.8 Disagreement on disclosure

If Unibail and Westfield disagree on the form or content of the Scheme Booklet, New Unibail Shares Listing Prospectus, New Unibail Stapled Securities Listing Prospectus, Unibail Convening Notice or Unibail Notice of Meeting, they must consult in good faith to try to settle an agreed form of the document. If complete agreement is not reached after reasonable consultation, then:

- if the disagreement relates to the form or content of the Unibail (a) Information contained in the Scheme Booklet, Westfield will make any amendments as Unibail reasonably requires;
- (b) if the disagreement relates to the form or content of any other part of the Scheme Booklet, the Westfield Board will, acting in good faith, decide the final form or content of the disputed part of the Scheme Booklet;

© King & Wood Mallesons 42 Implementation Agreement

- (c) if the disagreement relates to the form or content of the Westfield Information in the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice or the Unibail Notice of Meeting, Unibail will make any amendments as Westfield reasonable requires; and
- (d) if the disagreement relates to the form or content of any other part of the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice or the Unibail Notice of Meeting, the Unibail Board will, acting in good faith, decide the final form or content of the disputed part of the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice or the Unibail Notice of Meeting.

5.9 Verification

Each party must undertake appropriate verification processes for the information supplied by that party in the Scheme Booklet.

5.10 Conduct of Scheme Court proceeding

Westfield and Unibail are entitled to separate representation at all Scheme Court proceedings relating to the Schemes. This document does not give Westfield or Unibail any right or power to give undertakings to the Scheme Court for or on behalf of the other party without that party's written consent. Westfield and Unibail must give all undertakings to the Scheme Court in all Scheme Court proceedings which are reasonably required to obtain Scheme Court approval and confirmation of the Share Scheme as well as the Judicial Advice, as contemplated by this document.

5.11 Appeal process

If the Scheme Court refuses to make orders convening the Scheme Meetings, approving the Schemes, giving the First Judicial Advice or giving the Second Judicial Advice, Unibail and Westfield must appeal the Scheme Court's decision to the fullest extent possible except to the extent that:

- (a) the parties agree otherwise; or
- (b) an independent senior counsel of the New South Wales bar appointed by both parties advises that, in their opinion, an appeal would have no reasonable prospect of success before the End Date,

in which case either party may terminate this document in accordance with clause 14.1(g)(iii). If the parties cannot agree on the appointment of an independent senior counsel for the purposes of 5.11(b) within three Business Days of the Scheme Court's refusal to make orders, the independent senior counsel will be appointed by the chairperson or other senior office bearer for the time being of the New South Wales Chapter of the Resolution Institute or a delegate of such person, on the application of any party.

5.12 No partnership or joint venture

Subject to this document, nothing in this clause requires either party to act at the direction of the other. The business of each party will continue to operate independently from the other until the Implementation Date. The parties agree that nothing in this document constitutes the relationship of a partnership or a joint venture between the parties.

© King & Wood Mallesons | Implementation Agreement 43

6 Recommendations

6.1 **Westfield Board recommendation**

Westfield must procure that the Westfield Board does not withdraw or change (and must use its best endeavours to ensure that there is no individual Westfield Director who withdraws, or changes) their recommendation in favour of the Transaction and the Schemes, unless:

- (a) there is a Superior Proposal; or
- (b) the Independent Expert concludes that the Transaction and the Schemes are not in the best interests of the Westfield Securityholders, or adversely changes its previously given opinion that the Transaction and the Schemes are in the best interests of the Westfield Securityholders: and

the Westfield Board determines in good faith and acting reasonably, having received expert advice in writing from their legal advisors (who must be reputable advisers experienced in transactions of this nature), that they must do so because of their fiduciary or statutory duties to the Westfield Securityholders.

Westfield Board Voting Intention 6.2

Westfield must procure that none of the Westfield Directors who holds or controls the voting rights attached to Westfield Stapled Securities withdraw, or change their statement of intent under clause 5.2(a) to vote his or her Westfield Stapled Securities in favour of the Transaction and the Schemes, subject to:

- the Independent Expert concluding, and continuing to conclude, that the (a) Transaction and the Schemes are in the best interests of the Westfield Securityholders; and
- (b) there being no Superior Proposal.

6.3 **Westfield Directors**

Westfield represents and warrants to Unibail that it has been advised by each director of Westfield in office at the date of this document that he or she will make and maintain the recommendation and voting intention statement in accordance with clauses 6.1 and 6.2.

6.4 **Notification obligation**

If any fact matter or circumstance arises (including the receipt or expected receipt of an unfavourable report from the Independent Expert including any supplementary or updated report) which may lead to any one or more Westfield Directors changing, withdrawing or modifying his or her recommendation in favour of the Transaction and the Schemes (other than such circumstances as provided for in clause 10.7 or 10.8 which shall be regulated by those clauses), Westfield must promptly notify Unibail of this fact.

6.5 Unibail Board recommendation

Unibail must procure that the Unibail Board does not withdraw or change (and must use its best endeavours to ensure that no individual Unibail Director withdraws or changes) their recommendation referred to in:

(a) clause 5.4(a) before the expiry of the Consideration Period; or

(b) clause 5.5(a) after the expiry of the Consideration Period,

unless Unibail has received a Unibail Change of Control Proposal.

7 **Directors and employees**

Release of Westfield and Westfield directors and officers 7.1

Subject to the Corporations Act, Unibail releases its rights, and agrees with Westfield that it will not make a claim, against any Westfield Indemnified Party (other than Westfield and its Subsidiaries) as at the date of this document and from time to time in connection with:

- (a) any breach of any representations and warranties of Westfield or any other Westfield Group entity in this document; or
- any disclosures containing any statement which is false or misleading (b) whether in content or by omission,

whether current or future, known or unknown, arising at common law, in equity, under statute or otherwise, except where the Westfield Indemnified Party has not acted in good faith or has engaged in wilful misconduct or fraud. Nothing in this clause 7.1 limits Unibail's rights to terminate this document under clause 14.1.

7.2 **Benefit for Westfield Indemnified Parties**

Westfield receives and holds the benefit of this clause to the extent it relates to each Westfield Indemnified Party on behalf of each of them.

7.3 Release of Unibail and Unibail directors and officers

Subject to the Corporations Act, Westfield releases its rights, and agrees with Unibail that it will not make a claim, against any Unibail Indemnified Party (other than Unibail and its Subsidiaries) as at the date of this document and from time to time in connection with:

- (a) any breach of any representations and warranties of Unibail or any other member of the Unibail Group in this document: or
- (b) any disclosures containing any statement which is false or misleading whether in content or by omission.

whether current or future, known or unknown, arising at common law, in equity, under statute or otherwise, except where the Unibail Indemnified Party has not acted in good faith or has engaged in wilful misconduct or fraud. Nothing in this clause 7.3 limits Westfield's rights to terminate this document under clause 14.1.

7.4 **Benefit for Unibail Indemnified Parties**

Unibail receives and holds the benefit of this clause to the extent it relates to each Unibail Indemnified Party on behalf of each of them.

7.5 Appointment/retirement of directors

On the Implementation Date, but subject to the Scheme Consideration having been provided to the Scheme Participants in accordance with the Schemes and receipt by Westfield of signed consents to act:

Westfield must: (a)

- (i) cause the appointment of each Incoming Westfield Director to each Westfield Board: and
- procure that each of the Outgoing Westfield Directors retire from (ii) the Westfield Board and agree to provide a release by way of deed poll in favour of Westfield and Unibail and their Related Bodies Corporate, that they have no Claims outstanding, and waive all rights to bring any future Claims, against Westfield or Unibail (subject to the continued application of the indemnification and other provisions referred to in clauses 7.6 and 7.7) including for loss of office or remuneration or otherwise; and
- (b) Unibail must have taken all reasonable steps to have caused the appointment of the Incoming Unibail Director to the Unibail Board, including by having called and held a meeting of Unibail shareholders to approve such appointment subject to the Scheme Consideration having been provided to the Scheme Participants in accordance with the Schemes and receipt by Unibail of signed consents to act, and effective on the Implementation Date,

in each case, in accordance with the applicable entity's constituent documents and all applicable laws, regulations and rules.

7.6 Directors' and officers' insurance

Subject to the Schemes becoming Effective and subject to the Corporations Act, Unibail undertakes in favour of Westfield and each other person who is a Westfield Indemnified Party that it will:

- (a) for a period of 7 years from the Implementation Date, ensure that the constitutions of Westfield and each other Westfield Group entity continue to contain such rules as are contained in those constitutions at the date of this document that provide for each company to indemnify each of its directors and officers against any liability incurred by that person in his or her capacity as a director or officer of the company to any person other than a Unibail Group entity; and
- procure that Westfield and each other Westfield Group entity complies (b) with any deeds of indemnity, access and insurance made by them in favour of their respective directors and officers from time to time and without limiting the foregoing, not take any action which would prejudice or adversely affect any directors' and officers' run-off insurance cover taken out prior to the Implementation Date.

7.7 Run-off

Each party acknowledges that, notwithstanding any other provision of this document, Westfield and its Subsidiaries may, prior to the Implementation Date, enter into a run-off insurance policy in respect of any directors or officers of a Westfield Group for a 7 year period (or longer if Unibail agrees acting reasonably) provided that:

- (a) the scope of cover of the policy will be on the same or substantially the same terms as the existing insurance policies in place for directors or officers of the Westfield Group at the date of this document; and
- (b) Westfield will use reasonable endeavours to obtain the most attractive commercial terms for the policy from a reputable insurer and in no circumstances shall the aggregate cost of such insurance to the Westfield Group exceed the amount agreed between the parties in

writing for the purposes of this clause without the prior written consent of Unibail.

7.8 Period of undertaking

The undertakings contained in clause 7.6 are given until the earlier of the end of the relevant period specified in that clause or the relevant Westfield Group entity ceasing to be part of the Unibail Group.

7.9 **Benefit of undertaking for Westfield Group**

Westfield acknowledges that it receives and holds the benefit of clause 7.6 to the extent it relates to each director and officer of a Westfield Group entity on behalf of each of them.

Westfield conduct of business 8

8.1 Westfield conduct

Subject to clause 8.3, from the date of this document up to and including the Implementation Date, and without limiting any other obligations of Westfield under this document:

- (a) Westfield must conduct, and must procure that each Westfield Group entity conducts, its business in the ordinary course, and in substantially the same manner in which has been conducted in the 12 month period prior to the date of this document:
- Westfield must, and must procure that each Westfield Group entity uses (b) its reasonable endeavours to ensure that it:
 - (i) maintains the condition of its business and material assets:
 - keeps available the services of its officers and employees; (ii)
 - preserves its material relationships with Regulatory Authorities, (iii) customers, suppliers, licensors, licensees, joint venturers and others with whom it has business dealings:
 - (iv) complies in all material respects with all material contracts to which a member of the Westfield Group is a party, and with the material laws, authorisations and licences applicable to the Westfield Group; and
 - (v) complies with any anti-corruption or anti-bribery laws applicable to the Westfield Group, including the FCPA and the UK Bribery Act:
- subject to the Confidentiality Deed, Westfield must keep Unibail informed (c) of any material developments concerning the conduct of Westfield Group's business;
- Westfield must not enter into, and must ensure that none of its (d) Subsidiaries enters into, any line of business or other activities which are material to the business of the Westfield Group taken as a whole, and which it is not engaged as of the date of this document; and
- (e) Westfield must ensure that, between (and including) the date of this document and 8.00am on the Second Court Date, there is no occurrence

47 © King & Wood Mallesons Implementation Agreement

within its control or the control of any of its Subsidiaries that would constitute a Westfield Prescribed Event or Westfield Material Adverse Change.

8.2 **Prohibited actions**

Other than with the prior approval of Unibail (and for this purpose Unibail will use its reasonable endeavours to respond to any such request for approval within 5 days after such matter is submitted for approval by the Integration Committee), as notified by Westfield to Unibail in writing prior to the date of this document, or as required by this document, Westfield must not, and must ensure that each Westfield Group entity does not, during the period referred to in clause 8.1:

- (a) (real estate divestments) sell, dispose of or otherwise agree to sell or dispose of any interest in real property (whether directly or indirectly or through an agreement arrangement or understanding involving securities or assets) the value of which exceeds \$20,000,000;
- (b) (acquisitions, disposals or tenders):
 - (i) acquire or dispose of;
 - (ii) agree to acquire or dispose of; or
 - (iii) offer, propose, announce a bid or tender for,

any business, assets (other than as contemplated by clause 8.2(a)), entity or undertaking the value of which exceeds \$50,000,000 individually or \$150,000,000 in aggregate, (outside transactions otherwise Disclosed or approved) provided that with respect to real estate divestments the provisions of clause 8.2(a) shall apply:

(c) (employment arrangements):

- (i) other than in the ordinary course, enters into any collective bargaining agreement (or equivalent) or terminates or extends any such agreement or amends any such agreement in any material respect;
- (ii) increase the remuneration of (including with regard to superannuation benefits) or benefits provided to or pay any retention amounts or bonus (other than in accordance with existing arrangements) or otherwise vary the employment agreements with any of its directors or Key Management Personnel:
- (iii) issue any securities or options to any of its directors or employees except as contemplated by clause 4.5:
- materially amend or modify any the terms of any other employee (iv) entitlement or employee benefit plan (including any superannuation or pension arrangements) in place at the date of this document; and
- waive any non-compete rights against Westfield Group directors, (v) officers or Key Management Personnel;
- (accelerate rights) accelerate the rights of any officer, employee, (d) contractor or consultant to compensation (including any incentives such as (without limitation) performance shares, performance rights, options

and other incentive rights) except as contemplated in clause 4.5 in relation to outstanding Employee Rights;

- (e) (employee loans) waive or forgive any loans made to any director, officer or employee of the Westfield Group, including any loan in respect of Westfield Stapled Securities issued under any employee or director share, Employee Rights, option or incentive plan except in accordance with the terms of those loans in place as at 1 September 2017;
- (f) (termination payments) pay a termination payment:
 - (i) to any Westfield Director or any Key Management Personnel; or
 - (ii) in excess of \$250,000, to any employee of Westfield other than any Key Management Personnel,

except as provided for in an existing employment contract or redundancy or termination policy in place as at 1 September 2017 (or any later date of employment);

- (g) (distributions) make or declare, or announce an intention to make or declare or pay, any distribution (whether by way of trust distribution, dividend, capital reduction or otherwise and whether in cash or in specie) other than:
 - (i) a distribution of no more than \$0.255 per Westfield Stapled Securities for the financial year ending 31 December 2017;
 - (ii) a distribution by any Westfield Group entity where the recipient of that distribution is Westfield or a Wholly Owned Subsidiary of Westfield; or
 - (iii) distributions made to any joint venture in the ordinary course under a joint venture or partnership agreement to which a Westfield Group entity is a party and which has been Disclosed to Unibail prior to the date of this document or otherwise agreed to in writing by Unibail;
- (h) (material agreements):
 - (i) enter into or materially adversely amend or terminate any Westfield Significant Contract or third party management agreement over a shopping centre or airport; or
 - (ii) enter into, materially adversely amend or terminate any joint venture or partnership agreement which is for a duration of over 2 years and which involves expenditure greater than \$20,000,000;
- (i) (debt) incur any financial indebtedness to a Third Party financier except for:
 - (i) drawing down on any Westfield Debt Facility within existing facility limits and consistent with past practice in the ordinary course of business:
 - (ii) the refinancing of any Westfield Debt Facility;
 - (iii) the financing of any Westfield global insurance policy premiums;

© King & Wood Mallesons 35219573_1

(iv) amounts less than \$5,000,000.

and excluding any such indebtedness, arrangement or financial accommodation (irrespective of what form that accommodation takes) to or among members of the Westfield Group;

- (advisor fees) pay any success or similar fee to any advisor in (j) connection with the Transaction or the Schemes in excess of \$1,000,000;
- (k) (loans) make any loans, advances or capital contributions to, or investments in, any other person of an amount greater than \$500,000, other than:
 - (i) to a Wholly Owned Subsidiary of Westfield; or
 - to a joint venture in which Westfield is a joint venture member in (ii) the ordinary course of business (including the financing of development projects) consistent with past practice:
- (I) (security) create, or agree to create, any mortgage, charge, lien or other encumbrance over the whole, or a substantial part, of the business or assets of the Westfield Group other than in the ordinary course of its business;
- (intellectual property) abandon or permit to let lapse or expire any (m) registration of the "Westfield" trademark or grant a new licence of the "Westfield" trademark to any person except a Wholly Owned Subsidiary or a joint venture of Westfield;
- (derivatives) enter into an agreement, arrangement or transaction with (n) respect to derivative instruments (including swaps, futures contracts, forward commitments, commodity derivatives or options) or similar instruments, except:
 - (i) derivative instruments entered in the ordinary course of business and consistent with past practice or in relation to the payment of a distribution by Westfield; or
 - (ii) in connection with the rolling over or renewals of existing positions in the ordinary course of business;
- (o) (related party) any Westfield Group entity enters into any transaction or agreement with a Related Party of any Westfield Group entity exceeding \$100,000;
- (organisational structure) any Westfield Group entity undertakes a (q) material restructure or reorganisation of its employees:
- (accounting policies) changes its accounting policy other than any (q) change required by applicable accounting standards;
- (Claims) any Westfield Group entity settles or admits liability in relation (r) to, or waives, any Claim for an amount exceeding \$5,000,000 excluding Claims covered by insurance where the insurer has accepted liability for the Claim: or
- (agree) authorises or agrees, offers, commits or resolves to do any of (s) the things referred to above whether conditionally or otherwise.

8.3 **Exceptions to conduct of business provisions**

Nothing in this clause 8 restricts the ability of Westfield to take any action which:

- (a) is required or expressly permitted by this document, the Transaction or the Schemes:
- (b) is required by any applicable law;
- is Fairly Disclosed in the Westfield Budgets; (c)
- (d) has been unanimously approved by the Integration Committee; or
- (e) has been agreed to in writing by Unibail.

8.4 Access to people and Westfield Information

Between the date of this document and the Implementation Date, Westfield must, and must procure that each of its Subsidiaries, respond to reasonable requests from Unibail and its representatives for information concerning the Westfield Group (as soon as reasonably practicable after such requests are made) and provide Unibail and its officers and advisers with reasonable access to Westfield Group's officers and advisers and documents, records, and other information (subject to any existing confidentiality obligations owed to third parties, or applicable privacy laws) which Unibail reasonably requires, in each case for the purposes of:

- applying for all relevant Regulatory Approvals including responding to (a) any follow-up questions or engaging in any discussion in connection with any Regulatory Approvals;
- (b) the implementation of the Transaction and the Schemes;
- the refinancing of any Westfield Group debt facilities which may need to (c) be repaid in connection with the Transaction and the Schemes:
- (d) the integration of the Westfield Group and the Unibail Group following the implementation of the Transaction and the Schemes; or
- any other purpose that is agreed in writing between the parties. (e)

8.5 **OneMarket Demerger**

- WCL proposes to implement the OneMarket Demerger concurrently with (a) the implementation of the Transaction. The OneMarket Demerger will have the general commercial features that have been agreed between Westfield and Unibail as at the date of this document, and as may otherwise be agreed between Westfield and Unibail from time to time.
- (b) In order to implement the OneMarket Demerger, WCL and OneMarket will enter into a demerger implementation deed and a demerger deed on reasonable commercial terms for demerger transactions of this nature. Westfield must obtain the prior approval of Unibail before entering into these OneMarket Demerger Deeds, such approval not to be unreasonably withheld or delayed. Without limitation, it would be reasonable for Unibail to withhold its approval in circumstances where the terms of the OneMarket Demerger Deeds do not include a full indemnity from OneMarket to each Westfield Group entity for any contingent liabilities for Taxes and other liabilities arising from the reorganisation relating to the Onemarket Demerger. The Costs in connection with the OneMarket Demerger will be borne equally between

- WCL and OneMarket save that WCL's liability in respect of such Costs will be capped at A\$2,500,000.
- (c) WCL must implement the OneMarket Demerger in accordance with the OneMarket Demerger Deeds. Any amendment to or waiver of the terms of the OneMarket Demerger Deeds will require the prior approval of Unibail, such approval not to be unreasonably withheld or delayed.
- (d) In connection with the OneMarket Demerger Westfield and Unibail will negotiatein good faith the entry of a network participation agreement with OneMarket. Westfield must obtain the prior approval of Unibail before entering into a network participation agreement with OneMarket.

8.6 Integration planning

- Without limiting clause 8.4, during the period from the date of this (a) document up to and including the Implementation Date. Westfield must co-operate in good faith with Unibail and provide Unibail with reasonable assistance, and commit sufficient resources, in connection with Unibail's' planning for integration of the Westfield Group and the Unibail Group.
- (b) Without limiting clause 8.6(a), on and from the date of this document, the parties agree to establish a committee ("Integration Committee") initially comprising of the individuals agreed by Westfield and Unibail prior to the date of this document.
- The role of the Integration Committee will be to act as a forum for (c) discussion and planning in respect of the following:
 - (i) matters related to integration planning, including employee retention and incentivisation, stakeholder engagement and communications, consolidation of operations, functions or processes and shared services, of the future combined Westfield Group and Unibail Group and other relevant matters:
 - (ii) approvals for purposes of clause 8.2;
 - (iii) the process and management of requests for Unibail's approval under clause 8.2:
 - the manner in which information and access will be provided (iv) pursuant to clause 8.4; and
 - (v) the list of Third Parties to whom notice must be given, or from whom consent or approval is required, as contemplated by clause 8.7(a), and the process for giving notice to, or seeking consent or approval from, such Third Parties (as applicable).
- (d) Each party must ensure that its representatives on the Integration Committee:
 - have the requisite level of experience, knowledge and seniority (i) to represent it on the Integration Committee and to meaningfully contribute to discussion of the relevant matters at meetings of the Integration Committee; and
 - (ii) act in good faith in their capacity as members of the Integration Committee with a view to fulfilling the role and objectives of such committee (to the extent within their power).

- The Integration Committee will meet at least weekly and at such times (e) and places as agreed between the members of the Integration Committee from time to time. Meetings may be held via telephone.
- The members of the Integration Committee may agree to invite other (f) persons to attend meetings of the Integration Committee from time to time.
- From time to time, representatives of the parties will meet separately to (g) meetings of the Integration Committee to discuss and progress matters considered or plans developed by the Integration Committee.
- (h) The parties acknowledge and agree that:
 - the Integration Committee is a discussion and planning forum (i) only, and the members of the Integration Committee do not have power to bind the other party or to give any consent, approval or waiver on behalf of such other party:
 - (ii) nothing in this clause 8.6 or elsewhere in this document requires a party to act at the direction of the other party or is intended to create a relationship of partnership, joint venture or similar between the parties;
 - (iii) the respective businesses of the Westfield Group and the Unibail Group are to continue to operate independently and will not implement any integration plans until (and subject to) implementation of the Schemes; and
 - (iv) they will not, and must ensure that none of their representatives on the Integration Committee use the Integration Committee to impede the conduct of, or otherwise interfere with the operations of the Westfield Group or the Unibail Group.

8.7 Change of control consents

- (a) Subject to clause 8.7(d), Westfield must use reasonable endeavours to identify contracts or other arrangements with Third Parties under which Westfield Group may need to seek consent or approval in connection with the Transaction (including in respect of the change in control of Westfield resulting from implementation of the Transaction and the Schemes).
- Westfield and Unibail must consult as to the process for giving notice to, (b) or seeking approval or consent from, Third Parties (as applicable) to any of the contracts or arrangements referred to in clause 8.7(a) and
 - Westfield must procure that each such notice is given to, or (i) approval or consent is requested from, such Third Parties (as applicable) as soon as reasonably practicable after the date of this document and in any event prior to the Implementation Date:
 - (ii) Westfield must use reasonable endeavours to procure that each approval or consent is obtained from such Third Parties (as applicable) prior to the Implementation Date; and
 - (iii) Unibail must provide reasonable assistance to Westfield in connection with giving any such notice or seeking any such approval or consent.

53 © King & Wood Mallesons Implementation Agreement

- Without limiting the foregoing. Westfield must promptly provide Unibail (c) with details and (where applicable) copies of all material communications with Third Parties in connection with seeking such approval or consent.
- Any notice, approval or consent of a kind referred to in clause 8.7(a) may (d) only be given or sought by Westfield in a form and on terms approved by Unibail (such approval not to be unreasonably withheld or delayed).
- Nothing in clause 8.6(c)(v) or in this clause 8.7 prevents Westfield from (e) contacting any third party referred to in clause 8.7(a) at or around the time that the Transaction is first announced by Westfield to ASX to notify that third party of the Transaction provided copies of all such correspondence is provided to Unibail promptly and without delay.

Unibail conduct of business 9

9.1 **Unibail conduct**

Subject to clause 9.2, from the date of this document up to and including the Implementation Date, and without limiting any other obligations of Unibail under this document:

- Unibail must conduct, and must procure that each Unibail Group entity (a) conducts, its business in the ordinary course, and in substantially the same manner in which has been conducted in the 12 month period prior to the date of this document:
- Unibail must, and must procure that each Unibail Group entity uses its (b) reasonable endeavours to ensure that it:
 - (i) maintains the condition of its business and material assets:
 - (ii) keeps available the services of its officers and employees;
 - preserves its material relationships with Regulatory Authorities, (iii) customers, suppliers, licensors, licensees, joint venturers and others with whom it has business dealings;
 - (iv) complies in all material respects with all material contracts to which a member of the Unibail Group is a party, and with the material laws, authorisations and licences applicable to the Unibail Group; and
 - complies with any anti-corruption or anti-bribery laws applicable (v) to the Unibail Group, including the FCPA and the UK Bribery Act:
- (c) subject to the Confidentiality Deed, Unibail must keep Westfield informed of any material developments concerning the conduct of the Unibail Group's business;
- (d) Unibail must not enter into, and must ensure none of its Subsidiaries enters into, any line of business or other activities which are material to the business of the Unibail Group taken as a whole, in which it is not engaged as of the date of this document; and
- (e) Unibail must ensure that, between (and including) the date of this document and 8.00am on the Second Court Date there is no occurrence within its control or the control of any of its Subsidiaries that would

© King & Wood Mallesons 54 Implementation Agreement

constitute a Unibail Prescribed Event or Unibail Material Adverse Change.

9.2 Exceptions to conduct of business provisions

Nothing in this clause 8 restricts the ability of Unibail to take any action which:

- (a) is required or expressly permitted by this document, the Transaction or the Schemes:
- (b) is necessary or desirable in connection with the financing or implementation of the Schemes or the Transaction;
- (c) is required by any applicable law;
- (d) has been Disclosed to Westfield prior to the date of this document; or
- (e) has been agreed to in writing by Westfield.

10 Exclusivity

10.1 No existing discussions

Westfield represents and warrants that:

- other than the discussions with Unibail in respect of the Transaction, it is not currently in negotiations or discussions relating to any actual, proposed or potential Competing Proposal; and
- (b) no confidential information has been provided to any Third Party since 1 January 2017 under a confidentiality agreement in relation to an actual, proposed or potential Competing Proposal.

10.2 No-shop

During the Exclusivity Period, Westfield must ensure that neither it nor any of its Representatives directly or indirectly:

- (a) solicits, invites, encourages or initiates any enquiries, negotiations or discussions; or
- (b) communicates any intention to do any of these things,

with a view to obtaining, or which may reasonably be expected to lead to, any offer, proposal or expression of interest from any person in relation to a Competing Proposal.

10.3 No-talk

Subject to clause 10.5, during the Exclusivity Period, Westfield must ensure that neither it nor any of its Representatives:

- (a) negotiates or enters into; or
- (b) participates in negotiations or discussions with any other person in relation to.

a Competing Proposal or any agreement, understanding or arrangement that may be reasonably expected to lead to a Competing Proposal, even if that

© King & Wood Mallesons | Implementation Agreement 55

person's Competing Proposal was not directly or indirectly solicited, invited. encouraged or initiated by Westfield or any of its Representatives or the person has publicly announced the Competing Proposal.

10.4 Due diligence information

Subject to clauses 10.5 and 10.6, during the Exclusivity Period, Westfield must ensure that neither it nor any of its Representatives:

- (a) enables any other person other than Unibail to undertake due diligence investigations on any Westfield Group entity or their businesses or operations; or
- (b) makes available to any other person, or permits any other person to receive, other than Unibail, any non-public information relating to any Westfield Group entity or their businesses or operations,

in connection with the formulation, development or finalisation of an actual, proposed or potential Competing Proposal (including by way of providing information and access to perform due diligence on the Westfield Group or any part of it).

10.5 **Exceptions**

Clauses 10.3 and 10.4 do not apply to the extent that they restrict Westfield or any Westfield Board from taking or refusing to take any action with respect to a genuine Competing Proposal (which was not solicited, invited, encouraged or initiated by Westfield in contravention of clause 10.2) provided that the Westfield Board has determined, in good faith and acting reasonably that:

- after consultation with its financial advisors, such a genuine Competing (a) Proposal is, or could reasonably be considered to become, a Superior Proposal; and
- (b) after receiving written legal advice from its external legal advisers (who must be reputable advisers experienced in transactions of this nature) that failing to respond to such a genuine Competing Proposal would be likely to constitute a breach of the Westfield Directors' fiduciary or statutory obligations.

10.6 **Further exceptions**

Nothing in this clause 10 prevents Westfield from:

- (a) continuing to make normal presentations to, and responding to enquiries from, brokers, portfolio investors and analysts in the ordinary course in relation to the Transaction or its business generally; or
- (b) fulfilling its continuous disclosure obligations under the ASX Listing Rules.

10.7 Notice of approaches

- During the Exclusivity Period, Westfield must promptly notify Unibail if it, (a) or any of its Related Bodies Corporate or Representatives, becomes aware (and in any event within 24 hours of becoming aware) of any:
 - (i) negotiations discussions or, approach in respect of any actual, proposed or potential Competing Proposal:

- provision by Westfield or any of its Related Bodies Corporate or (ii) Representatives of any non-public information concerning the business or operations or Westfield or the Westfield Group to any person in connection with an actual, proposed or potential Competing Proposal.
- (b) A notification given under clause 10.7(a) must include the identity of the relevant person making or proposing to make the actual, proposed or potential Competing Proposal, together with the material terms and conditions of the actual, proposed or potential Competing Proposal.

10.8 Matching right

Without limiting clauses 10.2 and 10.3, during the Exclusivity Period, Westfield:

- must not enter into any legally binding agreement, arrangement or (a) understanding (whether or not in writing) pursuant to which a Third Party. Westfield or both proposes or propose to undertake or give effect to an actual, proposed or potential Competing Proposal; and
- (b) must procure that none of the Westfield Directors change their recommendation in favour of the Schemes to publicly recommend an actual, proposed or potential Competing Proposal (or recommend against the Schemes),

unless:

- the Westfield Board acting in good faith and in order to satisfy what the (c) Westfield Board consider to be their statutory or fiduciary obligations (having received written advice from its external legal advisers who must be reputable advisers experienced in transactions of this nature), determines that the Competing Proposal would be or would be reasonably likely to be an actual, proposed or potential Superior Proposal:
- (d) Westfield has provided Unibail with the material terms and conditions of the actual, proposed or potential Competing Proposal, including price and the identity of the Third Party making the actual, proposed or potential Competing Proposal;
- (e) Westfield has given Unibail at least 7 days after the date of the provision of the information referred to in clause 10.8(d) to provide a matching or superior proposal to the terms of the actual, proposed or potential Competing Proposal; and
- Unibail has not announced or otherwise formally proposed to Westfield a (f) matching or superior proposal to the terms of the actual, proposed or potential Competing Proposal by the expiry of the 7 day period referred to in clause 10.8(e).

Any modification to any Competing Proposal (which will include any modification relating to the price or value of any Competing Proposal) will be taken to make that proposal a new Competing Proposal in respect of which Westfield must comply with its obligations under this clause 10.8 again.

10.9 Unibail counterproposal

If Unibail proposes to Westfield, or announces amendments to the Transaction and the Schemes or a new proposal that constitute a matching or superior proposal to the terms of the actual, proposed or potential Competing Proposal ("Unibail Counterproposal") by the expiry of the 7 day period referred to in

clause 10.8(e), Westfield must procure that the Westfield Board considers the Unibail Counterproposal and if the Westfield Board, acting reasonably and in good faith, determines that the Unibail Counterproposal would provide an equivalent or superior outcome for the Westfield Securityholders as a whole compared with the Competing Proposal, taking into account all of the terms and conditions of the Unibail Counterproposal, then:

- (a) Westfield and Unibail must negotiate in good faith and use their best endeavours to agree the amendments to this document and, if applicable, the Schemes, the Deed Poll and the Westfield Constitutional Amendments that are reasonably necessary to reflect the Unibail Counterproposal and to implement the Unibail Counterproposal, in each case as soon as reasonably practicable; and
- (b) Westfield must procure that each of the Westfield Directors continues to recommend the Transaction and the Schemes (as modified by the Unibail Counterproposal) to the Westfield Securityholders.

10.10 Provision of Information

During the Exclusivity Period, Westfield must as soon as possible provide Unibail with:

- (a) in the case of written materials, a copy of; and
- (b) in any other case, a written statement of,

any non-public information about the business or affairs of Westfield or the Westfield Group disclosed or otherwise provided to any Third Party in connection with an actual, proposed or potential Competing Proposal that has not previously been provided to Unibail. For the avoidance of doubt, any such provision of information to a Third Party may only be undertaken if permitted by clause 10.5.

10.11 Legal advice

Westfield acknowledges that it has received legal advice on this document and the operation of this clause from reputable advisers experienced in transactions of this nature.

11 Break Fee

11.1 Background

This clause has been agreed in circumstances where:

- (a) Unibail and Westfield believe that the Transaction and the Schemes will provide significant benefits to Unibail, Westfield and their respective securityholders, and Unibail and Westfield acknowledge that, if they enter into this document and the Transaction and the Schemes are subsequently not implemented, Unibail will incur significant costs, including those set out in clause 11.5;
- (b) Unibail requested that provision be made for the Break Fee, without which Unibail would not have entered into this document;
- (c) both the Unibail Board and the Westfield Board believe that it is appropriate for both parties to agree to the payment referred to in this clause to secure Unibail's participation in the Schemes; and

© King & Wood Mallesons | Implementation Agreement 58

(d) both parties have received legal advice on this document and the operation of this clause.

11.2 Payment by Westfield to Unibail

Westfield agrees to pay the Break Fee to Unibail without withholding (unless required by law, in which case the amount of the payment due by Westfield shall be increased to an amount which (after making any such withholding) leaves an amount equal to the payment which would have been due if no withholding had been required) or set off if:

- (a) (Competing Proposal) on or before the End Date a Competing Proposal is announced and within 12 months of the End Date the Third Party who announced or made the Competing Proposal (or any of its Associates) completes a Competing Proposal;
- (b) (change of recommendation) any Westfield Director fails to recommend the Schemes or withdraws their recommendation, adversely changes or qualifies their recommendation or otherwise makes a public statement indicating that he or she no longer supports the Transaction and the Schemes, except where the change of recommendation or statement is made after the Independent Expert concludes that in the opinion of the Independent Expert the Transaction and the Schemes are not in the best interests of the Westfield Securityholders (other than where the conclusion is due wholly or in material part to the existence, announcement or publication of a Competing Proposal); or
- (Termination) Unibail validly terminates this document in accordance (c) with clause 14.1(d).

11.3 No amount payable if Schemes become Effective

Notwithstanding the occurrence of any event in clause 11.2, if the Transaction and the Schemes become Effective:

- (a) no amount is payable by Westfield under clause 11.2; and
- (b) if any amount has already been paid under clause 11.2 that amount must be refunded by Unibail less the incremental costs incurred by Unibail as a result of the event giving rise to the obligation to pay the Break Fee.

11.4 Timing of payment

- (a) A demand by Unibail for payment of the Break Fee under clause 11.2 must:
 - be in writing; (i)
 - (ii) be made after the occurrence of the event in that clause giving rise to the right to payment;
 - (iii) state the circumstances which give rise to the demand; and
 - nominate an account in the name of Unibail into which Westfield (iv) must pay the Break Fee.
- (b) Westfield must pay the Break Fee to Unibail under clause 11.2 within 5 Business Days of receipt by Westfield of a valid demand for payment from Unibail under clause 11.4(a), which demand may only be made after the occurrence of an event referred to in clause 11.2.

Without limiting clause 14.3, this clause 11.4 survives termination of this (c) document such that Unibail may give Westfield a demand for payment of the Break Fee even if this document has been terminated, provided that the event referred to in clause 11.2 occurred prior to the termination of this document.

11.5 Nature of payment

The Break Fee is an amount to compensate Unibail for:

- (a) advisory costs;
- (b) costs of management and directors' time;
- (c) out-of-pocket expenses:
- (d) the distraction of Unibail's management from conducting Unibail's business as usual caused by pursuing the Transaction and the Schemes:
- reasonable opportunity costs incurred by Unibail in pursuing the (e) Transaction and the Schemes or in not pursuing alternative acquisitions or strategic initiatives which Unibail could have developed to further its business and objectives; and
- damage to Unibail's reputation associated with a failed transaction and (f) the implications of that damage to Unibail's business.

The parties agree that the costs incurred are of a nature that they cannot be accurately quantified and that a genuine pre-estimate of the costs would equal or exceed the amount payable under clause 11.2. The parties intend and shall use their reasonable endeavours to secure that the Break Fee is not treated for VAT purposes or for GST purposes as consideration for a taxable supply. If, however, the Break Fee is treated by a Taxing Authority, in whole or in part, as being subject to VAT which is payable otherwise than by way of reverse charge, then the amount of the Break Fee shall be regarded as inclusive of such VAT or GST.

11.6 Break Fee does not limit liability

Notwithstanding any other provision in this document, this clause 11, or a payment by Westfield of the Break Fee in accordance with this clause 11, does not limit the liability of Westfield to any other party under or in connection with this document or otherwise reduce or limit the amount which may be recovered by Unibail as a result of a Claim against Westfield in respect of any matter under or in connection with this document.

12 **Reverse Break Fee**

12.1 **Background**

This clause has been agreed in circumstances where:

(a) Unibail and Westfield believe that the Transaction and the Schemes will provide significant benefits to Unibail. Westfield and their respective securityholders, and Unibail and Westfield acknowledge that, if they enter into this document and the Transaction and the Schemes are subsequently not implemented. Westfield and the Westfield Securityholders will incur significant costs including those set out in clause 12.5;

60 © King & Wood Mallesons Implementation Agreement

- Westfield requested that provision be made for the payment of the (b) Reverse Break Fee, without which Westfield would not have entered into this document:
- both the Unibail Board and the Westfield Board believe that it is (c) appropriate for both parties to agree to the payment referred to in this clause to secure Westfield's participation in the Transaction and the Schemes: and
- (d) both parties have received legal advice on this document and the operation of this clause.

12.2 Payment by Unibail to Westfield

Unibail agrees to pay the Reverse Break Fee to Westfield without withholding (unless required by law, in which case the amount of the payment due by Unibail shall be increased to an amount which (after making any such withholding) leaves an amount equal to the payment which would have been due if no withholding had been required) or set off if the Transaction and the Schemes do not proceed because:

- (a) (Bid for Unibail) the Unibail Board withdraws, adversely changes or qualifies its recommendation referred to in clause 5.4(a) or 5.5(a) in connection with, or as required under, a Unibail Change of Control Proposal;
- (b) (Termination due to ERB) Unibail validly terminates this document in accordance with clause 2.2(g); or
- (c) (Termination) Westfield validly terminates this document in accordance with clause 14.1(d).

12.3 No amount payable if Schemes become Effective

Notwithstanding the occurrence of any event in clause 12.2, if the Transaction and the Schemes become Effective:

- no amount is payable by Unibail under clause 12.2; and (a)
- (b) if any amount has already been paid under clause 12.2 that amount must be refunded by Westfield.

12.4 Timing of payment

- A demand by Westfield for payment of the Reverse Break Fee under (a) clause 12.2 must:
 - (i) be in writing;
 - (ii) be made after the occurrence of the event in that clause giving rise to the right to payment;
 - (iii) state the circumstances which give rise to the demand; and
 - nominate an account in the name of Westfield into which Unibail (iv) must pay the Reverse Break Fee.
- Unibail must pay the Reverse Break Fee to Westfield without withholding (b) or set off within 5 Business Days of receipt by Unibail of a valid demand for payment from Westfield under clause 12.4(a), which demand may only be made after the occurrence of an event referred to in clause 12.2.

Without limiting clause 14.3, this clause 12.4 survives termination of this (c) document such that Westfield may give Unibail a demand for payment of the Reverse Break Fee even if this document has been terminated. provided that the event referred to in clause 12.2 occurred prior to the termination of this document.

12.5 Nature of payment

The Reverse Break Fee is an amount to compensate Westfield for:

- (a) advisory costs;
- (b) costs of management and directors' time;
- (c) out-of-pocket expenses:
- (d) the distraction of Westfield's management from conducting Westfield's business as usual caused by pursuing the Transaction and the Schemes:
- reasonable opportunity costs incurred by Westfield in pursuing the (e) Transaction and the Schemes or in not pursuing strategic initiatives which Westfield could have developed to further its business and objectives; and
- damage to Westfield's reputation associated with a failed transaction and (f) the implications of that damage to Westfield's business.

The parties agree that the costs incurred are of a nature that they cannot be accurately quantified and that a genuine pre-estimate of the costs would equal or exceed the amount payable under clause 12.2. The parties intend and shall use their reasonable endeavours to secure that the Reverse Break Fee is not treated for VAT purposes or for GST purposes as consideration for a taxable supply. If, however, the Reverse Break Fee is treated by a Taxing Authority, in whole or in part, as being subject to VAT which is payable otherwise than by way of reverse charge, then the amount of the Reverse Break Fee shall be regarded as inclusive of such VAT or GST.

12.6 Reverse Break Fee does not limit liability

Notwithstanding any other provision in this document, this clause 12, or a payment by Unibail of the Reverse Break Fee in accordance with this clause 12, does not limit the liability of Unibail to any other party under or in connection with this document or otherwise reduce or limit the amount which may be recovered by Unibail as a result of a Claim against Westfield in respect of any matter under or in connection with this document.

13 Representations and warranties

13.1 Westfield's representations and warranties

Westfield represents and warrants to Unibail that each of the following statements is true and correct as at the dates specified in clause 13.2 of this document:

(a) (status) it has been duly incorporated or formed, and is validly existing under, in accordance with the laws of its place of incorporation;

- (power) it has power to enter into this document, to comply with its (b) obligations under it and exercise its rights under it:
- (no contravention) the entry by it into, its compliance with its obligations (c) and the exercise of its rights under, this document do not and will not conflict with:
 - (i) its constituent documents or cause a limitation on its powers or the powers of the Westfield Directors to be exceeded; or
 - (ii) any law or treaty or a judgment, ruling, order or decree binding on or applicable to it or its assets;
 - (iii) except as Disclosed, any other document or agreement that is binding on it or any of its Subsidiaries:

(WAT) (d)

- (i) WAML:
 - (A) is the only trustee and responsible entity of WAT;
 - (B) has been validly appointed as trustee and responsible entity of WAT; and
 - (C) is not in default under the WAT Constitution;
- (ii) no action has been taken to either terminate WAT or wind up WAT whether under Chapter 5C of the Corporations Act or otherwise other than any frivolous or vexatious application, or any application that is set aside within 14 days or, if earlier, by the Second Court Date: and
- (iii) subject to law and the WAT Constitution, WAML has the right to be fully indemnified out of the WAT Property in respect of the obligations incurred by it as responsible entity of WAT under the documents to which it is a party, and there is nothing that would prevent WAML from being fully indemnified out of the WAT Property for any obligations under or in connection with this document or any of the transactions contemplated by this document that it incurs in its capacity as responsible entity of WAT:

(WFDT) (e)

- (i) WAML:
 - (A) is the only trustee and responsible entity of WFDT:
 - (B) has been validly appointed as trustee and responsible entity of WFDT; and
 - (C) is not in default under the WFDT Constitution;
- (ii) no action has been taken to either terminate WFDT or wind up WFDT whether under Chapter 5C of the Corporations Act or otherwise other than any frivolous or vexatious application, or any application that is set aside within 14 days or, if earlier, by the Second Court Date; and

(iii) subject to law and the WFDT Constitution, WAML has the right to be fully indemnified out of the WFDT Property in respect of the obligations incurred by it as responsible entity of WFDT under the documents to which it is a party, and there is nothing that would prevent WAML from being fully indemnified out of the WFDT Property for any obligations under or in connection with this document or any of the transactions contemplated by this document that it incurs in its capacity as responsible entity of WFDT;

(f) (authorisations)

- (i) other than the approvals specified in the Conditions Precedent, it has in full force and effect each authorisation necessary for it to enter into this document, to comply with its obligations and exercise its rights under it, and to allow them to be enforced; and
- (ii) the Westfield Group holds all material licenses, franchises, permits, certificates, approvals or other similar authorisations issued by applicable Regulatory Authorities necessary for the lawful conduct of its businesses as presently conducted as of the date of this document, and except as Disclosed, to Westfield's knowledge, such licenses, franchises, permits, certificates, approvals or other similar authorisations are valid and in full force and effect, no Westfield Group entity is in default thereunder, except where the failure to hold or maintain such licenses, franchises, permits, certificates, approvals or other similar authorisations or a default thereunder would not have a material adverse effect on the Westfield business, and none of such material licenses, franchises, permits, certificates, approvals or other similar authorisations will be terminated or materially adversely modified pursuant to their terms or applicable law as a result of the Transaction or the Schemes;

(g) (Intellectual property):

- (i) the Westfield Group owns all right, title and interest in all registrations and applications for the "Westfield" brand name in connection with its ownership and operation of real estate malls in all jurisdictions in which it operates and in such other jurisdictions in which it reasonably considers that brand name should be protected; and
- (ii) except as Disclosed the Westfield Group has granted no licence to any Third Party to use the "Westfield" brand;
- (h) (Ordinary course) except as Disclosed during the 12 months prior to the date of this document, the Westfield Group has conducted its core mall business in the ordinary course and in substantially the same manner in which it had been conducted in the prior 12 month period (having regard to changes in activities caused by its property development activities, acquisitions and disposals) and has not engaged in any material new business activities outside its core mall business activities and the business of OneMarket;
- (i) (Restrictions on business activities) except as Disclosed there is no agreement or judgment, ruling, order or decree binding on or applicable to it or its assets or any Westfield Group entity that has or would reasonably be likely to have the effect of materially prohibiting, restricting or impairing after the Implementation Date the Westfield Group from entering any new country or territory;

- (Indebtedness) except as Disclosed: (j)
 - no Westfield Group entity has incurred or agreed to incur any (i) financial indebtedness material to the Westfield Group as a whole from a Third Party financier under any bank facility except for the refinancing of any Westfield Debt Facility or of amounts of less than \$5,000,000 (and excluding any such indebtedness, arrangement or financial accommodation (irrespective of what form that accommodation takes) to or among members of the Westfield Group); and
 - (ii) other than the Transaction, no event has occurred which would entitle any Third Party to require the repayment of any material borrowings of the Westfield Group unless that repayment has been waived by the Third Party or is not being enforced by the Third Party:
- (k) (validity of obligations) subject to clause 2.2, its obligations under this document are valid and binding and are enforceable against it in accordance with its terms;
- (I) (Westfield Information) the Westfield Information provided in accordance with this document:
 - (i) has been prepared and included in the Scheme Booklet in good faith and on the understanding that Unibail and each of its directors, officers and employees have relied on that information for the purposes of considering and approving the Unibail Information in the Scheme Booklet, New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting;
 - (ii) included in the Scheme Booklet is not misleading or deceptive and does not contain any material omission having regard to applicable disclosure requirements and complies in all material respects with the requirements of the Corporations Act, the ASX Listing Rules and all relevant regulatory guides and other guidelines and requirements of ASIC; and
 - (iii) included in the New Unibail Shares Listing Prospectus, the New Unibail Stapled Securities Listing Prospectus, the Unibail Convening Notice and the Unibail Notice of Meeting does not contain any material statement which is misleading or deceptive nor contain any material omission having regard to applicable disclosure requirements and complies in all material respects with all applicable laws and requirements of applicable Regulatory Authorities;
- (m) (continuous disclosure) Westfield is not in breach of its continuous disclosure obligations under the ASX Listing Rules and is not relying on the carve-out in ASX Listing Rule 3.1A to withhold any information from disclosure (other than the transaction contemplated by this document);
- (n) (No Unibail securities) Westfield has no Relevant Interest in Unibail Shares and neither it nor any of its Associates is a party to any agreement, arrangement or understanding the economic effect of which is to confer rights equivalent to, or substantially equivalent to, the acquisition, holding or disposal of Unibail Shares or a Relevant Interest in Unibail Shares (including any cash settled equity swap or similar derivative relating to any Unibail Shares);

- (financial statements) the audited consolidated financial statements of (o) the Westfield Group for the financial year ended 31 December 2016 and for the half year ended 30 June 2017 (and, when they are available, the audited consolidated financial statements of the Westfield Group for the financial year ending 31 December 2017) (together "Westfield Financial Reports"):
 - (i) have been prepared in accordance with and comply with Australian International Financial Reporting Standards (applied on a consistent basis during the periods involved) and all requirements of the Corporations Act and requirements of applicable Regulatory Authorities; and
 - (ii) provide a true and fair view of the financial position and performance of the Westfield Group as at the dates they were prepared:
- (p) (No Undisclosed Liabilities) except as Disclosed no member of the Westfield Group has any indebtedness or liabilities required under Australian International Financial Reporting Standards to be reflected on a balance sheet or the notes thereto, other than those (i) specifically reflected in, fully reserved against or otherwise described in the Westfield Financial Reports or the notes thereto; (ii) incurred under this document or in connection with the Transaction or the Schemes; or (iii) that would constitute a Westfield Material Adverse Effect:
- (q) (opinions) any statement of opinion or belief contained in the Westfield Information is honestly held and there are reasonable grounds for holding the opinion or belief;
- (r) (provision of information to Independent Expert) all information provided by or on behalf of Westfield to the Independent Expert to enable the Independent Expert's Report to be prepared and completed will be provided in good faith and on the understanding that the Independent Expert will rely upon that information for the purpose of preparing the Independent Expert's Report;
- (s) (Securities) Westfield's issued securities as at the date of this document
 - (i) 2,078,089,686 Westfield Stapled Securities; and
 - 22,602,950 Employee Rights,
- (t) (No agreements to issue securities) except as Disclosed
 - Westfield has not issued or agreed to issue any other securities (i) or instruments which are still outstanding and which may convert into or be exchanged for shares or units in Westfield (as applicable) or Westfield Stapled Securities; and
 - there are no pre-emptive or other outstanding rights, options, (ii) warrants, conversion rights, appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate Westfield to issue or sell any shares, units or other equity interests in Westfield or any securities or obligations convertible or exchangeable into or exercisable for, or giving any person a right to subscribe for or acquire, any shares, units or other equity interests in Westfield, and no securities or obligations evidencing such rights are authorised, issued or outstanding;

- (Insolvency event) no material Westfield Group entity is Insolvent; (u)
- (v) (compliance) except for circumstances that would not be expected to have a material adverse effect on the business of the Westfield Group or its financial position or performance (when taken as a whole) or as dealt with in clause 13.1(cc):
 - each member of the Westfield Group has complied in all (i) material respects with all laws and regulations applicable to them and orders of Regulatory Authorities having jurisdiction over them and have all material licenses, authorisations and permits necessary for them to conduct the business of the Westfield Group as presently being conducted; and
 - (ii) without limiting the generality of the foregoing or any other representation or warranty in this document, no member of the Westfield Group has to its knowledge any customer, supplier or distributor relationship with, or is a party to any contract with, any person:
 - (A) organised or domiciled in a country subject to comprehensive territorial sanctions as determined by the U.S. Department of the Treasury (including any Governmental Body within any such country); or
 - (B) that is the target of any sanction administered by the U.S. Department of the Treasury, the European Union (including under Council Regulation (EC) No. 194/2008), the United Nations Security Council or Her Majesty's Treasury, including any person that appears on the Specially Designated Nationals and Blocked Persons List of the Office of Foreign Assets Controls in the United States Department of the Treasury or in the Annexes to the United States Executive Order 13224 -Blocking Property and Prohibiting Transactions with Person Who Commit, Threaten to Commit, or Support Terrorism:
- (w) (No breach of contract) except as Disclosed or for circumstances that would not have a material adverse effect on the overall Westfield Group business or financial position or performance of the Westfield Group when taken as a whole:
 - neither Westfield nor any of its material Subsidiaries is in default (i) under any Westfield Significant Contract; and
 - neither the entry into this document by Westfield, nor the (ii) execution of its terms by Westfield will constitute (whether immediately of following the giving of notice of the lapse of time) an event of default, prepayment event, change of control or similar event, or give another party thereto a termination right or right to accelerate any right or obligation, or right to receive any payment, under any Westfield Significant Contract;
- (x) (Westfield Disclosure Materials) Westfield has collated and prepared all of the Westfield Disclosure Materials requested by Unibail in good faith;
- (Conduct of Business) Westfield is not in breach of any of its (y) obligations under clause 8.2.

(Environmental) except as Disclosed, there is no contamination from (z) exposure to hazardous substances in violation of applicable law affecting any of the real property occupied by Westfield Group entities that to the knowledge of Westfield would be reasonably likely to have a material adverse effect on the overall Westfield Group business or financial position when taken as a whole;

(Employment) (aa)

- (i) Except as Disclosed or for arrangements concerning employees whose responsibilities are not material to the conduct of Westfield Groups affairs, no Westfield Group entity is a party to any employee collective bargaining agreement with a trade union or industrial organisation, or group of employees in respect of the employees of the Westfield Group and their employment and no industrial awards or workplace agreements apply to any employees of a Westfield Group entity; and
- (ii) as of the date of this document, no Westfield Group entity has been involved in any material industrial law-related proceedings or material dispute with any union at any time within the 6 months preceding the date of this document that would reasonably be likely to give rise to any material adverse effect on the overall Westfield Group business or financial position when taken as a whole:
- (bb) (Employee Rights) in respect of all Employee Rights on issue at the date of this document, Westfield has Disclosed to Unibail a true and complete copy of each plan or policy pursuant to which those Employee Rights were issued and are currently governed; and
- (cc) (Unlawful payments) for the three year period prior to the date of this document, no Westfield Group entity or, to Westfield's knowledge, any of its respective officers, directors, employees, agents or representatives has, directly or indirectly, in connection with the business of the Westfield Group:
 - (i) requested, received, made, offered or promised to make or offer any unlawful payment, loan or transfer of anything of value or advantage to or for the benefit of or from any person, including any government official, candidate for public office, political party or political campaign;
 - requested, received, paid, offered or promised to make or offer (ii) any bribe, payoff, influence payment, kickback, unlawful rebate, or other similar unlawful payment of any nature;
 - (iii) requested, received, made, offered or promised to make or offer any unlawful contributions (including political or charitable contributions), gifts, entertainment or other unlawful expenditures;
 - (iv) established or maintained any unlawful fund of corporate monies or other properties:
 - created or caused the creation of any false or inaccurate books (v) and records of the Westfield Group or any of its members related to any of the foregoing; or

(vi) otherwise violated any provision of the FCPA, the UK Bribery Act, or any other local or international anti-corruption or anti-bribery law applicable to the Westfield Group.

and, so far as Westfield is aware, there are no, and no such person has received any notice of any, Claims threatening the commencement of any litigation, prosecution, mediation, arbitration, investigation or other proceeding in respect of such matters;

(dd) (Insurance)

- (i) Westfield Group maintains insurance in such amounts and against accident, physical loss or damage, third party liability (including general liability and professional negligence) and all other normal risks incidental to the operation of the business of the Westfield Group as required by law or is otherwise reasonably required and can be obtained on reasonable commercial terms for a group with activities of the nature of that undertaken by Westfield Group ("Insurance");
- each material policy of Insurance is currently in full force and effect and all applicable premiums that are due and payable have been paid;
- (iii) Westfield Group has not received any current written notice that it is not in compliance in all material respects with the current insurance policies covering the Westfield Group; and
- (iv) as of the date of this document, no written notice of cancellation, termination or revocation of a material policy of insurance has been received by a member of the Westfield Group that remains outstanding;
- (ee) (Taxes) neither Westfield nor any of its Subsidiaries:
 - is currently the subject of a material Tax audit or examination or other administrative or judicial proceeding arising therefrom, and to Westfield's knowledge, no such audit, examination or proceeding is threatened or contemplated, or
 - (ii) has received from any Taxing Authority any written notice of a material proposed adjustment, deficiency or underpayment of Taxes which has not been satisfied by payment or been withdrawn that would reasonably be likely to give rise to any material adverse effect on the overall Westfield Group business or financial position,

and all material Tax Returns required to be filed by any Westfield Group entity, have been timely filed and all Taxes shown as due and payable on such Tax Returns, and all material amounts of Taxes required to have been paid by Westfield and its subsidiaries have been paid or appropriate reserves have been recorded in the financial statements of Westfield;

(ff) (REIT status) Westfield America, Inc. elected to be taxed as a real estate investment trust (a "REIT") under the U.S. Internal Revenue Code of 1986, as amended (the "Code"), commencing with its taxable year ended December 31, 1994. Commencing with Westfield America Inc's taxable year ended December 31, 1994 and through the date of this warranty, Westfield America has been organized in conformity with the requirements for qualification as a REIT, and its planned method of

operation, and its actual method of operation from February 12, 1994 through the date of this warranty enable it to meet the requirements for qualification and taxation as a REIT under the Code.

(prior Claims) Except as Disclosed: (gg)

- neither ASIC nor ASX (as applicable) has made a determination (i) against any Westfield Group entity for any contravention of the requirements of the Corporations Act or the ASX Listing Rules or any rules, regulations or regulatory guides under the Corporations Act or the ASX Listing Rules;
- (ii) Westfield is not aware of any litigation, prosecution, mediation, arbitration, regulatory investigation or other proceeding in respect of the Westfield that is material to Westfield Group;
- (iii) the Westfield Group has not received any written notice of Claim threatening the commencement of any litigation prosecution, mediation, arbitration, regulatory investigation or other proceeding in respect of any of the business operated by the Westfield Group that is material to Westfield Group; and
- (iv) there are no unfulfilled or unsatisfied judgments relating to the business outstanding against the Westfield Group that is material to Westfield Group,

in each case excluding any Claim to the extent that the Claim is covered by insurance where the insurer has accepted liability for the Claim in accordance with the terms of the relevant policy; and

(hh) (Other Claims) Except as Disclosed:

- (i) neither ASIC nor ASX (as applicable) has made a determination against any Westfield Group entity for any contravention of the requirements of the Corporations Act or the ASX Listing Rules or any rules, regulations or regulatory guides under the Corporations Act or the ASX Listing Rules;
- (ii) Westfield is not aware of any material litigation, prosecution, mediation, arbitration, regulatory investigation or other proceeding in respect of the Westfield;
- (iii) the Westfield Group has not received any written notice of Claim threatening the commencement of any material litigation prosecution, mediation, arbitration, regulatory investigation or other proceeding in respect of any of the business operated by the Westfield Group; and
- (iv) there are no material unfulfilled or unsatisfied judgments relating to the business outstanding against the Westfield Group.

in each case that is not covered by insurance where the insurer has accepted liability for the Claim in full (in the case of a Claim) or would not constitute a Westfield Material Adverse Effect.

13.2 Timing of Westfield's representations and warranties

Each of the representations and warranties in clause 13.1 are given on each date from the date of this document until and including 8:00am on the Second Court Date except in respect of the representations and warranties:

- (a) in clause 13.1(s) and 13.1(gg), which are given on the date of this document only:
- in clause 13.1(I), which are given on the date of the document in which (b) the relevant Westfield Information appears; and
- in clause 13.1(o) with respect to the audited consolidated financial (c) statements for the year ended 31 December 2017, from the date of public release of those statements.

13.3 Unibail's representations and warranties

Unibail represents and warrants to Westfield that each of the following statements is true and correct as at the date of this document and as at the date specified in clause 13.4 of this document:

- (a) (status) it has been duly incorporated or formed in accordance with, and is validly existing under, the laws of its place of incorporation;
- (power) it has power to enter into this document, to comply with its (b) obligations under it and exercise its rights under it;
- (no contravention) the entry by it into, its compliance with its obligations (c) and the exercise of its rights under, this document do not and will not conflict with:
 - (i) its constituent and organisational documents or cause a limitation on its powers or the powers of its directors to be exceeded; or
 - any law or treaty or a judgment, ruling, order or decree binding (ii) on or applicable to it or its assets;

(d) (authorisations)

- (i) other than the approvals specified in the Conditions Precedent, it has in full force and effect each authorisation necessary for it to enter into this document, to comply with its obligations and exercise its rights under it, and to allow them to be enforced; and
- (ii) the Unibail Group holds all material licenses, franchises, permits, certificates, approvals or other similar authorisations issued by applicable Regulatory Authorities necessary for the lawful conduct of its businesses as presently conducted as of the date of this document, and, to Unibail's knowledge, such licenses, franchises, permits, certificates, approvals or other similar authorisations are valid and in full force and effect, no Unibail Group entity is in default thereunder, except where the failure to hold or maintain such licenses, franchises, permits, certificates, approvals or other similar authorisations or a default thereunder would not have a material adverse effect on the Unibail business, and none of such material licenses, franchises. permits, certificates, approvals or other similar authorisations will be terminated or materially adversely modified pursuant to their terms or applicable law as a result of the Transaction or the Schemes:
- (Ordinary course) except as Disclosed during the 12 months prior to the (e) date of this document, the Unibail Group has conducted its core mall business in the ordinary course and in substantially the same manner in which it had been conducted in the prior 12 month period (having regard

to changes in activities caused by its property development activities, acquisitions and disposals) and has not engaged in any material new business activities:

(f) (Indebtedness) Except as Disclosed

- no Unibail Group entity has incurred or agreed to incur any (i) financial indebtedness material to the Unibail Group as a whole from a Third Party financier under any bank facility except for purposes of financing the Transaction, the refinancing of any Unibail Debt Facility or of amounts of less than \$5,000,000 (and excluding any such indebtedness, arrangement or financial accommodation (irrespective of what form that accommodation takes) to or among members of the Unibail Group); and
- (ii) other than the Transaction, no event has occurred which would entitle any Third Party to require the repayment of any material borrowings of the Unibail Group unless that repayment has been waived by the Third Party or is not being enforced by the Third Partv:
- (g) (validity of obligations) subject to clause 2.2, its obligations under this document are valid and binding and are enforceable against it in accordance with its terms;
- (h) (Unibail Information) the Unibail Information provided in accordance with this document and included in the Scheme Booklet, as at the date of the Scheme Booklet, will not contain any material statement which is misleading or deceptive nor contain any material omission having regard to applicable disclosure requirements and will comply in all material respects with the requirements of the Corporations Act, the ASX Listing Rules and all relevant regulatory guides and other guidelines and requirements of ASIC;
- (continuous disclosure) Unibail is not in breach of its obligations to (i) disclose information to the public under all applicable laws, regulations and rules of any stock exchange, including the AMF, Euronext Paris and Euronext Amsterdam;
- (Unibail Reports) since 1 January 2016 (and, when available, for the (j) financial year ending 31 December 2017), the Unibail Reports were, to the extent required under applicable law, filed in a timely manner, and are in material compliance with all applicable laws and other requirements of Regulatory Authorities. As of their respective dates (or if amended prior to the date of this document, as of the date of such amendment), the Unibail Reports which were required to be filed with any relevant Regulatory Authority complied in all material respects with the requirements under applicable law and the requirement of any Regulatory Authority regarding the accuracy and completeness of the disclosures contained in the Unibail Reports;
- (k) (financial statements) the Unibail Annual & Sustainable Development Report for the financial year ended 31 December 2016 (and, when available, for the financial year ending 31 December 2017):
 - fairly present in all material respects the assets and liabilities (i) and financial position of Unibail and of the results of its operations as at the date they were prepared; and
 - (ii) have been or will be, as the case may be, prepared in accordance with International Financial Reporting Standards

applied on a consistent basis during the periods involved (except as may be indicated in the notes to the financial statements for normal year-end adjustments that are not material in amount or nature):

- (No Undisclosed Liabilities) except as Disclosed no member of the (I) Unibail Group has any indebtedness or liabilities required under International Financial Reporting Standards to be reflected on a balance sheet or the notes thereto, other than those (i) specifically reflected in. fully reserved against or otherwise described in the Unibail Annual & Sustainable Development Reports or the notes thereto; (ii) incurred under this document or in connection with the Transaction or the Schemes; or (iii) that would not constitute a Unibail Material Adverse Effect;
- (m) (opinions) any statement of opinion or belief contained in the Unibail Information is honestly held and there are reasonable grounds for holding the opinion or belief:
- (n) (no dealing with Westfield Securityholders) neither it nor any of its Associates has any agreement, arrangement or understanding with any Westfield Securityholder under which that Westfield Securityholder (or an Associate of that Westfield Securityholder) would be entitled to receive consideration for their Westfield Stapled Securities different from the Scheme Consideration as contemplated in this document or under which the Westfield Securityholder agrees to vote in favour of the Transaction and the Schemes or against any Competing Proposal;
- (provision of information to Independent Expert) all information (o) provided by or on behalf of Unibail to the Independent Expert to enable the Independent Expert's Report to be prepared and completed will be provided in good faith and on the understanding that the Independent Expert will rely upon that information for the purpose of preparing the Independent Expert's Report;
- (cash reserves on the Implementation Date) it has a reasonable basis (p) to expect that it will, by the Implementation Date, have available to it sufficient cash reserves (whether from internal cash reserves or external funding arrangements, including equity and debt financing or a combination of both) to satisfy Unibail's obligations to discharge the Scheme Consideration in accordance with its obligations under this document, the Schemes and the Deed Poll;
- (unconditional cash reserves on the Second Court Date and the (q) Implementation Date) by 8.00am on the Second Court Date and on the Implementation Date, Unibail will have available to it on an unconditional basis (other than, on the Second Court Date, conditions relating to the approval of the Scheme Court and other conditions within the sole control of Unibail) sufficient cash reserves (whether from internal cash reserves or external funding arrangements, including equity and debt financing or a combination of both) to satisfy Unibail's obligations to discharge the Scheme Consideration in accordance with its obligations under this document, the Transaction and the Schemes and the Deed Poll:
- (Securities) Except as Disclosed: (r)
 - (i) Unibail's issued and outstanding securities as at the date of this document are:
 - (A) 99,844,259 Unibail Shares;

- 1.260 bonds redeemable in cash and/or in new and/or (B) existing shares (obligations à option de remboursement en numéraire et/ou actions nouvelles et/ou existantes) issued on 19 September 2012:
- (C) 1,735,749 bonds redeemable in cash and/or in new and/or existing shares (obligations à option de remboursement en numéraire et/ou actions nouvelles et/ou existantes) issued on 25 June 2014; and
- (D) 1,441,462 bonds redeemable in cash and/or in new and/or existing shares (obligations à option de remboursement en numéraire et/ou actions nouvelles et/ou existantes) issued on 15 April 2015;
- (ii) in addition, 2.140.946Unibail Shares may be issuable pursuant to Unibail stock-option plans, and 104,269Unibail Shares may be issuable pursuant to Unibail performance shares plans;
- Unibail has not issued or agreed to issue any other securities or (iii) instruments which are still outstanding and which may convert into or be exchanged for shares in Unibail or any Subsidiaries: and
- (iv) there are no pre-emptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate Unibail to issue or sell any shares of capital stock or other equity interests in Unibail or any securities or obligations convertible or exchangeable into or exercisable for, or giving any person a right to subscribe for or acquire, any shares of capital stock or other equity interests in Unibail, and no securities or obligations evidencing such rights are authorised, issued or outstanding:
- (s) (No Westfield securities) Except as Disclosed Unibail has no Relevant Interest in Westfield Stapled Securities and neither it nor any of its Associates is a party to any agreement, arrangement or understanding the economic effect of which is to confer rights equivalent to, or substantially equivalent to, the acquisition, holding or disposal of Westfield Stapled Securities or a Relevant Interest in Westfield Stapled Securities (including any cash settled equity swap or similar derivative relating to any Westfield Stapled Securities);
- (t) (Insolvency event) no material Unibail Group entity is Insolvent;
- (REIT status) commencing with its taxable year ended December 31, (u) 2003, (i) Unibail SE has been organized and has operated in conformity with the requirements for qualification as a "société d'investissements immobiliers cotée" (SIIC) under article 208 C, II of the French Tax Code, as amended and (ii) its planned method of operation and its actual method of operation from January 1, 2003 through the date of this warranty enables it to meet the requirements for qualification and taxation as a SIIC with respect to the assets and activities of Unibail SE that are reported to the French tax authorities as included in the SIIC regime;
- (v) (compliance) except for circumstances that would not be expected to have a material adverse effect on the business of the Unibail Group or its

financial position or performance (when taken as a whole) or as dealt with in clause 13.3(w):

- (i) each member of the Unibail Group has complied in all material respects with all laws and regulations applicable to them and orders of Regulatory Authorities having jurisdiction over them and have all material licenses, authorisations and permits necessary for them to conduct the business of the Unibail Group as presently being conducted; and
- (ii) without limiting the generality of the foregoing or any other representation or warranty in this document, no member of the Unibail Group has to its knowledge any customer, supplier or distributor relationship with, or is a party to any contract with, any person:
 - (A) organised or domiciled in a country subject to comprehensive territorial sanctions as determined by the U.S. Department of the Treasury (including any Governmental Body within any such country); or
 - (B) that is the target of any sanction administered by the U.S. Department of the Treasury, the European Union (including under Council Regulation (EC) No. 194/2008), the United Nations Security Council or Her Majesty's Treasury, including any person that appears on the Specially Designated Nationals and Blocked Persons List of the Office of Foreign Assets Controls in the United States Department of the Treasury or in the Annexes to the United States Executive Order 13224 -Blocking Property and Prohibiting Transactions with Person Who Commit, Threaten to Commit, or Support Terrorism; and
- (w) (Unlawful payments) for the three year period prior to the date of this document, no Unibail Group entity or, to Unibail's knowledge, any of its respective officers, directors, employees, agents or representatives has, directly or indirectly, in connection with the business of the Unibail Group:
 - requested, received, made, offered or promised to make or offer (i) any unlawful payment, loan or transfer of anything of value or advantage to or for the benefit of or from any person, including any government official, candidate for public office, political party or political campaign;
 - (ii) requested, received, paid, offered or promised to make or offer any bribe, payoff, influence payment, kickback, unlawful rebate, or other similar unlawful payment of any nature;
 - (iii) requested, received, made, offered or promised to make or offer any unlawful contributions (including political or charitable contributions), gifts, entertainment or other unlawful expenditures:
 - (iv) established or maintained any unlawful fund of corporate monies or other properties;
 - (v) created or caused the creation of any false or inaccurate books and records of the Unibail Group or any of its members related to any of the foregoing; or

otherwise violated any provision of FCPA, the UK Bribery Act. or (vi) any other local or international anti-corruption or anti-bribery law applicable to the Unibail Group,

and, so far as Unibail is aware, there are no, and no such person has received any notice of any. Claims threatening the commencement of any litigation, prosecution, mediation, arbitration, investigation or other proceeding in respect of such matters.

13.4 Timing of Unibail's representations and warranties

Each of the representations and warranties in clause 13.3 are given on each date from the date of this document until and including 8:00am on the Second Court Date except in respect of the representations and warranties:

- in clauses 13.3(e) and 13.3(r), which are given on the date of this (a) document only:
- (b) in clause 13.3(h), which are given on the date of this document in which the relevant Westfield Information appears; and
- in clause 13.1(I) with respect to the 31 December 2017 Unibail Reports, (c) from the date of public release of those statements.

13.5 **Notification**

Each party will promptly advise the other parties in writing if it becomes aware of a fact, matter or circumstances that constitutes or may constitute a breach of any the representations or warranties given by it under this clause 13.

Termination 14

14.1 **Termination events**

This document may be terminated:

- (a) (End Date) by either party, if the Schemes have not become Effective on or before the End Date:
- (b) (lack of support) by Unibail at any time prior to 8.00am on the Second Court Date if the majority of a Westfield Board changes its recommendation to the Scheme Participants that they vote in favour of the Schemes, including any adverse modification to the recommendation, or otherwise makes a public statement indicating that it no longer supports the Schemes;
- (change of recommendation) by Westfield at any time prior to 8:00am (c) on the Second Court Date, if the majority of the Unibail Board changes its recommendation given in accordance with clause 5.4(a) or 5.5(a), including an adverse modification to the recommendation or otherwise makes a public statement indicating that it no longer supports the Transaction;
- (d) (material breach) by either Unibail or Westfield at any time prior to 8.00am on the Second Court Date, if the other is in material breach of a term of this document (including any representation and warranty not being true and correct or undertaking any action in breach of clause 8.2), taken in the context of the Transaction as a whole, provided that Unibail or Westfield (as the case may be) has, if practicable, given notice to the other setting out the relevant circumstances and the relevant

circumstances continue to exist 5 Business Days (or any shorter period ending at 8.00am on the Second Court Date) after the time such notice is given;

- (Competing Proposal) by Westfield if the Westfield Board has changed (e) its recommendation in circumstances permitted under clause 6.1 and has paid any applicable the Break Fee;
- (f) (Bid for Unibail) by Unibail if the Unibail Board has withdrawn, changed or qualified its recommendation and paid the Reverse Break Fee to Westfield:
- (g) (Consultation or appeal failure) in accordance with and pursuant to:
 - clause 3.7(a)(i); (i)
 - clause 3.7(a)(ii); and (ii)
 - (iii) clause 5.11; or
- (h) (agreement) if agreed to in writing by Unibail and Westfield.

Termination 14.2

Where a party has a right to terminate this document, that right for all purposes will be validly exercised if the party delivers a notice in writing to the other party stating that it terminates this document.

14.3 **Effect of Termination**

If this document is terminated by either party, or if this document otherwise terminates in accordance with its terms, then in either case all further obligations of the parties under this document, other than the obligations under the Preserved Clauses, will immediately cease to be of further force and effect without further liability of any party to the other, provided that nothing in this clause releases any party from liability for any pre-termination breach of this document.

14.4 **Damages**

In addition to the right of termination under clause 14.1, the non-defaulting party is entitled to damages for Losses suffered by it and expenses incurred by it as a result of the breach of the terms of this document.

14.5 Non-application to ERB termination

This clause does not apply to a termination under clause 2.2(g).

15 **Limitation of Liability**

(a) WAML enters into this document only in its capacity as responsible entity of WAT and WFDT and in no other capacity. A liability arising under or in connection with the document is limited and can be enforced against WAML only to the extent to which it can be satisfied out of property of the WAT or WFDT, as relevant, and for which, WAML is actually indemnified for the liability. This limitation of WAML's liability applies despite any other provisions of this document and extends to all liabilities and obligations of WAML in any way connected with any representation,

© King & Wood Mallesons 77 Implementation Agreement

- warranty, conduct, omission, agreement or transaction related to this document, any other document in connection with it. WAT or WFDT.
- (b) The other parties may not sue WAML in any capacity other than as responsible entity of WAT or WFDT, as applicable, including seeking the appointment of a receiver, a liquidator, an administrator or similar person to WAML or prove in any liquidation, administration or arrangement of or affecting WAML (except in relation to property of WAT or WFDT).
- The limitation of liability provisions shall not apply to any obligation or (c) liability of WAML to the extent that it is not or cannot be satisfied because, under this document or any other document in connection with it, or by operation of law, there is a reduction in the extent of WAML's indemnification out of the assets of WAT or WFDT, as applicable, or no such right of indemnification exists, as a result of WAML's fraud, negligence or breach of trust.
- (d) It is also acknowledged that a breach of an obligation imposed on, or a representation or warranty given by, WAML under or in connection with this or any other document in connection with it will not be considered a breach of trust by WAML unless WAML has acted with negligence, or without good faith, in relation to the breach.

16 **Public announcements**

16.1 **Public announcement of Schemes**

Immediately after signing this document, Westfield and Unibail must issue a joint public announcement of the proposed Transaction and Schemes in the form contained in Annexure A.

16.2 Required disclosure

Where a party is required by any applicable laws or regulations or any ASX Listing Rule or Euronext Rule to make any announcement or make any disclosure in connection with the Transaction and the Schemes, it must use all reasonable endeavours, to the extent possible, to consult with the other party prior to making the relevant disclosure.

16.3 Other announcements

Subject to clauses 16.1 and 16.2, no party may make any public announcement or disclosure in connection with the Transaction and the Schemes (excluding any disclosure to a Regulatory Authority undertaken in accordance with clause 3) other than in a form approved by each party (acting reasonably). Each party will use all reasonable endeavours to provide such approval as soon as practicable.

17 Confidential Information

Each party acknowledges and agrees that it continues to be bound by the Confidentiality Deed in respect of all information received by it from the other party on, before or after the date of this document.

18 Notices and other communications

18.1 **Form**

Unless this document expressly states otherwise, all notices, demands, certificates, consents, approvals, waivers and other communications in connection with this document must be in writing and signed by the sender (if an individual) or an Authorised Officer of the sender.

All communications (other than email communications) must also be marked for the attention of the person referred to in the Details (or, if the recipient has notified otherwise, then marked for attention in the way last notified).

Email communications must state the first and last name of the sender and are taken to be signed by the named sender.

18.2 **Delivery**

Communications must be:

- (a) left at the address referred to in the Details;
- sent by regular ordinary post (airmail if appropriate) to the address (b) referred to in the Details; or
- (c) sent by email to the address referred to in the Details.

If the intended recipient has given notice of changed contact details, then communications must be sent to the changed contact details.

18.3 When effective

Communications take effect from the time they are received or taken to be received under clause 18.4 (whichever happens first) unless a later time is specified in the communication.

18.4 When taken to be received

Communications are taken to be received:

- (a) if sent by post, 6 Business Days after posting (or 10 days after posting if sent from one country to another);
- (b) if sent by email:
 - (i) when the sender receives an automated message confirming delivery; or
 - (ii) 4 hours after the time sent (as recorded on the device from which the sender sent the email) unless the sender receives an automated message that delivery failed,

whichever happens first.

18.5 Receipt outside business hours

Despite anything else in this clause 18, if communications are received or taken to be received under clause 18.4 after 5.00pm on a Business Day or on a non-Business Day, they are taken to be received at 9.00am on the next Business Day. For the purposes of this clause, the place in the definition of Business Day

is taken to be the place specified in the Details as the address of the recipient and the time of receipt is the time in that place.

19 **GST and VAT**

19.1 **Definitions and interpretation**

For the purposes of this clause:

- (a) "GST Act" means the A New Tax System (Goods and Services Tax) Act 1999 (Cth) (Australia);
- (b) a term which has a defined meaning in the GST Act has the same meaning when used in this clause, unless the contrary intention appears; and
- each periodic or progressive component of a supply to which section (c) 156-5(1) of the GST Act applies will be treated as if it were a separate supply.

19.2 **GST** exclusive and VAT exclusive

Unless this document expressly states otherwise, all consideration to be provided under this document is exclusive of GST and VAT (as applicable).

19.3 Payment of GST and VAT

- If GST or VAT (as applicable) is payable, or notionally payable, on a (a) supply in connection with this document, the party providing the consideration for the supply agrees to pay to the supplier an additional amount equal to the amount of GST or VAT (as applicable) payable on that supply ("GST Amount" or "VAT Amount", as applicable).
- (b) Subject to the prior receipt of a valid tax invoice, the GST Amount or VAT Amount (as applicable) is payable at the same time as the GSTexclusive or VAT-exclusive (as applicable) consideration for the supply, or the first part of the GST-exclusive or VAT-exclusive (as applicable) consideration for the supply (as the case may be), is payable or is to be provided.
- (c) This clause does not apply to the extent that the consideration for the supply is expressly stated to include GST or VAT (as applicable) or the supply is subject to a reverse-charge.

19.4 **Adjustment events**

If an adjustment event arises for a supply made in connection with this document, the GST Amount or VAT Amount (as applicable) must be recalculated to reflect that adjustment. The supplier or the recipient (as the case may be) agrees to make any payments necessary to reflect the adjustment and the supplier agrees to issue an adjustment note.

19.5 Reimbursements

Any payment, indemnity, reimbursement or similar obligation that is required to be made in connection with this document which is calculated by reference to an amount paid by another party must be reduced by the amount of any input tax credits which the other party (or the representative member of any GST or VAT (as applicable) group of which the other party is a member) is entitled. If the

reduced payment is consideration for a taxable supply, clause 19.3 will apply to the reduced payment.

20 Costs

20.1 Costs

The parties agree to pay their own Costs in connection with the preparation, negotiation, execution and completion of this document, except for amounts covered by clause 20.2.

20.2 Stamp duty and registration fees

Unibail:

- agrees to pay or reimburse all stamp duty, registration fees, financial (a) transaction tax and similar taxes payable or assessed as being payable in connection with this document or any other transaction contemplated by this document (including any fees, fines, penalties and interest in connection with any of those amounts); and
- (b) indemnifies (on an after-Tax basis) Westfield against, and agrees to reimburse and compensate it for, any liability in respect of stamp duty, registration fees, financial transaction tax and similar taxes under clause 20.2(a).

Unibail agrees to pay amounts due to Westfield under this clause within 3 Business Days of demand from Westfield.

21 General

21.1 Variation and waiver

A provision of this document, or right, power or remedy created under it, may not be varied or waived except in writing signed by the party to be bound.

21.2 Consents, approvals or waivers

By giving any approval, consent or waiver a party does not give any representation or warranty as to any circumstance in connection with the subject matter of the consent, approval or waiver.

21.3 Discretion in exercising rights

Unless this document expressly states otherwise, a party may exercise a right, power or remedy or give or refuse its consent, approval or a waiver in connection with this document in its absolute discretion (including by imposing conditions).

21.4 Partial exercising of rights

Unless this document expressly states otherwise, if a party does not exercise a right, power or remedy in connection with this document fully or at a given time, they may still exercise it later.

21.5 Conflict of interest

Each party may exercise their rights, powers and remedies in connection with this document even if this involves a conflict of duty or they have a personal interest in their exercise.

21.6 Remedies cumulative

The rights, powers and remedies in connection with this document are in addition to other rights, powers and remedies given by law independently of this document.

21.7 Indemnities and reimbursement obligations

Any indemnity, reimbursement or similar obligation in this document.

- (a) is a continuing obligation despite the satisfaction of any payment or other obligation in connection with this document, any settlement or any other thing;
- (b) is independent of any other obligations under this document; and
- (c) continues after this document, or any obligation arising under it, ends.

It is not necessary for a party to incur expense or make payment before enforcing a right of indemnity in connection with this document.

21.8 Inconsistent law

To the extent the law permits, this document prevails to the extent it is inconsistent with any law.

21.9 Supervening law

Any present or future law which operates to vary the obligations of a party in connection with this document with the result that another party's rights, powers or remedies are adversely affected (including, by way of delay or postponement) is excluded except to the extent that its exclusion is prohibited or rendered ineffective by law.

21.10 Counterparts

This document may consist of a number of copies, each signed by one or more parties to it. If so, the signed copies are treated as making up a single document.

21.11 Entire agreement

This document constitutes the entire agreement of the parties about its subject matter and supersedes all previous agreements, understandings and negotiations on that subject matter.

21.12 Further steps

Each party agrees to do anything (such as obtaining consents, signing and producing documents, producing receipts and getting documents completed and signed), which the other party asks and considers necessary to:

- bind the party and any other person intended to be bound under this (a) document; or
- (b) show whether the party is complying with this document.

21.13 No liability for loss

Unless this document expressly states otherwise, a party is not liable for any loss, liability or costs arising in connection with the exercise or attempted exercise of, failure to exercise, or delay in exercising, a right, power or remedy in connection with this document.

21.14 Severability

If the whole or any part of a provision of this document is void, unenforceable or illegal in a jurisdiction it is severed for that jurisdiction. The remainder of this document has full force and effect and the validity or enforceability of that provision in any other jurisdiction is not affected. This clause has no effect if the severance alters the basic nature of this document or is contrary to public policy.

21.15 Rules of construction

No rule of construction applies to the disadvantage of a party because that party was responsible for the preparation of, or seeks to rely on, this document or any part of it.

21.16 Assignment

A party may not assign or otherwise deal with its rights under this document or allow any interest in them to arise or be varied without the consent of the other party.

21.17 Enforceability

For the purpose of this document:

- (a) Westfield is taken to be acting as agent and trustee on behalf of and for the benefit of all Westfield Indemnified Parties; and
- (b) Unibail is taken to be acting as agent and trustee on behalf of and for the benefit of all Unibail Indemnified Parties.

and all of those persons are to this extent taken to be parties to this document.

21.18 No representation or reliance

Each party acknowledges that:

- (a) no party (nor any person acting on its behalf) has made any representation or other inducement to it to enter into this document, except for representations or inducements expressly set out in this document;
- (b) it does not enter into this document in reliance on any representation or other inducement by or on behalf of any other party, except for any representation or inducement expressly set out in this document; and
- (c) clauses 21.18(a) and 21.18(b) above do not prejudice any rights a party may have in relation to information which had been filed by the other party with a Regulatory Authority.

22 Governing law

22.1 Governing law and jurisdiction

The law in force in the place specified in the Details governs this document. The parties submit to the non-exclusive jurisdiction of the courts of that place.

22.2 Serving documents

Without preventing any other method of service, any document in an action in connection with this document may be served on a party by being delivered or left at that party's address for service of notices under clause 18.2 or with its process agent (as contemplated by clause 22.3).

22.3 Service of process

(a) Without preventing any other mode of service, any document in an action (including, any writ of summons or other originating process or any third or other party notice) may be served on Unibail by being delivered to or left for Unibail at the following address:

Allens Level 28, Deutsche Bank Place 126 Phillip Street Sydney NSW 2000 Australia

(b) Unibail irrevocably appoints Allens as its agent for the service of process in Australia in relation to any matter arising out of this document. If Allens ceases to be able to act as such or have an address in Australia, Unibail agrees to appoint a new process agent in Australia and deliver to Westfield within 2 Business Days a copy of a written acceptance of appointment by the process agent, upon receipt of which the new appointment becomes effective for the purpose of this document. Unibail must inform Westfield in writing of any change in the address of its process agent within 2 Business Days of the change.

EXECUTED as an agreement

Schedule 1 Timetable (clause 5.1)

Event	Date
Lodge Regulator's Draft (and Independent Expert's Report) with ASIC and ASX	Q1 2018
Application in respect of the Scheme Court hearing filed with the Scheme Court, served on ASIC	Q1 2018
First Court Date	Q2 2018
Printing and despatch of Scheme Booklet and other meeting materials to Westfield Securityholders	Q2 2018
Unibail Shareholder Approval is obtained or condition relating to Unibail Shareholder Approval is waived (or deemed waived)	
Scheme Meetings held	Q2 2018
Second Court Date	Q2 2018
Lodge Scheme Court order with ASIC and ASX (Effective Date)	Q2 2018
Execute and lodge Westfield Constitutional Amendments	
Record Date (5.00pm on the date which is the Effective Date plus 4 Business Days or such other date as Westfield and Unibail agree)	Q2 2018
Implementation Date (Record Date plus 5 Business Days)	Q2 2018

Schedule 2 Transaction Steps

Each Transaction Step is to be undertaken in the following order on or before the Implementation Date.

Step	Action	Responsibility
1.	Formation of Newco's	Unibail
	Unibail to incorporate the following entities as direct or indirect Wholly Owned Subsidiaries of Unibail:	
	TH Newco;	
	Newco BI; and	
	US Newco.	
2.	Destapling	Westfield
	WCL Shares, WAT Units and WFDT Units are destapled.	
3.	WAT Trust Scheme	Westfield
	Scheme Participants will transfer their WAT Trust Scheme Units to US Newco in return for the WAT Trust Scheme Consideration, pursuant to the WAT Trust Scheme.	
4.	WFDT Trust Scheme	Westfield
	Scheme Participants will transfer their WFDT Trust Scheme Units to TH Newco in return for the WFDT Trust Scheme Consideration, pursuant to the WFDT Trust Scheme	
5.	WCL Scheme	Westfield
	Scheme Participants (or, in the case of TH Newco Shares, the Third Party Nominee on their behalf) will transfer their WCL Scheme Shares and their TH Newco Scheme Shares to Unibail in return for the Scheme Consideration, pursuant to this Scheme (provided that in respect of Scheme Participants which are Ineligible Foreign Holders, the Sale Agent under the WFDT Trust Scheme will transfer the TH Newco Shares which were issued to that Sale Agent under the WFDT Trust Scheme to Unibail in return for that part of the Scheme Consideration attributable to TH Newco Shares, pursuant to the WFDT Trust Scheme).	
6.	Listing of New Unibail	Unibail
	New Unibail Shares (or New Unibail CDIs) issued as part of the Scheme Consideration, and Newco BI Shares (or Newco BI CDIs) issued as part of the WAT Trust Scheme Consideration, will be stapled together and will trade on Euronext Paris and Euronext Amsterdam and on ASX	

Schedule 3 Newco BI

Newco Bl's share capital will consist of class A and class B shares. Class A shares will be ordinary shares, each carrying equal voting rights and entitlement to dividends. It is envisaged that class B shares would carry limited financial rights compared to class A shares. The financial rights of the class B shares will be finally determined by Unibail (in consultation with Westfield) prior to the First Court Date. The aggregate nominal value of the outstanding class B shares will be 40% of Newco Bl's total issued share capital, enabling Unibail as the holder thereof to cast 40% of the total voting rights in Newco Bl (on the basis that all class A shares can be voted).

© King & Wood Mallesons Implementation Agreement 87

Signing page

DATED: 12 DECEMBER 2017

SIGNED by C. CUVILLIER as authorised representative for UNIBAIL-RODAMCO SE in the presence Signature of witness By executing this document the signatory warrants that the signatory is duly authorised to execute this D. ZEITOUN document on behalf of UNIBAIL-Name of witness (block letters) **RODAMCO SE** SIGNED by as attorney for WESTFIELD **CORPORATION LIMITED** under power of attorney dated 9 December 2017 in the presence of: By executing this document the attorney states that the attorney has Signature of witness received no notice of revocation of the power of attorney

Name of witness (block letters)

Signing page

DATED: 12 DECEMBER 2017	
SIGNED by as authorised representative for UNIBAIL-RODAMCO SE in the presence of:))))))))))
Signature of witness Name of witness (block letters)	 By executing this document the signatory warrants that the signatory is duly authorised to execute this document on behalf of UNIBAIL- RODAMCO SE
FRANK LOWY	
as attorney for WESTFIELD CORPORATION LIMITED under power of attorney dated 9 December 2017 in the presence of Signature of witness	By executing this document the attorney states that the attorney has received no notice of revocation of the
CLASSIAL TUNES	power of attorney

Name of witness (block letters)

SIGNED by

FRANK LOWY

as attorney for WESTFIELD AMERICA MANAGEMENT LIMITED as responsible entity of WESTFIELD AMERICA TRUST under power of attorney dated 9 December 2017 in the presence of:

Signature of witness

SIMON TUXEN

Name of witness (block letters)

By executing this document the attorney states that the attorney has received no notice of revocation of the power of attorney

SIGNED by

FRANK LOWY

as attorney for WESTFIELD AMERICA MANAGEMENT LIMITED as responsible entity of WFD TRUST under power of attorney dated 9 December 2017 in the presence of:

Signature of witness

SIMON TUXEN

Name of witness (block letters)

By executing this document the attorney states that the attorney has received no notice of revocation of the power of attorney

Annexure A Public announcement



PRESS RELEASE

UNIBAIL-RODAMCO AND WESTFIELD CORPORATION ANNOUNCE THE CREATION OF THE WORLD'S PREMIER DEVELOPER AND OPERATOR OF FLAGSHIP SHOPPING DESTINATIONS

Paris, Amsterdam, Sydney - December 12, 2017

Unibail-Rodamco SE ("Unibail-Rodamco") and Westfield Corporation ("Westfield") announce that Unibail-Rodamco has entered into an agreement to acquire Westfield to create the world's premier developer and operator of flagship shopping destinations (the "Group"). The proposed transaction (the "Transaction") has been unanimously recommended by Westfield's Board of Directors 1 and Unibail-Rodamco's Supervisory Board.

Under the terms of the agreement, Westfield securityholders will receive a combination of cash and shares in Unibail-Rodamco², valuing each Westfield security at a price of US\$7.55 (or A\$10.01)³ and representing a premium of 17.8% to Westfield's closing security price on December 11, 2017. The Transaction implies an enterprise value for Westfield of US\$24.7 billion⁴.

The business combination is a unique value proposition for both Unibail-Rodamco shareholders and Westfield securityholders, who will benefit from:

- The creation of a global property leader with €61.1 billion (US\$72.2 billion) of Gross Market Value ("GMV")⁵, strategically positioned in 27 of the world's most attractive retail markets and cities;
- A unique platform of 104 assets⁶ attracting 1.2 billion visits annually, creating a must-have partner for all global retailers and brands across Europe and select markets in the United States. 56 of the 104 assets, representing 84% of the combined GMV, are flagship shopping destinations;
- The progressive roll-out of the world famous Westfield brand in the Group's flagship shopping destinations;
- Strong organic long-term growth prospects through the world's largest development pipeline of €12.3 billion focused on flagship assets in key markets;
- A value creating transaction, which Unibail-Rodamco expects to be accretive to its recurring earnings per share in the first full year⁸, with expected run-rate synergies of €100 million per annum⁹;
- A robust balance sheet, with an expected credit rating in the "A" category post Transaction and an expected pro forma Loan-To-Value ("LTV")¹⁰ of 39% as at June 30, 2017¹¹, well within Unibail-Rodamco's historical objective of between 35% and 45%;

¹ In the absence of a Superior Proposal and subject to an Independent Expert concluding that the Transaction is in the best interests of Westfield securityholders.

Comprising Unibail-Rodamco shares and shares in a new Dutch REIT (Real Estate Investment Trust) to be formed which, at closing, will be stapled together and will trade as a single security. Refer to structure details below.

Based on Unibail-Rodamco's closing share price of €224.10 on December 11, 2017 and exchange rates of €1 = US\$1.1801 and

US\$1 = AU\$1.3266 as at December 11, 2017 (Source: Bloomberg).

⁴ Enterprise value stated on a proportionate basis. Based on Unibail-Rodamco's closing share price on December 11, 2017 and

assumes 2,100,692,636 Westfield securities on issue (fully diluted).

⁵ Westfield GMV figures based on net market value as of June 30, 2017, and Unibail-Rodamco's estimates for transfer taxes and other adjustments from net to gross market value. Unless otherwise stated, all GMV figures in this statement are quoted on a proportionate basis: €43.0 billion for Unibail-Rodamco and €18.1 billion for Westfield. Including 100% of the assets fully consolidated and share of assets accounted for under equity method.

As at June 30, 2017, adjusted for disposals and new centre openings in H2-2017.

⁷ Proportionate total investment cost as of June 30, 2017, adjusted for new centre openings in H2-2017. Westfield figures based on Unibail-Rodamco's definition of total investment cost.

Source: Unibail-Rodamco.

⁹ Source: Unibail-Rodamco: accretion reflects impact of expected run-rate synergies.



- An attractive dividend distribution policy, representing 85% to 95% of the Group's pro forma recurring net earnings;
- An efficient structure with the Group operating as a REIT in France, the Netherlands, the United Kingdom and the United States;
- An enlarged capital markets profile with a pro forma market capitalization of €31.1 billion¹², increased stock liquidity and inclusion in major equity indices in Europe and Australia¹³;
- A best-in-class management team, leveraging Unibail-Rodamco and Westfield's superior track records and the strengths of both groups, with the support of their 3,700 employees;
- Post Transaction, Christophe Cuvillier will be the Group Chief Executive Officer and Colin Dyer will be the Group Chairman of the Supervisory Board.

Commenting on the Transaction, Christophe Cuvillier, Chairman of the Management Board and Chief Executive Officer of Unibail-Rodamco, said: "We are delighted to announce this transaction today. All of us at Unibail-Rodamco have immense respect for what the Lowy family and the Westfield team have accomplished with the Westfield brand and the company's iconic collection of world class shopping destinations. The acquisition of Westfield is a natural extension of Unibail-Rodamco's strategy of concentration, differentiation and innovation. It adds a number of new attractive retail markets in London and the wealthiest catchment areas in the United States. It provides a unique platform of superior quality shopping destinations supported by experienced professionals of both Unibail-Rodamco and Westfield. We believe that this transaction represents a compelling opportunity for both companies to realize benefits not available to each company on a standalone basis, and creates a strong and attractive platform for future growth. We look forward to welcoming Westfield's securityholders as shareholders in the new Group and continuing to create significant value for our existing and new shareholders."

Commenting on the Transaction, Sir Frank Lowy AC, Chairman of the Westfield Board of Directors, said: "The transaction announced today is the culmination of the strategic journey Westfield has been on since its 2014 restructure. We see this transaction as highly compelling for Westfield's securityholders and Unibail-Rodamco's shareholders alike. Unibail-Rodamco's track record makes it the natural home for the legacy of Westfield's brand and business. We look forward to seeing Westfield continue to grow as part of the world's premier owner of flagship shopping destinations."

Key transaction terms

Under the Transaction terms, Unibail-Rodamco will acquire Westfield for shares and cash via Australian company and trust Schemes of Arrangement. The consideration offered to Westfield's securityholders comprises 0.01844 Unibail-Rodamco stapled securities (which will trade as one single security)¹⁴ ("Scrip Consideration") plus US\$2.67 in cash for each Westfield security ("Cash Consideration") (together, the "Offer").

¹⁰ Consolidated LTV as per Unibail-Rodamco's definition as at June 30, 2017 on a pro forma basis. Assuming issuance of c.€2 billion of deeply subordinated hybrid securities as part of the take out of the acquisition bridge facility. Pro forma proportionate LTV would be 41%.

¹¹ Source: Unibail-Rodamco.

¹² Based on Unibail-Rodamco's share price of €224.10 as at December 11, 2017 and approximately 138.6 million Unibail-Rodamco securities on issue post Transaction.

Through CHESS Depositary Interests (CDIs) to be listed on the Australian Securities Exchange.

¹⁴ Stapled securities to comprise Unibail-Rodamco shares and shares in a new Dutch REIT to be formed by Unibail-Rodamco, which, at closing, will be stapled together and will trade as a single security. Refer to structure details below.



The Offer implies a total value of US\$7.55 (or A\$10.01)¹⁵ per Westfield security based on Unibail-Rodamco's closing price of €224.10 on December 11, 2017, representing:

- A 17.8% premium based on Westfield's closing security price of US\$6.41 (A\$8.50) on December 11, 2017;
- A 22.7% premium based on Westfield's volume weighted average closing security price over the three months ending December 11, 2017 of US\$6.15¹⁶.

Approximately 38.7 million Unibail-Rodamco stapled securities will be issued to Westfield securityholders to fund the Scrip Consideration¹⁷ and a total of US\$5.6 billion will be paid as the Cash Consideration, resulting in a 65% stock and 35% cash consideration mix. Post Transaction, existing Unibail-Rodamco shareholders will hold c.72% of the Group's stapled securities and Westfield securityholders will hold c.28%.

The Group intends to establish Chess Depositary Interest ("CDI") listed on the Australian Securities Exchange ("ASX"), which will be fully exchangeable with the new Group's stapled securities ¹⁸ listed in Amsterdam and Paris. Westfield securityholders will be able to elect whether to receive the Scrip Consideration in Unibail-Rodamco stapled securities or the Group's CDIs. The Group's CDIs listed on the ASX are expected to be eligible for S&P index inclusion in Australia.

The dividends payable by each of Unibail-Rodamco and Westfield with respect to profits generated during fiscal year 2017 will be paid separately to each set of securityholders on or before closing in line with their respective dividend payout policies. All dividends with respect to fiscal year 2018 will be paid to shareholders of the new Group post Transaction. Unibail-Rodamco expects to maintain its distribution policy and calendar.

The Supervisory Board and the Management Board of Unibail-Rodamco unanimously support the Transaction. The Board of Directors of Westfield has also unanimously recommended the Transaction in the absence of a Superior Proposal and subject to an Independent Expert concluding that the Transaction is in the best interests of the Westfield securityholders.

The Lowy family has entered into a Voting Agreement under which it has agreed not to sell its interest in Westfield during the period of the Transaction, and to vote in favour of the Transaction in the absence of the Westfield Board recommending a Superior Proposal and subject to an Independent Expert concluding that the Transaction is in the best interests of Westfield securityholders. In addition, Unibail-Rodamco holds a 4.9% economic interest in Westfield securities.

Following the Transaction, the Lowy family is committed to the success of the Group and intends to maintain a substantial investment in the Group.

The Transaction is conditional upon the satisfaction of customary conditions, including Australian court approval and the approval of Unibail-Rodamco shareholders and Westfield securityholders, and is expected to close in H1-2018.

Prior to implementation of the Transaction, it is proposed that a 90% interest in OneMarket (formerly Westfield Retail Solutions), Westfield's retail technology platform, will be spun-off from Westfield into a newly formed ASX listed entity. The Group will retain the remaining 10% interest in OneMarket. OneMarket will have approximately US\$200 million in cash at December 31, 2017. Steven Lowy will chair the OneMarket Board and Don Kingsborough will be its CEO.

_

¹⁵ Based on Unibail-Rodamco's closing share price of €224.10 on December 11, 2017 and exchange rates of €1 = US\$1.1801 and US\$1 = AU\$1.3266 as at December 11, 2017 (Source: Bloomberg).

¹⁶ Refer to footnote 15.

¹⁷ Comprising Unibail-Rodamco shares and shares in a new Dutch REIT to be formed by Unibail-Rodamco, which, at closing, will be stapled together and will trade as a single security. Refer to structure details below.

¹⁸ Refer to footnote 17.



OneMarket has been developed within Westfield and it has always been envisaged that it would be separated at the appropriate time.

Westfield's Board believes that ultimately OneMarket will be better positioned as an independent company given its technology focus, financial profile and ability to develop its products as a standalone industry platform.

The ASX listed entity that will hold the 90% interest in OneMarket will be demerged to Westfield's securityholders on a pro rata basis shortly prior to closing of the Transaction. The Transaction is not conditional on the demerger occurring.

The Group and OneMarket intend to enter into a commercial arrangement in relation to OneMarket's services.

Compelling strategic rationale

With 17 iconic flagship assets across 9 gateway cities representing 85% of its GMV, Westfield's high quality portfolio is a perfect strategic fit for Unibail-Rodamco. The combination with Westfield's platform will add highly attractive markets: London and the wealthiest catchment areas in the United States. The Transaction will create the world's premier developer and operator of flagship shopping destinations:

- A unique portfolio of 104 prime assets across 13 countries attracting over 1.2 billion visits annually;
- The best shopping destinations in the world's wealthiest cities such as London, Los Angeles, Munich, New York, Paris, San Francisco, San Jose, Stockholm, Vienna, Madrid and Warsaw;
- An unparalleled collection of 56 high quality flagship shopping destinations, representing 84% of the GMV, with an average footfall of 15.7 million per annum, such as Westfield London, Westfield Stratford City, Les Quatre Temps, Westfield Century City, Le Forum des Halles, Westfield World Trade Center, La Maquinista, Shopping City Süd, Mall of Scandinavia, Centrum Chodov and Arkadia:
- An average GMV per shopping centre of €599 million (€942 million for the flagship portfolio)¹⁹, significantly ahead of all other large shopping centre REITs;
- A combination of two development pipelines for a total amount of €12.3 billion²⁰, with iconic developments in London, Milan, Hamburg, Brussels, Paris, San Jose, Lyon and other major cities. In addition to benefiting from the strong embedded organic growth potential, Unibail-Rodamco expects to capitalise on Westfield's strong development and investment track record and know-how;
- The creation of the world's leading platform for global retailers and brand events, positioning the Group as a must-have partner for all global retailers across the most attractive markets in the United States and Europe;
- Best-in-class management team, capitalising on each of Unibail-Rodamco's and Westfield's superior track records and strengths, with the support of their 3,700 employees.

Post Transaction, Unibail-Rodamco will own and operate the world's leading high quality commercial property portfolio with a total GMV of over €61.1 billion (US\$72.2 billion) and a pro forma proportionate

-

¹⁹ Based on 100% GMV for each shopping centre.

²⁰ Proportionate total investment cost as of June 30, 2017, adjusted for new centre openings in H2 2017. Westfield figures based on Unibail-Rodamco's definition of total investment cost.



net rental income of €2.3 billion (US\$2.6 billion) for the twelve months to June 30, 2017²¹. The shopping centre portfolio will represent 87% of the pro forma Group's GMV alongside Unibail-Rodamco's existing office (7% of pro forma GMV) and convention & exhibition (6% of pro forma GMV)²² portfolios, both located in Paris and providing significant earnings growth potential and attractive development projects.

The combination of Unibail-Rodamco and Westfield is a natural extension of Unibail-Rodamco's strategy of concentration, differentiation and innovation and is consistent with its objective to focus on high quality large shopping destinations in wealthy capital cities, the most prestigious office buildings and major convention and exhibition venues, vertically integrating the entire real estate value creation chain. With this Transaction, Unibail-Rodamco will acquire a leading portfolio and an extensive development pipeline managed by an outstanding team of experienced professionals, in addition to an iconic brand. The Westfield brand, the strongest in the industry, will gradually be deployed across Unibail-Rodamco's flagship assets.

The Group will accelerate its digital innovation strategy to strengthen connections with visitors and retailers. The use of innovative apps and services (Smart map, Find my car, Connect, Click & Services) and social media (10 million Facebook and Instagram fans) will enhance the visitor experience and foster communities, with whom the Group's shopping centres can engage.

Clear value proposition

Unibail-Rodamco has identified a total of €100 million of expected run-rate synergies, reflecting a unique opportunity to create value for both sets of securityholders and deliver stronger returns than either of the two groups could achieve on a standalone basis. Approximately €40 million of synergies are anticipated to be driven by the incremental rental income expected from implementation of leasing efforts, leveraging the Westfield brand and specialty leasing expertise across Unibail-Rodamco's platform as well as the deployment of Unibail-Rodamco's best practices and capabilities in asset management and capturing reversionary potential. An additional €60 million is expected to come from corporate overheads.

Unibail-Rodamco expects the Transaction to be accretive for Unibail-Rodamco's shareholders from the first full year.

Financing

Deutsche Bank and Goldman Sachs have provided a €6.1 billion committed acquisition financing facility to cover the cash portion of the Offer, refinancing requirements at Westfield and Unibail-Rodamco and transaction costs. The bridge facility is expected to be repaid with a combination of senior debt and deeply subordinated hybrid securities (€2 billion). Pursuant to its ongoing portfolio review, Unibail-Rodamco has earmarked an amount of approximately €3 billion of assets to be disposed over the next several years.

Unibail-Rodamco expects the Group to retain an "A" category credit rating.

Governance and organisation

Following the Transaction, Unibail-Rodamco will maintain its two-tier Board structure which consists of a Supervisory Board and a Management Board.

The Supervisory Board of Unibail-Rodamco will be chaired by Colin Dyer. Two Westfield board members, including Peter Lowy, will be appointed to the Supervisory Board.

A newly created Advisory Board, to be chaired by Sir Frank Lowy AC, will provide the Group with independent advice from outside experts on its strategy.

5

²¹ Proportionate net rental income based on Unibail-Rodamco's definition of net rental income, i.e. net of lease incentive amortisation.
²² Including services.



The Management Board of Unibail-Rodamco will consist of Christophe Cuvillier, Group Chief Executive Officer, and Jaap Tonckens, Group Chief Financial Officer.

The Senior Management Committee will include top executives of both Westfield and Unibail-Rodamco, reflecting the combination of their strengths. It will assist the Management Board in the execution of the Group's strategy.

A dedicated committee, comprising members of the Unibail-Rodamco and Westfield management teams, will oversee integration planning and execution to ensure efficient realisation of the expected benefits of the Transaction. Subject to the Transaction being approved, Sir Frank Lowy AC will retire as Chairman of Westfield. Peter Lowy and Steven Lowy will retire as Co-CEOs of Westfield, Michael Gutman will retire as President and COO of Westfield and Elliott Rusanow will retire as CFO of Westfield.

The Group will have its headquarters in Paris and Schiphol (Netherlands), with two regional headquarters in Los Angeles and London.

Unibail-Rodamco and a newly created Dutch REIT to become stapled entities

Upon the closing of the Transaction, Unibail-Rodamco shareholders and Westfield securityholders will hold Unibail-Rodamco stapled securities, each comprising one Unibail-Rodamco share and one share in a newly formed Dutch company ("Newco"), which will hold Westfield's US operations²³ and will become a Dutch REIT (*fiscale beleggingsinstelling*).

Newco will have two classes of shares. The ordinary class A shares will be held by current Unibail-Rodamco shareholders (received as a distribution on their Unibail-Rodamco shares) and Westfield securityholders (received as part of the Transaction consideration, along with Unibail-Rodamco shares and cash). Each Newco ordinary class A share will be simultaneously stapled to one Unibail-Rodamco share and will only be tradable as one single security. The class B shares will be held by Unibail-Rodamco and will represent a 40% stake in Newco.

Newco will also oversee future investments in the Netherlands. Christophe Cuvillier and Jaap Tonckens will be members of the Supervisory Board of Newco, chaired by Christophe Cuvillier.

Following the completion of the Transaction, Unibail-Rodamco stapled securities will be listed on Euronext Amsterdam (market of reference) and Paris. Unibail-Rodamco will also establish a secondary listing on the Australian Securities Exchange to allow former Westfield securityholders to trade Unibail-Rodamco stapled securities locally in the form of CDIs. Each Unibail-Rodamco CDI will be fully exchangeable with one Unibail-Rodamco stapled security.

Post Transaction, it is expected that Unibail-Rodamco will benefit from continued inclusion in CAC40, AEX, Eurostoxx50, SBF120 and EPRA indices, and, in respect of its ASX listed CDIs, S&P index inclusion in Australia.

Documentation relating to the Transaction²⁴ will be filed with the relevant stock market authorities and will be subject to their approval.

Holders of Unibail-Rodamco ORNANEs maturing in 2021 and 2022 will be consulted prior to closing of the Transaction in connection with the stapling of the Unibail-Rodamco and Newco shares.

Transaction structure preserves existing REIT status

The proposed structure for the Transaction retains the benefit of the REIT status applicable to Unibail-Rodamco and Westfield in France (SIIC), the Netherlands (FBI), the UK (REIT) and the US (REIT) so as to preserve the current tax situation of both Westfield and Unibail-Rodamco securityholders and reflect the geography of the Group's assets.

-

²³ As well as some select Dutch assets.

²⁴ Including a prospectus (document E) relating to the issuance and listing of Unibail-Rodamco shares as Scrip Consideration and a prospectus relating to the listing of Unibail-Rodamco's stapled securities.



Westfield expects that a portion of the Scrip Consideration that Australian securityholders of Westfield receive in the Transaction will be eligible for capital gains tax roll-over relief. Westfield has discussed this aspect of the proposal with the Australian Taxation Office (ATO) and will work with the ATO to obtain a class ruling for the benefit of Australian securityholders that confirms the treatment of the cash and share component of the offer. This treatment will be described in the explanatory memoranda expected to be sent to securityholders in Q2-2018.

Conditions to the Transaction

The Transaction is subject to customary conditions which are described in the Implementation Agreement, including the receipt of certain regulatory approvals, the approval of Unibail-Rodamco's shareholders and Westfield's securityholders, Australian court approval for the proposed Schemes, and admission of the Unibail-Rodamco stapled security to trading on Euronext and the ASX. The Transaction is also subject to the conclusion of Unibail-Rodamco's works council consultation process.

Further process and indicative timetable

Following the execution of the Implementation Agreement between the parties and today's announcement, Unibail-Rodamco intends to commence its works council consultation process immediately. Unibail-Rodamco and Westfield expect to issue explanatory memoranda to their respective shareholders and securityholders and convene shareholder and securityholder meetings in Q2-2018. The Transaction is expected to complete in Q2-2018.

Indicative timetable is as follows:

- Signing of the Implementation Agreement: December 12, 2017
- Consultation of Unibail-Rodamco's works council: Q1-2018
- Receipt of regulatory approvals and satisfaction of other customary closing conditions: Q2-2018
- Notice of meeting and information dispatched to Unibail-Rodamco shareholders and Westfield securityholders: Q2-2018
- Unibail-Rodamco shareholder and Westfield securityholder meetings and Australian court approval: Q2-2018
- Closing: Q2-2018.

The Implementation Agreement is available on Unibail-Rodamco's website.

Financial, legal and tax advisors to Unibail-Rodamco

Deutsche Bank and Goldman Sachs are acting as financial advisors to Unibail-Rodamco. Darrois Villey Maillot Brochier, Allens, NautaDutilh, Shearman & Sterling LLP, Clifford Chance Europe LLP and Capstan Avocats are acting as legal advisors. Lacourte Raquin Tatar, Loyens & Loeff and Allen & Overy are acting as tax advisors.

Financial, legal and tax advisors to Westfield

Rothschild & Co is acting as lead financial advisor to Westfield. Jefferies and UBS are acting as joint financial advisors. King & Wood Mallesons, Skadden Arps, Slate, Meagher & Flom LLP and Debevoise & Plimpton LLP are acting as legal advisors. Greenwoods and Herbert Smith Freehills are acting as Australian tax advisors.



Unibail-Rodamco investor and analyst events

Investor and analyst conference

- 9.00am CET on December 12, 2017
- 7, place du Chancelier Adenauer 75116 Paris France.

The conference will be accessible via live video webcast on the Unibail-Rodamco's website.

Dial in details:

+61 291011926 France (EN): +33 172727403 France (FR): +33 170710159 Netherlands: +31 207095119 United Kingdom: +44 2071943759 United States: +1 8442860643

Code: 67634765#

For further information, please contact:

Unibail-Rodamco

Investor Relations
Paul Douay
D: +33 1 76 77 58 02
paul.douay@unibail-rodamco.com

Media Relations Caroline Bruel D: +33 1 53 43 57 94

caroline.bruel@unibail-rodamco.com

Westfield

Investor Relations
Josh Itzkowic
D: +61 293587011
Jltzkowic@westfield.com

Media Relations Anita Sulentic M: + 61 404 045 049 D: +61 2 9358 7997

asulentic@westfield.com

Brunswick (international media):

France: UK: US:

Jérôme Biscay Nina Coad Arif Shah

Mari 22 C 00 04 70 00 Mari 44 7074 003 550 Mari 44 640

M: +33 6 09 94 79 88 M: +44 7974 982 558 M: +1 646 404 4875 D: +33 1 53 96 83 88 D: +44 207 396 3558 D: +1 646 289 8791 Aurélia de Lapeyrouse Emily Trapnell

M: +33 6 21 06 40 33 D: +44 207 936 7423 D: +33 1 53 96 83 72 M:+44 7823 527137

Image 7 (French media):

Myriam Lévy Anne-France Malrieu M: +33 6 80 34 99 66 M: +33 1 53 70 74 51 D: +33 1 53 70 74 66



Domestique (Australian media)

Jim Kelly M: +61 412 549 083 D: +61 2 9119 3352 Lauren Thompson M: +61 438 954 729 D: +61 2 9119 3078



About Unibail-Rodamco

Created in 1968, Unibail-Rodamco SE is Europe's largest listed commercial property company, with a presence in 11 EU countries, and a portfolio of assets valued²⁵ at €42.5 billion as of June 30, 2017. As an integrated operator, investor and developer, the Group aims to cover the whole of the real estate value creation chain. With the support of its 2,000 professionals, Unibail-Rodamco applies those skills to highly specialised market segments such as large shopping centres in major European cities and large offices and convention & exhibition centres in the Paris region.

Unibail-Rodamco owns and operates 69 shopping centres, of which 56 attract more than 6 million visits per annum. These shopping centres are located in the largest and wealthiest cities in Europe, such as Paris, Madrid, Stockholm, Amsterdam, Munich, Vienna, Warsaw and Prague.

In addition to its standing assets, Unibail-Rodamco has €8.1 billion of development projects²⁶ as of June 30, 2017. This pipeline includes iconic world-class retail projects such as Mall of Europe in Brussels and Überseequartier in Hamburg.

The Group distinguishes itself through its focus on the highest architectural, city planning and environmental standards. Its long term approach and sustainable vision focuses on the development or redevelopment of outstanding places to shop, work and relax. Its commitment to environmental, economic and social sustainability has been recognised by inclusion in the FTSE4Good and STOXX Global ESG Leaders indexes.

The Group is a member of the CAC 40, AEX 25 and EuroSTOXX 50 indices. It benefits from an A rating from Standard & Poor's and Fitch Ratings.

For more information, please visit our website: www.unibail-rodamco.com

About Westfield

Westfield Corporation (ASX Code: WFD) is an internally managed, vertically integrated, shopping centre group undertaking ownership, development, design, construction, funds/asset management, property management, leasing and marketing activities and employing approximately 2,000 staff worldwide. Westfield Corporation has interests in 35 shopping centres in the United States, and the United Kingdom, encompassing approximately 6,400 retail outlets and total assets under management of US\$32 billion²⁷.

²⁷ Source: Westfield.

²⁵ Consolidated gross market value.

²⁶ Consolidated figures.



IMPORTANT NOTICE

By reading the following release, you further agree to be bound by the following limitations and qualifications:

This press release is for informational purposes only and is not intended to and does not constitute an offer or invitation to sell or solicitation of an offer to subscribe for or buy, or an invitation to purchase or subscribe for, any securities, any part of the business or assets described herein, or any other interests or the solicitation of any vote or approval in any jurisdiction in connection with the proposed Transaction or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law. This press release should not be construed in any manner as a recommendation to any reader of this press release.

This press release is not a prospectus, product disclosure statement or other offering document for the purposes of Directive 2003/71/EC (and any amendments thereto) and under French, Dutch or Australian law and will not be lodged with the Australian Securities and Investments Commission. The information in this press release does not purport contain all the information that a prospective investor or security holder in Unibail-Rodamco or Westfield may require in evaluating the Transaction or a possible investment in Unibail-Rodamco or Westfield or that would be required to be included in a prospectus or product disclosure statement prepared in accordance with the requirements of Directive 2003/71/EC (and any amendments thereto), French, Dutch and Australian law.

This press release and the information contained herein are not an offer of securities for sale in the United States or to U.S. persons. Any securities referred to herein have not been and will not be registered under the Securities Act, or any U.S. state securities laws, and may not be offered or sold in the United States or to, or for the account or benefit of, U.S. persons except pursuant to an exemption from, or in a transaction not subject to, the registration requirements of the Securities Act.

By viewing this press release, you agree to and acknowledge the above. Any failure to comply with these restrictions may constitute a violation of applicable securities laws.

This press release includes only summary information and does not purport to be comprehensive. The information contained in this press release has not been independently verified. No representation, warranty or undertaking, express or implied, is made as to, and no reliance should be placed upon, the quality, fairness, accuracy, completeness or correctness of the information or opinions contained in this press release or as to the reasonableness of any assumptions contained herein or any other information made available (whether in writing or orally) to the recipient and Unibail-Rodamco and Westfield, their respective affiliates, directors, officers, advisors, employees, representatives or advisors, expressly disclaim any and all liability based, in whole or in part, on such information, errors therein or omissions therefrom.

FORWARD LOOKING STATEMENTS

This press release contains certain statements that are forward-looking relating to Unibail-Rodamco and Westfield and their combined businesses, management's business strategies, expansion and growth of operations, future events, trends or objectives and expectations, which do not refer to historical facts but refer to expectations based on management's current views and assumptions and involve known and unknown risks, contingencies and uncertainties that could cause actual results, performance or events to differ materially from those explicitly or implicitly included in such statements.

Forward-looking statements generally will be accompanied by words such as "anticipate," "believe," "plan," "could," "estimate," "expect," "forecast," "guidance," "intend," "may," "possible," "potential," "predict," "project" or other similar words, phrases or expressions. Many of these risks and uncertainties relate to factors that are beyond Unibail-Rodamco's or Westfield's control. Therefore, investors and securityholders should not place undue reliance on such statements.



These statements or disclosures may discuss goals, intentions and expectations as to future trends, plans, events, results of operations or financial conditions, or state other information relating to Unibail-Rodamco and Westfield and their combined businesses, based on current beliefs of management as well as assumptions made by, and information currently available to, management.

Unibail-Rodamco and Westfield, their respective affiliates, directors, advisors, employees and representatives, expressly disclaim any liability whatsoever for such forward-looking statements.

Factors that could cause actual results to differ materially from those in the forward-looking statements include, but are not limited to: uncertainties related in particular to the economic, financial, competitive, tax or regulatory environment; the ability to obtain the approval of the Transaction by securityholders; failure to satisfy other closing conditions with respect to the Transaction on the proposed terms and timeframe; the possibility that the Transaction does not close when expected or at all; the risks that the new businesses will not be integrated successfully or that the Group will not realize estimated cost savings and synergies; Unibail-Rodamco's or Westfield's ability to successfully implement and complete its plans and strategies and to meet its targets; the benefits from Unibail-Rodamco's or Westfield's (and their combined businesses) plans and strategies being less than anticipated; and the risks described in Unibail-Rodamco's registration document (document de référence) filed with the Autorité des marchés financiers (French securities regulator), which are also available in English on Unibail-Rodamco's website (http://www.unibail-rodamco.com). The foregoing list of factors is not exhaustive.

Forward-looking statements speak only as of the date they are made. Unibail-Rodamco and Westfield do not assume any obligation to update any public information or forward-looking statement in this communication to reflect new information, future events or circumstances or for any other reason after the date of this communication, except as may be required by applicable laws, and any opinion expressed in this press release is subject to change without notice. Unibail-Rodamco and Westfield shall not have any obligation to correct any inaccuracies therein or omissions therefrom which may become apparent. This press release includes some information on specific transactions that shall be considered as projects only and remain subject to certain approvals and other conditions, including the prior information and consultation of Unibail-Rodamco's employee representative bodies.

All statements and other information about or related to earnings, or net asset, accretion or synergies consequent upon the Transaction are the responsibility of Unibail-Rodamco. Westfield takes no responsibility and disclaims all liability for such statements or other information.

The distribution of this press release may be subject to legal or regulatory restrictions in certain jurisdictions. Any person who comes into possession of this press release must inform itself, himself or herself of and comply with any such restrictions.

Investors are strongly advised to read, if and when they become available, information materials relating to the Transaction as they will contain important information.

Implementation Agreement

Annexure B Scheme of Arrangement

91



Scheme of Arrangement

Westfield Corporation Limited (ABN 12 166 995 197) ("WCL")

Scheme Participants

King & Wood Mallesons

Level 61
Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia
T +61 2 9296 2000
F +61 2 9296 3999
DX 113 Sydney
www.kwm.com

Scheme of Arrangement

Contents

Contents			
Details	Details		
Genera	al terms	5	
1	Definitions and interpretation	5	
1.1 1.2	Definitions General interpretation	5 10	
2	Preliminary	11	
2.1 2.2 2.3 2.4 2.5	Westfield Unibail Scheme as part of the Transaction Scheme Implementation Agreement Deed Poll	11 11 11 12 12	
2.5 2.6 2.7	Undertaking by Third Party Nominee Interpretation of references to Scheme Shares	12 12 12	
3	Conditions precedent	12	
3.1 3.2	Conditions precedent to Scheme Conditions precedent and operation of clause 5	12 13	
4	Scheme	13	
4.1 4.2	Effective Date End Date	13 13	
5	Implementation of Scheme	13	
5.1 5.2	Lodgement of Court orders with ASIC Transfer and registration of Scheme Shares and the TH	13	
5.3 5.4 5.5 5.6 5.7 5.8 5.9	Newco Shares Entitlement to receive Scheme Consideration Title and rights in Scheme Shares Scheme Participants' agreements Warranty by Scheme Participants Transfer free of encumbrances Appointment of Unibail as sole proxy Transaction Steps	13 14 14 14 14 14 15	
6	Scheme Consideration	15	
6.1 6.2 6.3 6.4 6.5 6.6	Scheme Consideration Fractions and splitting Currency election Scrip Consideration Minimum Holders Ineligible Foreign Holders Sale Facility	15 16 16 18 19 19	
6.8 6.9	Provision of Cash Consideration Provision of Scrip Consideration	21 22	

6.10	Joint holders	23
6.11	Unclaimed monies	23
6.12	Orders of a court or Regulatory Authority	23
6.13	Status of New Unibail Shares and New Unibail CDIs	24
7	Dealings in Scheme Shares	25
7.1	Determination of Scheme Shares	25
7.2	Register	25
7.3	No disposals after Effective Date	25
7.4	Maintenance of Share Register	25
7.5	Effect of certificates and holding statements	25
7.6	Details of Scheme Participants	26
7.7	Quotation of WCL Shares	26
7.8	Termination of quotation of WCL Shares	26
8	Power of attorney	26
9	Notices	26
9.1	No deemed receipt	26
9.2	Accidental omission	26
10	General	27
10.1	Variations, alterations and conditions	27
10.2	Further action by WCL	27
10.3	No liability when acting in good faith	27
10.4	Enforcement of Deed Poll	27
10.5	Stamp duty	27
11	Governing law	27
11.1	Coverning low and jurisdiction	27
	Governing law and jurisdiction	21

Details

Parties	WCL and Scheme Participants	
WCL	Name	Westfield Corporation Limited (ABN 12 166 995 197)
	Address	Level 29, 85 Castlereagh Street, Sydney NSW 2000, Australia
	Email	STuxen@westfield.com
	Attention	Simon Tuxen
Scheme Participants	Each person registered as a holder of fully paid ordinary shares in WCL as at the Record Date.	
Governing law	New South Wales	

General terms

1 **Definitions and interpretation**

1.1 **Definitions**

Unless the contrary intention appears, these meanings apply:

ASIC means the Australian Securities and Investments Commission.

ASX means ASX Limited (ACN 008 624 691) or the market operated by it, as the context requires.

Business Day means a business day as defined in the Listing Rules of the ASX and on which trading will take place on Euronext Paris and Euronext Amsterdam and the New York Stock Exchange.

Cash Consideration has the meaning given in clause 6.1(a).

CDN means CHESS Depository Nominees Pty Limited (ACN 071 346 506).

CHESS means the clearing house electronic sub register system of share transfers operated by ASX Settlement and Transfer Corporation Pty Ltd.

Corporations Act means the Corporations Act 2001 (Cth).

Court means the Federal Court of Australia, Sydney Registry, or such other court as is agreed in writing by Unibail and the Manager.

Currency Election means an election made or deemed to have been made under clause 6.3(a), 6.3(b) or 6.3(c).

Deed Poll means the deed poll dated [insert date] executed by Unibail under which Unibail covenants in favour of each Scheme Participant to perform its obligations under this Scheme.

Destapling means the destapling of WCL, WAT and WFDT as described in the Transaction Steps and **Destaple**, and **Destapled** each have a corresponding meaning.

Details means the section of this document headed "Details".

Effective means the coming into effect, pursuant to section 411(10) of the Corporations Act, of the order of the Court made under section 411(4)(b) of the Corporations Act in relation to this Scheme, but in any event at no time before an office copy of the order of the Court is lodged with ASIC.

Effective Date means the date on which this Scheme becomes Effective.

Election Form means an election form that Westfield will provide to any Westfield Securityholder upon request under which the Westfield Securityholder may make a Currency Election.

Election Time means 5.00pm on the first Business Day after the Effective Date or any other date agreed between Unibail and Westfield.

Encumbrance means any mortgage, charge, lien, pledge, trust, power or title retention, flawed deposit arrangement, "security interest" as defined in sections 12(1) or (2) of the Personal Property Securities Act 2009 (Cth) (Australia), or interest of a third party of any kind, whether legal or otherwise, or any agreement to create any of them or allowing them to exist.

End Date means 30 September 2018 or such other date as is agreed in writing by Unibail and Westfield.

Euronext Amsterdam means the regulated market operated by Euronext Amsterdam NV.

Euronext Paris means the regulated market operated by Euronext Paris S.A.

Equity Rights means rights to Westfield Stapled Securities issued under employee incentive arrangements by Westfield and its subsidiaries.

Implementation Agreement means the implementation agreement 12 December 2017 between Westfield and Unibail under which, amongst other things. WCL has agreed to propose this Scheme to the WCL Shareholders, and each of Westfield and Unibail has agreed to take certain steps to give effect to this Scheme (including the Transaction Steps).

Implementation Date means the fifth Business Day following the Record Date or such other date after the Record Date as is agreed in writing by Unibail and WCL.

Ineligible Foreign Holder means a Scheme Participant whose address shown in the Share Register on the Record Date is a place outside Australia and its external territories and New Zealand, unless Unibail determines that it is lawful and not unduly onerous or impracticable to issue or provide that Scheme Participant with New Unibail CDIs or New Unibail Shares under this Scheme.

Minimum Holder means a Scheme Participant who is entitled to receive a parcel of New Unibail Shares under this Scheme and Newco BI Shares under the WAT Trust Scheme which, when stapled together and trading as New Unibail Stapled Securities, will have a value of less than €500 (calculated with reference to the price of Unibail Shares on Euronext Paris as at the close of trading on the Effective Date).

Net Sale Proceeds has the meaning given in clause 6.7(b)(ii).

New Unibail CDI means a New Unibail CHESS Depositary Interest, being a unit of beneficial ownership in a New Unibail Share registered in the name of CDN, to be issued under this Scheme.

New Unibail Share means a Unibail Share issued under this Scheme.

New Unibail Stapled Security means a Unibail Share and a Newco BI Share stapled together in accordance with the articles of association of Unibail and Newco BI and in compliance with applicable French and Dutch laws.

Newco BI means a public company with limited liability (naamloze vennootschap) incorporated or to be incorporated in the Netherlands with shares as described in Schedule 3 of the Implementation Agreement, as referred to in the WAT Constitutional Amendment.

Newco BI CDI means a Newco BI CHESS Depositary Interest, being a unit of beneficial ownership in a Newco BI Share registered in the name of CDN, to be issued under the WAT Trust Scheme.

Newco BI Share means a Class A Share in the capital of Newco BI as described in Schedule 3 of the Implementation Agreement.

Record Date means 5.00pm on the fourth Business Day following the Effective Date or such other date as WCL and Unibail may agree in writing.

Registered Address means, in relation to a Westfield Securityholder, the address shown in the Share Register.

Regulatory Authority includes:

- (a) ASX:
- ASIC: (b)
- Australian Taxation Office: (c)
- (d) a government or governmental, semi-governmental or judicial entity or authority;
- a minister, department, office, commission, delegate, instrumentality, (e) agency, board, authority or organisation of any government; and
- any regulatory organisation established under statute. (f)

Sale Agent means an entity appointed by Unibail to sell the Sale Securities pursuant to clause 6.7.

Sale Facility means the facility provided for in clause 6.7 of this Scheme.

Sale Securities has the meaning given in clause 6.7(a).

Scheme means this scheme of arrangement under Part 5.1 of the Corporations Act between WCL and Scheme Participants, subject to any alterations or conditions made or required by the Court pursuant to section 411(6) of the Corporations Act to the extent they are approved in writing by WCL and Unibail.

Scheme Booklet means the scheme booklet published by Westfield and dated [**insert**] 2018.

Scheme Consideration means the consideration to be provided to Scheme Participants under the terms of this Scheme for the acquisition of the Scheme Shares, as ascertained in accordance with clause 6.

Scheme Meeting means the meeting of WCL Shareholders, ordered by the Court to be convened pursuant to section 411(1) of the Corporations Act at which WCL Shareholders will vote on this Scheme.

Scheme Participant has the meaning given in the Details.

Scheme Shares means WCL Scheme Shares and TH Newco Scheme Shares.

Scrip Consideration has the meaning given in clause 6.1(b).

Scrip Election has the meaning given in clause 6.4(c).

Second Court Date means the day on which the Court makes an order pursuant to section 411(4)(b) of the Corporations Act approving this Scheme.

Second Judicial Advice means confirmation from the Court under section 63 of the Trustee Act 1919 (NSW) (Australia) that WAML would be justified in acting

upon the Trust Scheme Resolutions in doing all things and taking all necessary steps to put the Trust Schemes into effect.

Selling Scheme Participant means a Scheme Participant in respect of whom New Unibail Shares are issued to the Sale Agent under this Scheme.

Share Register means the register of members of WCL maintained by or on behalf of WCL in accordance with section 168(1) of the Corporations Act.

Stapling Deed means the stapling deed between WCL and WAML (as responsible entity for WAT and for WFDT) dated 30 June 2014.

TH Newco means a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated or to be incorporated in the Netherlands as referred to in the WFDT Constitutional Amendment.

TH Newco Scheme Shares means the TH Newco Shares which are issued on the Implementation Date under the WFDT Trust Scheme.

TH Newco Share means a fully-paid ordinary share in the share capital (*capital social*) of TH Newco.

TH Newco Transfer Form means one or more duly executed notarial deed(s) of transfer for the transfer of the TH Newco Scheme Shares to Unibail.

Third Party Nominee means [*], as referred to in the WFDT Trust Constitutional Amendment.

Third Party Nominee Deed Poll means the deed poll referred to in clause 2.6.

Transaction means the acquisition of Westfield by Unibail contemplated by the Implementation Agreement.

Transaction Steps means each of the steps set out in Schedule 3 of the Implementation Agreement.

Trust Scheme Resolutions means the resolutions of Westfield Securityholders to approve the Trust Schemes (including resolutions required under section 611 item 7 and section 601GC(1) of the Corporations Act) and the Destapling.

Trust Schemes means the arrangements under which US Newco acquires all of the WAT Units and TH Newco acquires all of the WFDT Units to give effect to the Transaction, facilitated by amendments to the WAT Constitution and the WFDT Constitution as set out in the Westfield Constitutional Amendments.

Unibail means Unibail-Rodamco SE (Registration Number 682 024 096).

Unibail Registry means the register of interest-holders maintained by Unibail or its agent.

Unibail Share means a fully paid ordinary share in the share capital (*capital social*) of Unibail, with a nominal value of €5.

US Newco means a corporation incorporated or to be incorporated in the United States as referred to in the WAT Trust Constitutional Amendment.

WAML means Westfield America Management Limited (ABN 66 072 780 619).

WAT means Westfield America Trust (ARSN 092 058 449).

WAT Constitution means the trust deed establishing WAT dated 28 March 1996, as amended.

WAT Constitutional Amendment means the supplemental deed poll under which WAML will amend the WAT Constitution for the purposes of implementing the WAT Trust Scheme.

WAT Trust Scheme means the arrangement under which US Newco acquires all of the WAT Units, facilitated by amendments to the WAT Constitution, as set out in the WAT Constitutional Amendments.

WAT Trust Scheme Consideration means the scheme consideration to be provided to Westfield Securityholders under the WAT Trust Scheme.

WAT Trust Scheme Unit means a WAT Unit on issue as at the Record Date.

WAT Unit means a fully paid ordinary unit in WAT.

WCL Constitution means the constitution of WCL.

WCL Scheme Share means a WCL Share on issue as at the Record Date.

WCL Share means a fully paid ordinary share in the capital of WCL.

WCL Shareholder means each person registered in the Share Register as a holder of one or more WCL Shares.

WCL Transfer Form means, for each Scheme Participant, a duly completed and executed proper instrument of transfer of the WCL Scheme Shares held by that Scheme Participant, which may be a master transfer of all WCL Scheme Shares.

Westfield means WCL, WAML as responsible entity of WAT and as responsible entity of WFDT (together or any of them individually as the context requires).

Westfield Constitutional Amendments means the WAT Constitutional Amendment and the WFDT Constitutional Amendment.

Westfield Constitutions means the WCL Constitution, WAT Constitution and WFDT Constitution.

Westfield Registry means Computershare Investor Services Pty Limited (ACN 078 279 277) or such other entity which provides securities registry services to Westfield from time to time.

Westfield Securityholder means a person who is registered in the Share Register, WAT unit register of members and WFDT unit register of members as the holder of one or more Westfield Stapled Securities.

Westfield Stapled Security means a WCL Share, a WAT Unit and a WFDT Unit stapled to each other in accordance with the provisions of the Stapling Deed and the Westfield Constitutions.

WFDT means WFD Trust (ARSN 168 765 875).

WFDT Constitution means the trust deed establishing WFDT dated 26 March 2014, as amended.

WFDT Constitutional Amendment means the supplemental deed poll under which WAML will amend the WFDT Constitution for the purposes of implementing the WFDT Trust Scheme.

WFDT Trust Scheme means the arrangement under which TH Newco acquires all of the WFDT Units, facilitated by amendments to the WFDT Constitution, as set out in the WFDT Constitutional Amendments.

WFDT Trust Scheme Consideration means the scheme consideration to be provided to Westfield Securityholders under the WFDT Trust Scheme.

WFDT Trust Scheme Unit means a WFDT Unit on issue as at the Record Date.

WFDT Unit means a fully paid ordinary unit in WFDT.

1.2 General interpretation

Headings and labels used for definitions are for convenience only and do not affect interpretation. Unless the contrary intention appears, in this document:

- (a) the singular includes the plural and vice versa;
- (b) a reference to a document includes any agreement or other legally enforceable arrangement created by it (whether the document is in the form of an agreement, deed or otherwise);
- (c) a reference to a document also includes any variation, replacement or novation of it;
- (d) the meaning of general words is not limited by specific examples introduced by "including", "for example", "such as" or similar expressions;
- (e) a reference to "person" includes an individual, a body corporate, a partnership, a joint venture, an unincorporated association and an authority or any other entity or organisation;
- a reference to a particular person includes the person's executors, administrators, successors, substitutes (including persons taking by novation) and assigns;
- (g) a reference to a time of day is a reference to Sydney time;
- (h) where an act must take place on a day or a Business Day it must occur at a time of day that is both part of a day of the same date in Sydney, Australia and part of the same day of the same date in Paris, France;
- (i) a reference to:
 - (i) US\$ or US dollars is a reference to the currency of the United States of America;
 - (ii) A\$ or Australian dollars is a reference to the currency of Australia; and
 - (iii) euro or € is a reference to the currency of the European Union;
- a reference to any legislation includes regulations under it and any consolidations, amendments, re-enactments or replacements of any of them;
- (k) a reference to a group of persons is a reference to any 2 or more of them jointly and to each of them individually;
- (I) a reference to any thing (including an amount) is a reference to the whole and each part of it;

- (m) a period of time starting from a given day or the day of an act or event, is to be calculated exclusive of that day:
- if a party must do something under this document on or by a given day (n) and it is done after 5.00pm on that day, it is taken to be done on the next day; and
- if the day on which a party must do something under this document is (o) not a Business Day, the party must do it on the next Business Day.

2 **Preliminary**

2.1 Westfield

WCL is:

- (a) a public company limited by shares;
- (b) incorporated in Australia and registered in Victoria; and
- admitted to the official list of the ASX along with WAT and WFDT. One (c) share in WCL and one unit in each of WAT and WFDT are stapled together and are officially quoted on the stock market conducted by ASX as Westfield Stapled Securities.

As at 12 December 2017, 2,078,089,686 Westfield Stapled Securities were on issue and 22,602,950 Equity Rights were on issue.

2.2 Unibail

Unibail is a European Company (Societas Europaea) incorporated in accordance with the laws of France.

Unibail Shares trade on Euronext Paris and Euronext Amsterdam.

2.3 Scheme as part of the Transaction

If this Scheme becomes Effective, the following steps are intended to occur on the Implementation Date, in the following order:

- (a) WCL Shares, WAT Units and WFDT Units will be Destapled;
- Scheme Participants will transfer the WAT Trust Scheme Units to US (b) Newco in return for the WAT Trust Scheme Consideration, pursuant to the WAT Trust Scheme;
- Scheme Participants will transfer the WFDT Trust Scheme Units to TH (c) Newco in return for the WFDT Trust Scheme Consideration, pursuant to the WFDT Trust Scheme:
- (d) Scheme Participants will transfer the WCL Scheme Shares and the Third Party Nominee will transfer the TH Newco Scheme Shares to Unibail, and the Scheme Participants will receive the Scheme Consideration, pursuant to this Scheme; and
- New Unibail Shares transferred as part of the Scheme Consideration, (e) and Newco BI Shares issued as part of the WAT Trust Scheme Consideration, will be stapled together to form the New Unibail Stapled Securities and will be tradeable on Euronext Paris and Euronext Amsterdam, and on ASX as New Unibail CDIs stapled to Newco CDIs.

2.4 **Scheme Implementation Agreement**

Westfield and Unibail have agreed by executing the Implementation Agreement to implement the terms of this Scheme.

2.5 **Deed Poll**

Unibail has executed the Deed Poll for the purpose of covenanting in favour of the Scheme Participants to perform (or procure the performance of) its obligations as contemplated by this Scheme, including to provide or procure the provision of the Scheme Consideration.

2.6 **Undertaking by Third Party Nominee**

The Third Party Nominee has executed a deed poll dated [*] for the purpose of covenanting in favour of each Scheme Participant to agree to, and do all things necessary for, the transfer to Unibail under this Scheme of all TH Newco Shares issued to the Third Party Nominee under the WFDT Trust Scheme.

2.7 Interpretation of references to Scheme Shares

Any reference in this Scheme to a Scheme Participant's Scheme Shares is, for the purposes only of determining entitlements to Scheme Consideration, taken to include any TH Newco Scheme Shares issued in respect of such Scheme Participant to the Third Party Nominee, whether the Scheme Participant is an Ineligible Foreign Holder or otherwise (and, for the avoidance of doubt, it is acknowledged that no Ineligible Foreign Holder had or has any beneficial or other ownership interest in any TH Newco Scheme Shares).

3 **Conditions precedent**

3.1 **Conditions precedent to Scheme**

This Scheme is conditional on, and will have no force or effect until, the satisfaction of each of the following conditions precedent:

- as at 8.00am on the Second Court Date, each of the Implementation (a) Agreement, Deed Poll and Westfield Constitutional Amendments not having been terminated;
- (b) as at 8.00am on the Second Court Date, all of the conditions precedent in clause 3.1 of the Implementation Agreement (other than the condition in the Implementation Agreement relating to Court approval of this Scheme and the giving of the Second Judicial Advice) having been satisfied or waived in accordance with the terms of the Implementation Agreement;
- (c) the grant of the Second Judicial Advice and the coming into effect of amendments made to the WAT Constitution and WFDT Constitution by the Westfield Constitutional Amendments in accordance with section 601GC(2) of the Corporations Act;
- the Court having approved this Scheme, with or without any alteration or (d) condition, pursuant to section 411(4)(b) of the Corporations Act, and if applicable, WCL and Unibail having accepted in writing any alteration or condition made or required by the Court under section 411(6) of the Corporations Act; and
- the coming into effect, pursuant to section 411(10) of the Corporations (e) Act, of the orders of the Court made under section 411(4)(b) of the Corporations Act (and, if applicable, section 411(6) of the Corporations Act) in relation to this Scheme.

3.2 Conditions precedent and operation of clause 5

The satisfaction of each condition of clause 3.1 of this Scheme is a condition precedent to the operation of clause 5 of this Scheme.

4 Scheme

4.1 **Effective Date**

Subject to clause 4.2, this Scheme will come into effect on and from the Effective Date.

4.2 **End Date**

This Scheme will lapse and be of no further force or effect (other than clause 6.8(f) which will survive) if:

- the Effective Date does not occur on or before the End Date: or (a)
- (b) the Implementation Agreement or Deed Poll is terminated in accordance with its terms.

5 Implementation of Scheme

5.1 **Lodgement of Court orders with ASIC**

If the conditions precedent set out in clause 3.1 of this Scheme (other than the condition precedent in clause 3.1(e) of this Scheme) are satisfied, WCL must lodge with ASIC in accordance with section 411(10) of the Corporations Act an office copy of the Court order approving this Scheme as soon as possible, and in any event by no later than 5.00pm on the first Business Day after the day on which the Court approves this Scheme or such later time as Unibail and WCL agree in writing.

5.2 Transfer and registration of Scheme Shares and the TH Newco **Shares**

On the Implementation Date at the time specified in the Transaction Steps:

- (a) subject to the provision of the Scheme Consideration in the manner contemplated by clause 6 and Unibail having provided Westfield with written confirmation thereof, the Scheme Shares, together with all rights and entitlements attaching to the Scheme Shares, must be transferred to Unibail without the need for any further act by any Scheme Participant (other than acts performed by WCL as attorney and agent for Scheme Participants under clause 8 of this Scheme) by:
 - Westfield delivering to Unibail duly completed and executed (i) WCL Transfer Forms executed on behalf of the Scheme Participants:
 - (ii) Unibail duly executing (or procuring the due execution of) the WCL Transfer Forms and delivering them to Westfield and TH Newco respectively for registration; and
 - execution of the TH Newco Transfer Form by the Third Party (iii) Nominee, Unibail and TH Newco; and
- (b) as soon as practicable after receipt of the duly executed WCL Transfer Forms.

Scheme of Arrangement 13 © King & Wood Mallesons

Westfield must enter, or procure the entry of, the name of Unibail in the Share Register in respect of all WCL Scheme Shares transferred to Unibail in accordance with the terms of this Scheme; and

5.3 **Entitlement to receive Scheme Consideration**

On the Implementation Date, in consideration for the transfer to Unibail of the Scheme Shares, each Scheme Participant will be entitled to receive the Scheme Consideration in respect of each of their Scheme Shares in accordance with clause 6 of this Scheme.

Title and rights in Scheme Shares 5.4

Subject to the provision of the Scheme Consideration for the Scheme Shares as contemplated by clause 6 of this Scheme, on and from the Implementation Date Unibail will be beneficially entitled to the Scheme Shares transferred to it under this Scheme.

5.5 Scheme Participants' agreements

Under this Scheme, each Scheme Participant irrevocably:

- agrees to the transfer of their WCL Scheme Shares, together with all (a) rights and entitlements attaching to those Scheme Shares, in accordance with the terms of this Scheme;
- (b) agrees to the transfer of their TH Newco Scheme Shares by the Third Party Nominee, in accordance with the terms of this Scheme, together with all rights and entitlements attaching to the TH Newco Scheme Shares:
- (c) agrees to the variation, cancellation or modification of the rights attached to their WCL Shares and TH Newco Shares constituted by or resulting from this Scheme;
- acknowledges that this Scheme binds WCL and all Scheme Participants (d) (including those who do not attend the Scheme Meeting and those who do not vote, or vote against this Scheme, at the Scheme Meeting); and
- (e) consents to WCL, TH Newco and Unibail doing all things and executing all deeds, instruments, transfers or other documents as may be necessary or desirable to give full effect to this Scheme and the transactions contemplated by it.

5.6 **Warranty by Scheme Participants**

Each Scheme Participant warrants to Unibail and is deemed to have authorised WCL to warrant to Unibail as agent and attorney for the Scheme Participant by virtue of this clause 5.6, that:

- (a) all their Scheme Shares (including any rights and entitlements attaching to those securities) transferred to Unibail under this Scheme will, as at the date of the transfer, be fully paid and free from all Encumbrances;
- they have full power and capacity to sell and to transfer their Scheme (b) Shares (including any rights and entitlements attaching to those securities) to Unibail under this Scheme.

5.7 Transfer free of encumbrances

To the extent permitted by law, all Scheme Shares and TH Newco Shares (including any rights and entitlements attaching to those securities) which are

transferred to Unibail under this Scheme will, at the date of the transfer of them. to Unibail, vest in Unibail free from all Encumbrances and interests of third parties of any kind, whether legal or otherwise, and free from any restrictions on transfer of any kind not referred to in this Scheme.

5.8 Appointment of Unibail as sole proxy

Subject to the provision of the Scheme Consideration for the Scheme Shares as contemplated by clause 6 of this Scheme, on and from the Implementation Date until WCL registers Unibail as the holder of all of the WCL Scheme Shares in the Share Register, each Scheme Participant:

- irrevocably appoints WCL as attorney and agent (and directs WCL in (a) such capacity) to appoint Unibail and each of its directors from time to time (jointly and each of them individually) as its sole proxy, and where applicable corporate representative, to attend securityholders' meetings, exercise the votes attaching to WCL Shares registered in its name and sign any securityholders resolution, and no Scheme Participant may itself attend or vote at any of those meetings or sign any resolutions, whether in person, by proxy or by corporate representative (other than pursuant to this clause 5.8(a)); and
- (b) must take all other actions in the capacity of the registered holder of WCL Shares as Unibail directs.

WCL undertakes in favour of each Scheme Participant that it will appoint Unibail and each of its directors from time to time (jointly and each of them individually) as that Scheme Participant's proxy or, where applicable, corporate representative in accordance with clause 5.8(a) of this Scheme.

5.9 **Transaction Steps**

- WCL must implement, and must procure that WAT and WFDT (a) implements, the Transaction Steps.
- (b) Unibail must implement or procure the implementation of the Transaction Steps.
- (c) The Third Party Nominee must execute or provide duly executed powers of attorney to such person(s) and in such form as indicated by Unibail for the execution of the TH Newco Transfer Form.

6 Scheme Consideration

6.1 **Scheme Consideration**

The Scheme Consideration to be provided to each Scheme Participant in respect of each of their Scheme Shares will be as follows:

- (a) subject to clause 6.2, a cash amount which, when aggregated with the 'Cash Consideration' (as defined in the WAT Constitutional Amendment), equals US\$2.67 which will be paid in US\$, unless the Scheme Participant makes a valid Currency Election in accordance with clause 6.3 in which case the payment will be in A\$ in such amount as determined in accordance with clause 6.3 (the **Cash Consideration**); and
- (b) subject to clauses 6.2, 6.4, 6.5, 6.6 and 6.7, 0.01844 New Unibail Shares (the Scrip Consideration).

6.2 Fractions and splitting

- (a) Where the calculation of the aggregate number of New Unibail Shares or New Unibail CDIs to be issued to a particular Scheme Participant would result in the issue of a fraction of a New Unibail Share or New Unibail CDI (after applying the Scheme Participant's entitlement (prior to rounding) to all of its Scheme Shares), the aggregate number will, subject to clause 6.2(b), be rounded down to the nearest whole number of New Unibail Shares or New Unibail CDIs.
- (b) If the number of New Unibail Shares or New Unibail CDIs to which a Scheme Participant would be entitled pursuant to clause 6.2(a) is different to the number of Newco BI Shares or Newco BI CDIs that that Scheme Participant is entitled to under the WAT Trust Scheme, the number of New Unibail Shares or New Unibail CDIs that that Scheme Participant will be entitled to under this Scheme will be the same as the number of Newco BI Shares or Newco BI CDIs that that Scheme Participant is entitled to under the WAT Trust Scheme.
- (c) Any cash amount payable to a Scheme Participant under this Scheme (after applying the Scheme Participant's entitlement (prior to rounding) to its entire holding of Scheme Shares), when aggregated with any cash amount payable to the Scheme Participant under the WAT Trust Scheme, must be rounded to the nearest whole cent.
- (d) If Unibail is of the opinion, formed reasonably, that several Scheme Participants, each of which holds a holding of Scheme Shares have, before the Record Date, been party to a shareholding splitting or division in an attempt to obtain an advantage by reference to the rounding provided for in the calculation of each Scheme Participant's entitlement to the Scheme Consideration, Unibail may direct Westfield to give notice (and if Westfield is given such a direction it must give notice) to those Scheme Participants:
 - (i) setting out the names and Registered Addresses of all of them;
 - (ii) stating that opinion; and
 - (iii) attributing to one of them specifically identified in the notice the Scheme Shares held by all of them,

and, after the notice has been so given, the Scheme Participant specifically identified in the notice shall, for the purposes of this Scheme, be taken to hold all those Scheme Shares and each of the other Scheme Participants whose names are set out in the notice shall, for the purposes of this Scheme, be taken to hold no Scheme Shares.

6.3 Currency election

- (a) Subject to clause 6.3(c), each Scheme Participant whose Registered Address as at the Record Date is within Australia is deemed to have made an election to receive all of their Cash Consideration in the form of Australian dollars.
- (b) Subject to clause 6.3(c), each Scheme Participant whose Registered Address as at the Record Date is outside Australia is deemed to have made an election to receive their Cash Consideration in the form of US dollars.
- (c) A Scheme Participant whose Registered Address as at the Record Date is within Australia may make an election to receive all (or, where clause 6.3(d) applies, some) of their Cash Consideration in the form of US

dollars, and a Scheme Participant whose Registered Address as at the Record Date is outside Australia may make an election to receive all (or, where clause 6.3(d) applies, some) of their Cash Consideration in the form of Australian dollars. Such an election will be valid only if the following requirements are satisfied (or waived by Unibail in its sole discretion in respect of any particular Scheme Participant):

- completing and signing an Election Form in accordance with the (i) instructions on the Election Form; and
- (ii) the Election Form is received by the Unibail Registry before the Election Time at the address specified by Unibail on the Election Form.
- (d) If a Scheme Participant is noted on the Share Register as holding one or more parcels of WCL Shares as trustee or nominee for, or otherwise on account of, another person, Unibail may in its sole discretion allow the Scheme Participant to make an election under clause 6.3(c) in relation to each of those parcels of WCL Shares, and an election made in respect of any such parcel, or an omission to make an election in respect of any such parcel, will not be taken to extend to the other parcels. The manner in which such election is to be made, and the information to be provided to Unibail in relation to that, is to be as advised by Unibail to the Scheme Participant. Whether or not such an election is accepted is in the sole discretion of Unibail.
- (e) Subject to clause 6.3(d), a Currency Election made or deemed to be made by a Scheme Participant under this clause 6.3 will be deemed to apply in respect of the Scheme Participant's entire registered holding of Scheme Shares on the Record Date, regardless of whether the Scheme Participant's holding of Scheme Shares on the Record Date is greater or less than the Scheme Participant's holding at the time it made its Currency Election.
- An election made under clause 6.3(c) may be revoked in accordance (f) with the instructions on the Election Form.
- Unibail will determine, in its sole discretion, all questions as to the correct (g) completion of an Election Form, and time of receipt of an Election Form. Unibail is not required to communicate with any Scheme Participant prior to making this determination. The determination of Unibail will be final and binding on the Scheme Participant.
- All Cash Consideration payable to a Scheme Participant in the form of (h) A\$ will be based on the US\$/A\$ exchange rate on the Record Date, being such market-published rate selected by Unibail.
- (i) Notwithstanding anything else in this Scheme:
 - no Cash Consideration will be paid to a Scheme Participant in (i) the form of A\$ if that Scheme Participant is to receive cash consideration under the WAT Trust Scheme in the form of US\$ rather, the Scheme Participant will receive Cash Consideration in the form of US\$: and
 - no Cash Consideration will be paid to a Scheme Participant in (ii) the form of US\$ if that Scheme Participant is to receive cash consideration under the WAT Trust Scheme in the form of A\$ rather, the Scheme Participant will receive Cash Consideration in the form of A\$.

Scrip Consideration 6.4

- If the Registered Address of a Scheme Participant as at the Record Date is within Australia and the Scheme Participant is not an Ineligible Foreign
 - (i) the New Unibail Shares to which the Scheme Participant is entitled under this Scheme will be in the form of New Unibail CDIs; and
 - (ii) the Scheme Participant will receive New Unibail CDIs.

However, if either:

- the condition precedent in clause 3.1(n)(ii) of the Implementation (iii) Agreement is waived in accordance with the Implementation Agreement; or
- (iv) Unibail in its sole discretion determines to do so following a valid election made by the Scheme Participant in accordance with clause 6.4(c),

that Scheme Participant will receive their Scrip Consideration in the form of New Unibail Shares.

- (b) If the Registered Address of a Scheme Participant as at the Record Date is outside Australia and the Scheme Participant is not an Ineligible Foreign Holder the Scheme Participant will receive their Scrip Consideration in the form of New Unibail Shares. However, if Unibail in its sole discretion determines to do so following a valid election made by that Scheme Participant in accordance with clause 6.4(c), that Scheme Participant will receive their Scrip Consideration in the form of New Unibail CDIs.
- An election by a Scheme Participant to receive their Scrip Consideration (c) in a form other than as prescribed by clause 6.4(a) or 6.4(b) (as applicable) (being a Scrip Election) will be valid only if the following requirements are satisfied to Unibail's sole satisfaction (or waived by Unibail in its sole discretion) in respect of any particular Scheme Participant:
 - (i) completing and signing an election form (that will be provided by WCL to the Scheme Participant upon request) in accordance with the instructions on the election form; and
 - (ii) the election form is received by the Unibail Registry before the Election Time at the address specified by Unibail on the election form.
- (d) If a Scheme Participant is noted on the Share Register as holding one or more parcels of WCL Shares as trustee or nominee for, or otherwise on account of, another person, Unibail may in its sole discretion allow the Scheme Participant to make an election under clause 6.4(c) in relation to each of those parcels of WCL Shares, and an election made in respect of any such parcel, or an omission to make an election in respect of any such parcel, will not be taken to extend to the other parcels. The manner in which such election is to be made, and the information to be provided to Unibail in relation to that, is to be as advised by Unibail to the Scheme Participant. Whether or not such an election is accepted is in the sole discretion of Unibail.

- Subject to clause 6.4(d), an election made or deemed to be made under (e) this clause 6.4 will be deemed to apply in respect of a Scheme Participant's entire registered holding of Scheme Shares at the Record Date, regardless of whether the Scheme Participant's holding of Scheme Shares at the Record Date is greater or less than the Scheme Participant's holding at the time it made its election under clause 6.4(c).
- An election made under clause 6.4(c) may be revoked in accordance (f) with the instructions on the election form.
- Unibail will determine, in its sole discretion, all questions as to the correct (g) completion of an election form, and time of receipt of an election form. Unibail is not required to communicate with any Scheme Participant prior to making this determination. The determination of Unibail will be final and binding on the Scheme Participant.
- (h) Notwithstanding anything else in this Scheme:
 - no New Unibail Shares will be issued to a Scheme Participant (i) under this Scheme if that Scheme Participant has received Newco BI CDIs under the WAT Trust Scheme – rather, such Scheme Participant's Scrip Consideration will be in the form of New Unibail CDIs; and
 - (ii) no New Unibail CDIs will issued to a Scheme Participant under this Scheme if that Scheme Participant's scrip consideration under the WAT Trust Scheme was in the form of Newco BI Shares – rather, the Scheme Participant will receive New Unibail Shares.

Minimum Holders 6.5

- (a) Subject to clause 6.5(b), Unibail will be under no obligation to issue or provide, and must not issue or provide, any New Unibail CDIs or New Unibail Shares under this Scheme to any Minimum Holder and, instead, clause 6.7 will apply.
- (b) This clause 6.5 only applies if Westfield obtains any necessary relief from the operation of section 601FC(1)(d) of the Corporations Act to enable it to apply.

6.6 **Ineligible Foreign Holders**

Unibail will be under no obligation to issue or provide, and must not issue or provide, any New Unibail CDIs or New Unibail Shares under this Scheme to any Ineligible Foreign Holder and, instead, clause 6.7 will apply.

6.7 Sale Facility

In respect of each Scheme Participant which is a Minimum Holder (if clause 6.5 applies) or an Ineligible Foreign Holder:

Unibail must not issue or provide any New Unibail CDIs or New Unibail (a) Shares that would otherwise be issued or provided to the Scheme Participant under this Scheme and, instead, Unibail must on the Implementation Date issue to the Sale Agent the New Unibail Shares that would otherwise have been issued to or in respect of the Scheme Participant (and all such New Unibail Shares together with the Newco BI Shares to which they will be stapled shall be referred to as the Sale Securities).

- (b) Unibail must procure that as soon as practicable after the Implementation Date and, in any event, not more than 20 Business Days after the Implementation Date:
 - the Sale Agent, in consultation with Unibail, sells or procures the (i) sale (including on an aggregated or partially aggregated basis), on Euronext Paris and/or Euronext Amsterdam, of all the Sale Securities in such manner, at such price or prices and on such other terms as the Sale Agent determines in good faith; and
 - (ii) remits to WCL in US dollars the proceeds of the sales, after deduction of any applicable brokerage, taxes, duty, currency conversion costs and other costs and charges (converted into US dollars in such manner, at such price or prices and on such other terms as the Sale Agent determines in good faith) (Net Sale Proceeds).
- (c) Within 10 Business Days after receiving the Net Sale Proceeds in respect of the sale of all of the Sale Securities, WCL must pay, or procure the payment, to each Selling Scheme Participant (in accordance with this clause 6.7) an amount calculated as follows:

$A \div B \times C$

Where:

A = the Net Sale Proceeds;

B = the total number of New Unibail Shares issued to the Sale Agent under clause 6.7(a).

C = the number of New Unibail Shares issued to the Sale Agent under clause 6.7(a) in respect of that Selling Scheme Participant (which, for the avoidance of doubt, must not include a fraction of a New Unibail Share).

Such payment represents each Selling Scheme Participant's entitlement to part of the Net Sale Proceeds under both this Scheme and the WAT Trust Scheme.

- (d) None of Westfield. Unibail or the Sale Agent gives any assurance as to the price that will be achieved for the sale of Sale Securities described in clause 6.7(b). The sale of Sale Securities under this clause 6.7 will be at the risk of the Selling Scheme Participant.
- WCL must make payments to Selling Scheme Participants under clause (e) 6.7(c) by either (in the absolute discretion of WCL):
 - (i) where a Selling Scheme Participant has, before the Record Date, made a valid election in accordance with the requirements of the Westfield Registry to receive distribution payments from Westfield by electronic funds transfer to a bank account nominated by the Selling Scheme Participant denominated in US dollars, paying, or procuring the payment of, the relevant amount in US dollars by electronic means in accordance with that election; or
 - otherwise, whether or not the Selling Scheme Participant has (ii) made an election referred to in clause 6.7(e)(i), dispatching, or procuring the dispatch of, a cheque for the relevant amount in US dollars to the Selling Scheme Participant by prepaid post to their Registered Address (as at the Record Date), such cheque being drawn in the name of the Selling Scheme Participant (or in the case of joint holders, in accordance with the procedures set out in clause 6.10).

- If WCL receives professional advice that any withholding or other tax is (f) required by law to be withheld from a payment to a Selling Scheme Participant, WCL is entitled to withhold the relevant amount before making the payment to the Selling Scheme Participant (and payment of the reduced amount shall be taken to be full payment of the relevant amount for the purposes of this Scheme, including clause 6.7(c)). WCL must pay any amount so withheld to the relevant taxation authorities within the time permitted by law, and, if requested in writing by the relevant Selling Scheme Participant, provide a receipt or other appropriate evidence of such payment (or procure the provision of such receipt or other evidence) to the relevant Selling Scheme Participant.
- Each Selling Scheme Participant appoints WCL as its agent to receive (g) on its behalf any financial services guide or other notices (including any updates of those documents) that the Sale Agent is required to provide to Selling Scheme Participant under the Corporations Act.

6.8 **Provision of Cash Consideration**

- (a) Unibail must, and WCL must use its best endeavours to procure that Unibail does by no later than the Business Day before the Implementation Date, deposit in cleared funds an amount equal to:
 - (i) the aggregate amount in US dollars of the Cash Consideration which is payable to Scheme Participants in US dollars by operation of clause 6.3, into a US dollar denominated trust account operated by WCL as trustee for the Scheme Participants (provided that any interest on the amounts deposited (less bank fees and other charges) will be credited to Unibail's account): and
 - the aggregate amount in Australian dollars of the Cash (ii) Consideration which is payable to Scheme Participants in Australian dollars by operation of clause 6.3 into an Australian dollar denominated trust account operated by WCL as trustee for the Scheme Participants (provided that any interest on the amounts deposited (less bank fees and other charges) will be credited to Unibail's account).
- (b) Upon Unibail having complied with its obligations in clause 6.8(a), Unibail will be taken to have satisfied its obligations to provide or procure the provision of Cash Consideration to each Scheme Participant.
- On the Implementation Date, subject to funds having been deposited in (c) accordance with clause 6.8(a), WCL must pay or procure the payment, from the trust accounts referred to in clause 6.8(a), to each Scheme Participant such amount of cash from the cash component of the Scheme Consideration as that Scheme Participant is entitled under this clause 6 (if any).
- (d) The obligations of WCL under clause 6.8(b) will be satisfied by WCL (in its absolute discretion):
 - (i) where a Scheme Participant has, before the Record Date, made a valid election in accordance with the requirements of the Westfield Registry to receive distribution payments from WCL by electronic funds transfer to a bank account nominated by the Scheme Participant which is denominated in the currency specified in a Currency Election, paying, or procuring the payment of, the relevant amount in the currency of that Scheme Participant's Currency Election by electronic means in accordance with that election: or

- otherwise, whether or not the Scheme Participant has made an (ii) election referred to in clause 6.8(d)(i), dispatching, or procuring the dispatch of, a cheque for the relevant amount in the currency of that Scheme Participant's Currency Election to the Scheme Participant by prepaid post to their Registered Address (as at the Record Date), such cheque being drawn in the name of the Scheme Participant (or in the case of joint holders, in accordance with the procedures set out in clause 6.10).
- To the extent that, following satisfaction of WCL's obligations under (e) clause 6.8(c) and 6.11, there is a surplus in the amount held by WCL as trustee for the Scheme Participants in the trust account referred to in clause 6.8(a), that surplus shall be paid by WCL to Unibail.
- (f) If, following satisfaction of Unibail's obligations under clause 6.8(a) but prior to the occurrence of all of the events described in clause 5.2, this Scheme lapses under 4.2(b):
 - (i) WCL must immediately repay (or cause to be repaid) to or at the direction of Unibail the funds that were deposited in the bank accounts described in clause 6.8(a) plus any interest on the amounts deposited (less bank fees and other charges);
 - (ii) the obligation to transfer Scheme Shares, together with all rights and entitlements attaching to the Scheme Shares as at the Implementation Date, to Unibail under clause 5.2 will immediately cease;
 - (iii) Unibail must return the WCL Transfer Forms, if provided pursuant to clause 5.2(a)(i), and Westfield is no longer obliged to enter, or procure the entry of, the name of Unibail in the Share Register in accordance with clause 5.2(b); and
 - if the TH Newco Transfer Form has been executed by the Third (iv) Party Nominee, Unibail and TH Newco in accordance with clause 5.2(b). Unibail must ensure that the transfer of TH Newco Scheme Shares to Unibail does not take effect or is reversed.

Provision of Scrip Consideration 6.9

- Subject to clauses 6.2, 6.4, 6.5, 6.6 and 6.7, the obligation of Unibail to (a) issue or provide the Scrip Consideration to each Scheme Participant will be satisfied:
 - in the case of Scrip Consideration that is required to be provided (i) to each Scheme Participant in the form of New Unibail Shares, by Unibail procuring that on the Implementation Date the name and address of each such Scheme Participant is entered in the Unibail Registry in respect of those New Unibail Shares; and
 - in the case of Scrip Consideration that is required to be provided (ii) to each Scheme Participant in the form of New Unibail CDIs, by Unibail procuring that, on the Implementation Date:
 - the New Unibail Shares that will underlie the New (A) Unibail CDIs are issued to CDN:
 - (B) CDN issues to each such Scheme Participant the New Unibail CDIs; and

- the name of each such Scheme Participant is entered in (C) the records maintained by CDN as the holder of those New Unibail CDIs; and
- in respect of each Scheme Participant which is a Minimum (iii) Holder (if clause 6.5 applies) or an Ineligible Foreign Holder, by Unibail complying with clause 6.7(a).
- (b) Unibail must procure that, on or before the date that is two Business Days after the Implementation Date, a securities certificate or holding statement is sent to the Registered Address of each Scheme Participant to whom New Unibail Shares or New Unibail CDIs are issued or provided in accordance with clause 6.9(a) representing the number of New Unibail Stapled Securities or New Unibail CDIs stapled to Newco BI CDIs issued or provided to that Scheme Participant pursuant to this Scheme and the WAT Trust Scheme.

6.10 Joint holders

In the case of Scheme Shares held in joint names:

- (a) subject to clause 6.8(d), any amount comprising the cash component of the Scheme Consideration payable in respect of those Scheme Shares is payable to the joint holders and any cheque required to be sent under this Scheme will be made payable to the joint holders and sent to either, at the sole discretion of WCL, the holder whose name appears first in the Share Register as at the Record Date or to the joint holders:
- (b) any New Unibail Shares or New Unibail CDIs to be issued or provided under this Scheme must be issued or provided to and registered in the names of the joint holders; and
- (c) any document required to be sent under this Scheme, will be forwarded to either, at the sole discretion of WCL, the holder whose name appears first in the Share Register as at the Record Date or to the joint holders.

6.11 **Unclaimed monies**

- (a) WCL may cancel a cheque issued under this clause 6 if the cheque:
 - is returned to WCL; or (i)
 - (ii) has not been presented for payment within six months after the date on which the cheque was sent.
- (b) During the period of 12 months commencing on the Implementation Date, on request in writing from a Scheme Participant to WCL (or the Westfield Registry) (which request may not be made earlier than the date which is 20 Business Days after the Implementation Date), WCL must reissue a cheque that was previously cancelled under this clause 6.11.
- The Unclaimed Money Act 1995 (NSW) will apply in relation to any (c) amount payable to a Scheme Participant under this Scheme which becomes 'unclaimed money' (as defined in section 7 of the Unclaimed Money Act 1995 (NSW)).

6.12 Orders of a court or Regulatory Authority

(a) WCL may deduct and withhold from any consideration which would otherwise be payable by WCL to a Scheme Participant in accordance with this clause 6, any amount which Unibail and WCL determine is

- required to be deducted and withheld from that consideration under any applicable law, including any order, direction or notice made or given by a court of competent jurisdiction or by another Regulatory Authority.
- To the extent that amounts are so deducted or withheld, such deducted (b) or withheld amounts will be treated for all purposes under this Scheme as having been paid to the person in respect of which such deduction and withholding was made, provided that such deducted or withheld amounts are actually remitted to the appropriate taxing agency.
- (c) If written notice is given to WCL (or the Westfield Registry) of an order, direction or notice made or given by a court of competent jurisdiction or by another Regulatory Authority that:
 - (i) requires consideration which would otherwise be payable or provided to a Scheme Participant by WCL in accordance with this clause 6 must instead be paid or provided to a Regulatory Authority or other third party (either through payment of a sum or the issuance of a security), then WCL shall be entitled to procure that payment or provision of that consideration is made in accordance with that order, direction or notice (and payment or provision of that consideration in accordance with that order, direction or notice will be treated for all purposes under this Scheme as having been paid or provided to that Scheme Participant); or
 - (ii) prevents WCL from providing consideration to any particular Scheme Participant in accordance with this clause 6, or the payment or provision of such consideration is otherwise prohibited by applicable law, WCL shall be entitled to (as applicable):
 - retain an amount equal to the cash component of the (A) Scheme Consideration to which that Scheme Participant would otherwise be entitled to under this clause 6; and
 - (B) direct Unibail not to issue or provide (or procure the issue or provision of), or to issue or provide to a trustee or nominee, such number of New Unibail Shares or New Unibail CDIs as that Scheme Participant would otherwise be entitled to under this clause 6,

until such time as payment or provision of the consideration in accordance with this clause 6 is permitted by that order or direction or otherwise by law.

6.13 Status of New Unibail Shares and New Unibail CDIs

Subject to this Scheme becoming Effective, Unibail must:

- (a) issue the Unibail Shares such that each such share will rank equally in all respects with all other Unibail Shares which are on issue at the time;
- ensure that each such Unibail Share is duly and validly issued in (b) accordance with all applicable laws and the constituent and organisational documents of each entity, fully paid and free from any Encumbrance: and
- (c) use all reasonable endeavours to ensure that:

- the New Unibail Stapled Security of which each New Unibail (i) Share issued under this Scheme is a part is, from the Business Day following the date this Scheme becomes Effective (or such later date as the relevant exchange requires), quoted for trading on Euronext Paris and Euronext Amsterdam; and
- (ii) each New Unibail CDI issued under this Scheme is, from the Business Day following the date this Scheme becomes Effective (or such later date as ASX requires) and stapled together with a Newco BI CDI, quoted for trading on the ASX initially on a deferred settlement basis and thereafter on an ordinary settlement basis.

7 **Dealings in Scheme Shares**

7.1 **Determination of Scheme Shares**

To establish the identity of the Scheme Participants, dealings in WCL Scheme Shares will only be recognised by WCL if:

- (a) in the case of dealings of the type to be effected using CHESS, the transferee is registered in the Share Register as the holder of the relevant WCL Scheme Shares on or before the Record Date: and
- (b) in all other cases, registrable transmission applications or transfers in registrable form in respect of those dealings are received on or before the Record Date at the place where the Share Register is kept.

7.2 Register

Westfield must register any registrable transmission applications or transfers of WCL Shares received in accordance with clause 7.1(b) of this Scheme on or before the Record Date.

7.3 No disposals after Effective Date

- (a) If this Scheme becomes Effective, a holder of Scheme Shares (and any person claiming through that holder) must not dispose of or purport or agree to dispose of any Scheme Shares or any interest in them after the Effective Date in any way except as set out in this Scheme and any such disposal will be void and of no legal effect whatsoever.
- Neither Westfield nor TH Newco will accept for registration or recognise (b) for any purpose any transmission, application or transfer in respect of Scheme Shares received after the Record Date (except a transfer to Unibail pursuant to this Scheme or any subsequent transfer by Unibail or its successors in title).

7.4 **Maintenance of Share Register**

For the purpose of determining entitlements to the Scheme Consideration, WCL will maintain the Share Register in accordance with the provisions of this clause 7.4 until the Scheme Consideration has been provided to the Scheme Participants and Unibail has been entered in the Share Register as the holder of all the WCL Scheme Shares. The Share Register in this form will solely determine entitlements to the Scheme Consideration.

7.5 Effect of certificates and holding statements

Subject to provision of the Scheme Consideration and registration of the transfers to Unibail contemplated in clauses 5.2 and 7.4 of this Scheme, any statements of holding in respect of WCL Scheme Shares will cease to have effect after the Record Date as documents of title in respect of those shares

Scheme of Arrangement 25 © King & Wood Mallesons

(other than statements of holding in favour of Unibail and its successors in title). After the Record Date, each entry current on the Share Register as at the Record Date (other than entries in respect of Unibail or its successors in title) will cease to have effect except as evidence of entitlement to the Scheme Consideration.

7.6 **Details of Scheme Participants**

Within 3 Business Days after the Record Date, WCL will ensure that details of the names, Registered Addresses and holdings of WCL Scheme Shares for each Scheme Participant, as shown in the Share Register at the Record Date are available to Unibail in such form as Unibail reasonably requires.

7.7 **Quotation of WCL Shares**

Suspension of trading on ASX in WCL Shares will occur from the close of trading on ASX on the Effective Date.

7.8 Termination of quotation of WCL Shares

After the Scheme has been implemented, WCL will apply:

- for termination of the official quotation of WCL Shares on ASX; and (a)
- to have itself removed from the official list of the ASX. (b)

8 Power of attorney

Each Scheme Participant, without the need for any further act by any Scheme Participant, irrevocably appoints WCL and each of its directors and secretaries (jointly and each of them individually) as its attorney and agent for the purpose of:

- executing any document necessary or expedient to give effect to this (a) Scheme and the Transaction Steps including:
 - (i) executing any Transfer Forms; and
 - applying for shares in (and agreeing to become a member of) (ii) any of the entities in which Scheme Participants are to become members in accordance with the Transaction Steps; and
- (b) enforcing the Deed Poll against Unibail; and
- enforcing the Third Party Nominee Deed Poll against the Third Party (c) Nominee,

and WCL accepts such appointment.

Notices 9

9.1 No deemed receipt

If a notice, transfer, transmission application, direction or other communication referred to in this Scheme is sent by post to WCL, it will not be taken to be received in the ordinary course of post or on a date and time other than the date and time (if any) on which it is actually received at WCL's registered office or at the office of the Westfield Registry.

9.2 **Accidental omission**

The accidental omission to give notice of the Scheme Meeting or the non-receipt of such a notice by any WCL Shareholder will not, unless so ordered by the

Court, invalidate the Scheme Meeting or the proceedings of the Scheme Meeting.

10 General

10.1 Variations, alterations and conditions

WCL may, with the consent of Unibail, by its counsel or solicitor consent on behalf of all persons concerned to any variations, alterations or conditions to this Scheme which the Court thinks fit to impose.

10.2 Further action by WCL

WCL will execute all documents and do all things (on its own behalf and on behalf of each Scheme Participant) necessary or expedient to implement, and perform its and each Scheme Participant's obligations under, this Scheme and the Transaction Steps.

10.3 No liability when acting in good faith

Neither WCL nor Unibail, nor any of their respective officers, will be liable for anything done or omitted to be done in the performance of this Scheme in good faith.

10.4 Enforcement of Deed Poll

WCL undertakes in favour of each Scheme Participant to enforce the Deed Poll against Unibail on behalf of and as agent and attorney for the Scheme Participants.

10.5 Stamp duty

Unibail will pay all stamp duty (including any fines, penalties and interest) payable in respect of this Scheme.

11 Governing law

11.1 Governing law and jurisdiction

The law in force in the place specified in the Details governs this document. WCL submits to the non-exclusive jurisdiction of the courts of that place.

11.2 Serving documents

Without preventing any other method of service, any document in an action in connection with this document may be served on a party by being delivered or left at that party's address set out in the Details.

Implementation Agreement

Annexure C Deed Poll

© King & Wood Mallesons Implementation Agreement 92



Dated

Given by Unibail-Rodamco SE ("Unibail")

In favour of each registered holder of Westfield Stapled Securities as at the Record Date ("Scheme Participants")

King & Wood Mallesons

Level 61
Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia
T +61 2 9296 2000
F +61 2 9296 3999
DX 113 Sydney
www.kwm.com

Contents

Details		
Gene	eral terms	2
1	Definitions and interpretation	2
1.1	Definitions	2
1.2	General interpretation	2
1.3	Nature of deed poll	2
2	Conditions precedent and termination	2
2.1	Conditions precedent	2
2.2	Termination	2
2.3	Consequences of termination	3
3	Scheme Consideration	3
3.1	Scheme Consideration	3
4	Representations and warranties	3
5	Continuing obligations	4
6	Stamp duty	4
7	Notices	4
8	General	4
8.1	Variation	4
8.2	Partial exercising of rights	5
8.3	Remedies cumulative	5
8.4	Assignment or other dealings	5
8.5	Further steps	5
9	Governing law and jurisdiction	5
9.1	Governing law and jurisdiction	5
9.2	Serving documents	5
9.3	Service of process	5
Signi	ing page	7

Details

Parties

Unibail	Name		Unibail-Rodamco SE	
	Registration number		682 024 096	
	Formed in		France	
	Addres	SS	7, place du Chancelier Adenauer, 75016 Paris France	
	Email		David.Zeitoun@unibail-rodamco.com	
	Attenti	on	David Zeitoun	
In favour of	Each registered holder of Westfield Stapled Securities as at the Record Date.			
Governing law	New South Wales			
should propose the Share Scheme and the Share Schem		ors of Westfield have resolved that Westfield pose the Share Scheme and the Trust Schemes.		
		of the Share Scheme and the Trust Schemes will Westfield Stapled Securities will be transferred to return for the Scheme Consideration.		
	С	Westfield and Unibail have entered into the Implementation Agreement.		
	D	Unibail is entering into this deed poll for the purpose of covenanting in favour of Scheme Participants to perform obligations in relation to the Share Scheme, the Trust Schemes, the Implementation Agreement and the Transaction Steps.		

General terms

1 Definitions and interpretation

1.1 Definitions

Unless the contrary intention appears, these meanings apply:

Implementation Agreement means the implementation agreement dated 12 December 2017 between WCL, WAML as responsible entity of WAT and as responsible entity of WFDT and Unibail under which, amongst other things, WCL has agreed to propose the Share Scheme and Trust Schemes to Westfield Securityholders, and each of WCL, WAT, WFDT and Unibail has agreed to take certain steps to give effect to the Share Scheme and Trust Schemes (including the Transaction Steps).

All other words and phrases used in this document have the same meaning as given to them in the Implementation Agreement.

1.2 General interpretation

Clause 1.2 of the Implementation Agreement applies to this document.

1.3 Nature of deed poll

Unibail acknowledges that:

- (a) this document may be relied on and enforced by any Scheme Participant in accordance with its terms even though the Scheme Participants are not a party to it;
- (b) under the Share Scheme, each Scheme Participant irrevocably appoints WCL and each of its directors, officers and secretaries (jointly and each of them severally) as its agent and attorney to enforce this document against Unibail; and
- (c) under the Trust Schemes, each Scheme Participant irrevocably appoints WAML and each of its directors, officers and secretaries (jointly and each of them severally) as its agent and attorney to enforce this document against Unibail.

2 Conditions precedent and termination

2.1 Conditions precedent

Unibail's obligations under this document are subject to the Share Scheme and the Trust Schemes becoming Effective.

2.2 Termination

Unibail's obligations under this document will automatically terminate and the terms of this document will be of no further force or effect if:

- (a) the Share Scheme and the Trust Schemes have not become Effective on or before the End Date; or
- (b) the Implementation Agreement is terminated in accordance with its terms.

2.3 Consequences of termination

If this document is terminated under clause 2.2, then, in addition and without prejudice to any other rights, powers or remedies available to Scheme Participants:

- (a) Unibail is released from its obligations to further perform this document except those obligations contained in clause 6; and
- (b) each Scheme Participant retains the rights, powers or remedies they have against Unibail in respect of any breach of this document which occurs before it is terminated.

3 Scheme Consideration

3.1 Scheme Consideration

Subject to clause 2, Unibail undertakes in favour of each Scheme Participant to:

- (a) provide, or procure the provision of, the Scheme Consideration to each Scheme Participant, in accordance with the terms of the Share Scheme and the Trust Schemes:
- (b) procure that TH Newco, Newco BI and US Newco undertake all actions attributed to them under the Share Scheme and the Trust Schemes,

subject to and in accordance with the terms of the Share Scheme and the Trust Schemes.

4 Representations and warranties

Unibail represents and warrants that:

- (a) (status) it has been incorporated or formed in accordance with the laws of its place of incorporation or formation, is validly existing under those laws and has power and authority to own its assets and carry on its business as it is now being conducted;
- (b) (**power**) it has power to enter into this document, to comply with its obligations under it and exercise its rights under it;
- (c) (no contravention) the entry by it into, its compliance with its obligations and the exercise of its rights under, this document do not and will not conflict with:
 - (i) its constituent documents or cause a limitation on its powers or the powers of its directors to be exceeded; or
 - (ii) any law binding on or applicable to it or its assets;
- (d) **(authorisations)** other than the approvals specified in the Conditions Precedent, it has in full force and effect each authorisation necessary for

© King & Wood Mallesons Deed Poll 3

it to enter into this document, to comply with its obligations and exercise its rights under it, and to allow them to be enforced;

- (e) (validity of obligations) its obligations under this document are valid and binding and are enforceable against it in accordance with its terms; and
- (f) (solvency) is not Insolvent.

5 Continuing obligations

This document is irrevocable and, subject to clause 2, remains in full force and effect until:

- (a) Unibail has fully performed its obligations under this document; or
- (b) the earlier termination of this document under clause 2.2.

6 Stamp duty

Unibail:

- agrees to pay or reimburse all stamp duty, registration fees and similar taxes payable or assessed as being payable in connection with this document or any other transaction contemplated by this document (including any fees, fines, penalties and interest in connection with any of these amounts); and
- (b) indemnifies each Scheme Participant against, and agrees to reimburse and compensate it, for any liability in respect of stamp duty under clause 6(a).

7 Notices

Notices and other communications in connection with this document must be in writing. They must be sent to the address or email address referred to in the Details and (except in the case of email) marked for the attention of the person referred to in the Details. If the intended recipient has notified changed contact details, then communications must be sent to the changed contact details.

8 General

8.1 Variation

A provision of this document or any right created under it may not be varied, altered or otherwise amended unless:

- (a) the variation is agreed to by Westfield and Unibail in writing; and
- (b) the Court indicates that the variation, alteration or amendment would not itself preclude approval of the Share Scheme or the grant of the Second Judicial Advice,

in which event Unibail must enter into a further deed poll in favour of the Scheme Participants giving effect to the variation, alteration or amendment.

© King & Wood Mallesons Deed Poll 4

8.2 Partial exercising of rights

Unless this document expressly states otherwise, if Unibail does not exercise a right, power or remedy in connection with this document fully or at a given time, it may still exercise it later.

8.3 Remedies cumulative

The rights, powers and remedies in connection with this document are in addition to other rights, powers and remedies given by law independently of this document.

8.4 Assignment or other dealings

The rights and obligations created by this document are personal to Unibail and each Scheme Participant, and Unibail and each Scheme Participant may not assign or otherwise deal with its rights under this document or allow any interest in them to arise or be varied without the consent of Unibail and Westfield.

8.5 **Further steps**

Unibail agrees to do anything including executing all documents and do all things (on its own behalf or on behalf of each Scheme Participant) necessary to give full effect to this document and the transactions contemplated by it.

9 Governing law and jurisdiction

9.1 Governing law and jurisdiction

The law in force in the place specified in the Details governs this document. Unibail submits to the non-exclusive jurisdiction of the courts of that place.

9.2 Serving documents

Without preventing any other method of service, any document in an action in connection with this document may be served on Unibail by being delivered or left at Unibail's address set out in the Details.

9.3 Service of process

Without preventing any other mode of service, any document in an (a) action (including, any writ of summons or other originating process or any third or other party notice) may be served on Unibail by being delivered to or left for Unibail at the following address:

> Allens Level 28, Deutsche Bank Place 126 Phillip Street Sydney NSW 2000 Australia

Unibail irrevocably appoints Allens as its agent for the service of process (b) in Australia in relation to any matter arising out of this document. If Allens ceases to be able to act as such or have an address in Australia, Unibail agrees to appoint a new process agent in Australia and deliver to Westfield within 2 Business Days a copy of a written acceptance of appointment by the process agent, upon receipt of which the new appointment becomes effective for the purpose of this document. Unibail must inform Westfield in writing of any change in the address of its process agent within 2 Business Days of the change.

Deed Poll 5

EXECUTED as a deed poll

Deed Poll

Signing page

DATED:	
SIGNED by [Name] as authorised representative for UNIBAIL-RODAMCO SE in the presence of:))))
Signature of witness	By executing this document the signatory warrants that the signatory is duly authorised to execute this document on behalf of UNIBAIL-
Name of witness (block letters)	RODAMCO SE

Implementation Agreement

Annexure D Westfield Constitutional Amendments

© King & Wood Mallesons Implementation Agreement 93



Westfield America Trust Supplemental Deed

Dated

Westfield America Management Limited (ABN 66 072 780 619) ("Manager")

King & Wood Mallesons

Level 61
Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia
T +61 2 9296 2000
F +61 2 9296 3999
DX 113 Sydney
www.kwm.com

Westfield America Trust Supplemental Deed

Details

Manager	Name		Westfield America Management Limited
	ABN		66 072 780 619
	Capac	ity	Responsible entity of the WAT
	Addres	ss	Level 29, 85 Castlereagh Street, Sydney NSW 2000, Australia
Governing law	New South Wales		
Recitals	Α	Manager is the responsible entity of the trust known as Westfield America Trust (ARSN 092 058 449) (WAT). WAT is registered as a managed investment scheme under Chapter 5C of the Corporations Act.	
	В		
	C WAT was o		constituted under the Constitution.
	D	WAT is admitted to the official list of the ASX along with WFDT and WCL. One share in WCL and one unit in each of WAT and WFDT are stapled together and are officially quoted on the stock market conducted by ASX as Westfield Stapled Securities.	
	E	Westfield and Unibail have agreed, by executing an implementation agreement dated 12 December 2017 (the Implementation Agreement), to propose and implement the Scheme, the WFDT Trust Scheme and the WCL Scheme to Westfield Securityholders, and each of WAT, WCL, WFDT and Unibail has agreed to take certain steps to give effect to the Scheme (including the Transaction Steps).	
	F The Const		itution must be amended to facilitate the Scheme.
	G	Section 601GC(1) of the Corporations Act provides that the constitution of a registered scheme may be modified, or repealed and replaced with a new constitution:	
		(a) by	y special resolution of the members of the scheme;
		th	y the responsible entity if it reasonably considers e change will not adversely affect members' ghts.
	1.1		40 6 (b 0 6 (c

Under clause 19 of the Constitution, the Constitution may be modified by Resolution and Manager may give effect to the

modification by executing a supplemental deed.

Н



Westfield America Trust Supplemental Deed

General terms

1 Interpretation

1.1 Definitions

In this document, these words and phrases have the following meanings and any other words and phrases have the meaning given to them in the Constitution, unless the contrary intention appears:

Constitution means the document dated 28 March 1996 under which the WAT was constituted as amended from time to time.

Effective Time means the date on and time at which a copy of this document is lodged with the Australian Securities and Investments Commission under section 601GC(2) of the Corporations Act.

Scheme means the arrangement under which US Newco acquires all of the Scheme Units from each Westfield Securityholder as part of the Transaction, facilitated by amendments to this document.

Scheme Unit means a Unit held by the Scheme Participants as at the Scheme Record Date.

Scheme Participant means a person registered in the Register as a holder of one or more Scheme Units as at the Scheme Record Date.

Scheme Record Date means 5.00pm on the fourth Business Day following the Effective Time or such other date as the Manager and Unibail may agree in writing.

Stapling Deed means the stapling deed between WCL and Manager (as responsible entity for WAT and for WFDT) dated 30 June 2014.

TH Newco means a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated or to be incorporated in the Netherlands as referred to in the WFDT Trust Scheme.

Transaction means the acquisition of Westfield by Unibail contemplated by the Implementation Agreement, including the implementation of all of the Transaction Steps in accordance with the Scheme, the WCL Scheme, the WFDT Trust Scheme and the Implementation Agreement.

Transaction Steps means each of the steps set out in Schedule 2 of the Implementation Agreement.

Unibail means Unibail-Rodamco SE (Registration Number 682 024 096).

US Newco means a corporation incorporated or to be incorporated in the United States.

WAT means Westfield America Trust (ARSN 092 058 499).

WCL means Westfield Corporation Limited (ABN 12 166 995 197).

WCL Scheme means the scheme of arrangement under Part 5.1 of the Corporations Act between WCL and each Westfield Securityholder as part of the

Transaction, subject to any alterations or conditions made or required by the Court pursuant to section 411(6) of the Corporations Act to the extent they are approved in writing by WCL and Unibail.

WCL Share means a fully paid ordinary share in the capital of WCL.

Westfield Securityholder means a person who is registered in the Register, WCL share register of members and WFDT unit register of members as the holder of one or more Westfield Stapled Securities.

Westfield Stapled Security means a WCL Share, a Unit and a WFDT Unit stapled to each other in accordance with the provisions of the Stapling Deed, the Constitution and the constitutions of WCL and WFDT.

WFDT means WFD Trust (ARSN 168 765 875).

WFDT Trust Scheme means the arrangement under which TH Newco acquires all of the WFDT Units from each Westfield Securityholder as part of the Transaction, facilitated by amendments to the constitution of WFDT, as set out in the supplemental deed poll under which the Manager will amend the constitution of WFDT.

WFDT Unit means a fully paid ordinary unit in WFDT.

1.2 Interpretation

Clause 1.2 of the Constitution applies to this document as if set out in this document, except that references to "this deed" or "this constitution" are references to this document.

1.3 Document supplemental to Constitution

This document is supplemental to the Constitution.

2 Benefit of this document

This document is made by the Manager with the intent that the benefit of this document shall ensure to the benefit of Members, jointly and severally.

3 Conditions

This document is conditional upon, and will have no force of effect until, the satisfaction of each of the conditions precedent stipulated in clause 3.1 of the WCL Scheme (as defined in the Implementation Agreement) other than the coming into effect of the amendments made by this document.

4 Modifications to the Constitution

The Constitution is modified on and from the Effective Time in the manner set out in Schedule 1.

5 No redeclaration

The Manager is not, by this document:

- (a) redeclaring or resettling WAT;
- (b) declaring any trust; or
- (c) causing the transfer, vesting or accruing of property in any person.

6 Governing law

This document is governed by the laws in force in the place specified in the Details. Each person affected by it must submit to the non-exclusive jurisdiction of the courts of that place and the courts of appeal from them.

EXECUTED as a deed poll	
DATED:	
EXECUTED by WESTFIELD AMERICA MANAGEMENT LIMITED as responsible entity of WESTFIELD AMERICA TRUST in accordance with section 127(1) of the Corporations Act 2001 (Cth) by authority of its directors:))))))))
Signature of director) Signature of director/company) secretary*) *delete whichever is not applicable
Name of director (block letters)	Name of director/company secretary* (block letters) *delete whichever is not applicable

Schedule 1

1 Clause 28

28 Interpretation

28.1 Definitions

In clause 28 and 29 the following definitions are inserted in alphabetical order:

Unless the contrary intention appears, these meanings apply:

Business Day means a business day as defined in the Listing Rules and on which trading will take place on Euronext Paris and Euronext Amsterdam and the New York Stock Exchange.

Cash Consideration has the meaning given in clause 29.10(a).

CDN means CHESS Depository Nominees Pty Limited (ACN 071 346 506).

CHESS means the clearing house electronic sub register system of share transfers operated by ASX Settlement and Transfer Corporation Pty Ltd.

Court means the Federal Court of Australia, Sydney Registry, or such other court as is agreed in writing by Unibail and the Manager.

Currency Election means an election made or deemed to have been made under clause 29.12(a), 29.12(b) or 29.12(c).

Deed Poll means the deed poll dated [insert date] executed by Unibail under which Unibail covenants in favour of each Scheme Participant to perform its obligations under the Scheme.

Effective Date means the date and time at which the supplemental deed making amendments to this constitution to facilitate the Scheme, take effect pursuant to section 601GC(2) of the Corporations Act.

Election Form means an election form that Westfield will provide to any Westfield Securityholder upon request under which the Westfield Securityholder may make a Currency Election.

Election Time means 5.00pm on the first Business Day after the Effective Date or any other date agreed between Unibail and Westfield.

Encumbrance means any mortgage, charge, lien, pledge, trust, power or title retention, flawed deposit arrangement, "security interest" as defined in sections 12(1) or (2) of the *Personal Property Securities Act 2009* (Cth) (Australia), or interest of a third party of any kind, whether legal or otherwise, or any agreement to create any of them or allowing them to exist.

Euronext Amsterdam means the regulated market operated by Euronext Amsterdam NV.

Euronext Paris means the regulated market operated by Euronext Paris S.A.

Implementation Agreement means the implementation agreement dated 12 December 2017 between Westfield and Unibail under which, amongst other things, the Manager has agreed to propose the Scheme to the Members, and each of Westfield and Unibail has agreed to take certain steps to give effect to the Scheme (including the Transaction Steps).

Implementation Date means the fifth Business Day following the Scheme Record Date or such other date after the Scheme Record Date as is agreed in writing by Unibail and the Manager.

Ineligible Foreign Holder means a Scheme Participant whose address shown in the Register on the Scheme Record Date is a place outside Australia and its external territories and New Zealand, unless Unibail determines that it is lawful and not unduly onerous or impracticable to provide that Scheme Participant with Newco BI CDIs or Newco BI Shares under the Scheme.

Listing Rules means the Listing Rules of the ASX.

Minimum Holder means a Scheme Participant who is entitled to receive a parcel of Newco BI Shares under the Scheme and New Unibail Shares under the WCL Scheme which, when stapled together and trading as New Unibail Stapled Securities, will have a value of less than €500 (calculated with reference to the price of Unibail Shares on Euronext Paris as at the close of trading on the Effective Date).

Net Sale Proceeds has the meaning given in clause 29.16(b)(ii).

New Unibail CDI means a New Unibail CHESS Depositary Interest, being a unit of beneficial ownership in a New Unibail Share registered in the name of CDN, to be issued under the WCL Scheme.

New Unibail Share means a Unibail Share issued under the WCL Scheme.

New Unibail Stapled Security means a Unibail Share and a Newco BI Share stapled together in accordance with the articles of association of Unibail and Newco BI and in compliance with applicable French and Dutch laws.

Newco BI means a public company with limited liability (*naamloze vennootschap*) incorporated or to be incorporated in the Netherlands with shares as described in Schedule 3 of the Implementation Agreement.

Newco BI CDI means a Newco BI CHESS Depositary Interest, being a unit of beneficial ownership in a Newco BI Share registered in the name of CDN, to be issued under the Scheme.

Newco BI Registry means the register of interest holders maintained by Newco BI or its agent.

Newco BI Share means a Class A Share in the capital of Newco BI as described in Schedule 3 of the Implementation Agreement.

Newco BI Transfer Form means one or more duly completed and executed private deed(s) of transfer of Newco BI Shares for the Scrip Consideration to the extent consisting of Newco BI Shares.

Registered Address means, in relation to a Member, the address shown in the Register.

Regulatory Authority includes:

(a) ASX;

- (b) ASIC;
- (c) Australian Taxation Office;
- a government or governmental, semi-governmental or judicial entity or authority;
- (e) a minister, department, office, commission, delegate, instrumentality, agency, board, authority or organisation of any government; and
- (f) any regulatory organisation established under statute.

Sale Agent means an entity appointed by Unibail to sell the Sale Securities pursuant to clause 29.16.

Sale Facility means the facility provided for in clause 29.16.

Sale Securities has the meaning given in clause 29.16(a).

Scheme means the arrangement under which US Newco acquires all of the Scheme Units, facilitated by amendments to this constitution.

Scheme Booklet means the scheme booklet published by Westfield and dated [insert] 2018.

Scheme Consideration means the consideration to be provided to Scheme Participants under the terms of the Scheme for the acquisition of their Scheme Units, as ascertained in accordance with clauses 29.10 to 29.22.

Scheme Meeting means the meeting of Members held on [*] for the purpose of, inter alia, voting on the Scheme Resolutions.

Scheme Participant means a person registered in the Register as a holder of one or more Scheme Units as at the Scheme Record Date.

Scheme Record Date means 5.00pm on the fourth Business Day following the Effective Date or such other date as the Manager and Unibail may agree in writing.

Scheme Resolutions means the resolutions of Members to approve the Scheme, being:

- (a) an ordinary resolution approving for all purposes, including item 7 of section 611 of the Corporations Act, the acquisition by US Newco of all the Scheme Units; and
- (b) a special resolution for the purpose of section 601GC(1) of the Corporations Act to approve amendments to this constitution to facilitate the implementation of the Scheme.

Scheme Unit means a Unit on issue as at the Scheme Record Date.

Scrip Consideration has the meaning given in clause 29.10(b).

Scrip Election has the meaning given in clause 29.13(b).

Selling Scheme Participant means a Scheme Participant in respect of whom Newco BI Shares are transferred to the Sale Agent under the Scheme.

Stapling Deed means the stapling deed between WCL and Manager (as responsible entity for the Trust and for WFDT) dated 30 June 2014.

TH Newco means a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated or to be incorporated in the Netherlands as referred to in the WFDT Constitutional Amendment.

Transaction means the acquisition of Westfield by Unibail contemplated by the Implementation Agreement.

Transaction Steps means each of the steps set out in Schedule 2 of the Implementation Agreement.

Unibail means Unibail-Rodamco SE (Registration Number 682 024 096).

Unibail Registry means the register of interest-holders maintained by Unibail or its agent.

Unibail Share means a fully paid ordinary share in the share capital (*capital social*) of Unibail, with a nominal value of €5.

US Newco means a corporation incorporated or to be incorporated in the United States.

WAT Transfer Form means, for each Scheme Participant, a duly completed and executed proper instrument of transfer of the Units held by that Scheme Participant, which may be a master transfer of all Scheme Units.

WCL means Westfield Corporation Limited (ABN 12 166 995 197).

WCL Scheme means the scheme of arrangement under Part 5.1 of the Corporations Act between WCL and each WCL shareholder as at the Scheme Record Date as part of the Transaction, subject to any alterations or conditions made or required by the Court pursuant to section 411(6) of the Corporations Act to the extent they are approved in writing by WCL and Unibail.

WCL Share means a fully paid ordinary share in the capital of WCL.

Westfield means WCL, the Manager as responsible entity of the Trust and as responsible entity of WFDT (together or any of them individually as the context requires).

Westfield Constitutions means this constitution, the WCL constitution and WFDT Constitution.

Westfield Securityholder means a person who is registered in the Register, WCL register of members and WFDT unit register of members as the holder of one or more Westfield Stapled Securities.

Westfield Stapled Security means a WCL Share, a Unit and a WFDT Unit stapled to each other in accordance with the provisions of the Stapling Deed and the Westfield Constitutions.

WFDT means WFD Trust (ARSN 168 765 875).

WFDT Constitution means the trust deed establishing WFDT dated 26 March 2014, as amended.

WFDT Constitutional Amendment means the supplemental deed poll under which the Manager will amend the WFDT Constitution for the purposes of implementing the WFDT Trust Scheme.

WFDT Trust Scheme means the arrangement under which TH Newco acquires all of the WFDT Units from each Westfield Securityholder as part of the

Transaction, facilitated by amendments to the WFDT Constitution, as set out in the WFDT Constitutional Amendments.

WFDT Unit means a fully paid ordinary unit in WFDT.

28.2 General interpretation

Headings and labels used for definitions are for convenience only and do not affect interpretation. Unless the contrary intention appears, in this clause 28 and 29 only:

- (a) the singular includes the plural and vice versa;
- (b) a reference to a document includes any agreement or other legally enforceable arrangement created by it (whether the document is in the form of an agreement, deed or otherwise);
- (c) a reference to a document also includes any variation, replacement or novation of it;
- (d) the meaning of general words is not limited by specific examples introduced by "including", "for example", "such as" or similar expressions;
- (e) a reference to "**person**" includes an individual, a body corporate, a partnership, a joint venture, an unincorporated association and an authority or any other entity or organisation:
- a reference to a particular person includes the person's executors, administrators, successors, substitutes (including persons taking by novation) and assigns;
- (g) a reference to a time of day is a reference to Sydney time;
- (h) where an act must take place on a day or a Business Day it must occur at a time of day that is both part of a day of the same date in Sydney, Australia and part of the same day of the same date in Paris, France;
- (i) a reference to:
 - (i) US\$ or US dollars is a reference to the currency of the United States of America:
 - (ii) A\$ or Australian dollars is a reference to the currency of Australia; and
 - (iii) euro or € is a reference to the currency of the European Union;
- a reference to any legislation includes regulations under it and any consolidations, amendments, re-enactments or replacements of any of them;
- (k) a reference to a group of persons is a reference to any 2 or more of them jointly and to each of them individually:
- (I) a reference to any thing (including an amount) is a reference to the whole and each part of it;
- (m) a period of time starting from a given day or the day of an act or event, is to be calculated exclusive of that day;

- (n) if a party must do something under this constitution on or by a given day and it is done after 5.00pm on that day, it is taken to be done on the next day; and
- (o) if the day on which a party must do something under this constitution is not a Business Day, the party must do it on the next Business Day.

28.3 Inconsistency

In this clause 28 and clause 29 defined terms in clause 28.1 and the interpretation principles in clause 28.2 apply to the exclusion of definitions or principles in the remainder of this constitution, to the extent of any inconsistency.

28.4 Cessation of operation

Clauses 28 and 29 (other than clause 29.17(f)) cease to have any force or operation if the Implementation Agreement or Deed Poll is terminated in accordance with its terms.

29 Scheme

29.1 Implementation of Scheme

- (a) Each Scheme Participant and the Manager must do all things and execute all deeds, instruments, transfers or other documents as the Manager considers are necessary or desirable to give full effect to the terms of the Scheme and the transactions contemplated by it.
- (b) Without limiting the Manager's other powers under this clause 29, the Manager has power to do all things that it considers necessary or desirable to give effect to the Scheme and the Implementation Agreement.
- (c) Subject to the Corporations Act, the Manager, the other Westfield entities, Unibail or any of their respective directors, officers, employees or associates, may do any act, matter or thing described in or contemplated by this clause 29 even if they have an interest (financial or otherwise) in the outcome.
- (d) Clause 28 and this clause 29:
 - (i) binds the Manager and all of the Members (including those who do not attend the Scheme Meeting, those who do not vote at the Scheme Meeting and those who vote against the Scheme Resolutions); and
 - (ii) to the extent of any inconsistency, overrides the other provisions of this constitution (but, for the avoidance of doubt, remains subject to the Corporations Act and the Listing Rules).

29.2 Transfer and registration of Scheme Units

On the Implementation Date at the time specified in the Transaction Steps:

- (a) subject to the provision of the Scheme Consideration in the manner contemplated by clauses 29.10 to 29.22 and Unibail having provided the Manager with written confirmation thereof, the Scheme Units, together with all rights and entitlements attaching to the Scheme Units, must be transferred to US Newco without the need for any further act by any Scheme Participant (other than acts performed by the Manager as attorney and agent for Scheme Participants under clause 29.31 of the Scheme) by:
 - (i) the Manager delivering to US Newco duly completed and executed WAT Transfer Forms executed on behalf of the Scheme Participants; and
 - (ii) US Newco duly executing the WAT Transfer Forms and delivering them to the Manager for registration; and
- (b) as soon as practicable after receipt of the duly executed WAT Transfer Forms the Manager must enter, or procure the entry of, the name of US Newco in the Register in respect of all Scheme Units transferred to US Newco in accordance with the terms of the Scheme.

29.3 Entitlement to receive Scheme Consideration

On the Implementation Date, in consideration for the transfer to US Newco of the Scheme Units, each Scheme Participant will be entitled to receive the Scheme Consideration in respect of each of their Scheme Units in accordance with clauses 29.10 to 29.22 of the Scheme.

29.4 Title and rights in Scheme Units

Subject to the provision of the Scheme Consideration for the Scheme Units as contemplated by clauses 29.10 to 29.22 of the Scheme, on and from the Implementation Date US Newco will be beneficially entitled to the Scheme Units transferred to it under the Scheme, pending registration by the Manager of US Newco in the Register as the holder of the Scheme Units.

29.5 Scheme Participants' agreements

Under the Scheme, each Scheme Participant irrevocably:

- (a) acknowledges that clause 28 and this clause 29 binds the Manager and all of the Members from time to time (including those who do not attend the Scheme Meeting, do not vote at the Scheme Meeting or vote against the Scheme Resolutions);
- (b) agrees to the transfer of their Scheme Units, together with all rights and entitlements attaching to those Scheme Units, in accordance with the terms of the Scheme;
- (c) agrees to the variation, cancellation or modification of the rights attached to their Scheme Units constituted by or resulting from the Scheme; and
- (d) consents to the Manager and Unibail doing all things and executing all deeds, instruments, transfers or other documents as may be necessary or desirable to give full effect to this clause 29 and the transactions contemplated by it.

29.6 Warranty by Scheme Participants

Each Scheme Participant warrants to US Newco and is deemed to have authorised the Manager to warrant to US Newco as agent and attorney for the Scheme Participant by virtue of this clause 29.6, that:

- (a) all their Scheme Units (including any rights and entitlements attaching to those securities) transferred to US Newco under the Scheme will, as at the date of the transfer, be fully paid and free from all Encumbrances; and
- (b) they have full power and capacity to sell and to transfer their Scheme Units (including any rights and entitlements attaching to those securities) to US Newco under the Scheme.

29.7 Transfer free of encumbrances

To the extent permitted by law, all Scheme Units (including any rights and entitlements attaching to those securities) which are transferred to US Newco under the Scheme will, at the date of the transfer of them to US Newco, vest in US Newco free from all Encumbrances and interests of third parties of any kind, whether legal or otherwise, and free from any restrictions on transfer of any kind not referred to in the Scheme.

29.8 Appointment of US Newco as sole proxy

Subject to the provision of the Scheme Consideration for the Scheme Units as contemplated by clauses 29.10 to 29.22 of the Scheme, on and from the Implementation Date until the Manager registers US Newco as the holder of all of the Scheme Units in the Register, each Scheme Participant:

- (a) irrevocably appoints the Manager as attorney and agent (and directs the Manager in such capacity) to appoint US Newco and each of its directors from time to time (jointly and each of them individually) as its sole proxy, and where applicable corporate representative, to attend securityholders' meetings, exercise the votes attaching to Units registered in its name and sign any securityholders resolution, and no Scheme Participant may itself attend or vote at any of those meetings or sign any resolutions, whether in person, by proxy or by corporate representative (other than pursuant to this clause 29.8(a)); and
- (b) must take all other actions in the capacity of the registered holder of Units as US Newco directs.

The Manager undertakes in favour of each Scheme Participant that it will appoint US Newco and each of its directors from time to time (jointly and each of them individually) as that Scheme Participant's proxy or, where applicable, corporate representative in accordance with clause 29.8(a) of the Scheme.

29.9 Transaction Steps

- (a) The Manager must implement, and must procure that WCL and WFDT implements, the Transaction Steps including to the extent required by counter-signing the Newco BI Transfer Form.
- (b) Unibail must implement or procure the implementation of the Transaction Steps.

29.10 Scheme Consideration

The Scheme Consideration to be provided to each Scheme Participant in respect of each Scheme Unit will be as follows:

- (a) subject to clause 29.11, a cash amount which, when aggregated with the 'Cash Consideration' (as defined in the WCL Scheme), equals US\$2.67 which will be paid in US\$, unless the Scheme Participant makes a valid Currency Election in accordance with clause 29.12 in which case the payment will be in A\$ in such amount as determined in accordance with clause 29.12 (the **Cash Consideration**); and
- (b) subject to clauses 29.11, 29.13, 29.14, 29.15 and 29.16, 0.01844 Newco BI Shares (the **Scrip Consideration**).

29.11 Fractions and splitting

- (a) Where the calculation of the aggregate number of Newco BI Shares or Newco BI CDIs to be transferred or issued to or in respect of a particular Scheme Participant would result in the provision of a fraction of a Newco BI Share or Newco BI CDI (after applying the Scheme Participant's entitlement (prior to rounding) to all of its Scheme Units), the aggregate number will be rounded down to the nearest whole number of Newco BI Shares or Newco BI CDIs.
- (b) Any cash amount payable to a Scheme Participant under the Scheme (after applying the Scheme Participant's entitlement (prior to rounding) to its entire holding of Scheme Units), when aggregated with any cash amount payable to the Scheme Participant under the WCL Scheme, must be rounded down to the nearest whole cent.
- (c) If Unibail is of the opinion, formed reasonably, that several Scheme Participants, each of which holds a holding of Scheme Units have, before the Scheme Record Date, been party to a shareholding splitting or division in an attempt to obtain an advantage by reference to the rounding provided for in the calculation of each Scheme Participant's entitlement to the Scheme Consideration, Unibail may direct the Manager to give notice (and if the Manager is given such a direction it must give notice) to those Scheme Participants:
 - (i) setting out the names and Registered Addresses of all of them;
 - (ii) stating that opinion; and
 - (iii) attributing to one of them specifically identified in the notice the Scheme Units held by all of them,

and, after the notice has been so given, the Scheme Participant specifically identified in the notice shall, for the purposes of the Scheme, be taken to hold all those Scheme Units and each of the other Scheme Participants whose names are set out in the notice shall, for the purposes of the Scheme, be taken to hold no Scheme Units.

29.12 Currency election

- (a) Subject to clause 29.12(c), each Scheme Participant whose Registered Address as at the Scheme Record Date is within Australia is deemed to have made an election to receive all of their Cash Consideration in the form of Australian dollars.
- (b) Subject to clause 29.12(c), each Scheme Participant whose Registered Address as at the Scheme Record Date is outside Australia is deemed to have made an election to receive their Cash Consideration in the form of US dollars.
- (c) A Scheme Participant whose Registered Address as at the Scheme Record Date is within Australia may make an election to receive all (or.

where clause 29.12(d) applies, some) of their Cash Consideration in the form of US dollars, and a Scheme Participant whose Registered Address as at the Scheme Record Date is outside Australia may make an election to receive all (or, where clause 29.12(d) applies, some) of their Cash Consideration in the form of Australian dollars. Such an election will be valid only if the following requirements are satisfied (or waived by Unibail in its sole discretion in respect of any particular Scheme Participant):

- (i) completing and signing an Election Form in accordance with the instructions on the Election Form; and
- (ii) the Election Form is received by the Unibail Registry before the Election Time at the address specified by Unibail on the Election Form.
- (d) If a Scheme Participant is noted on the Register as holding one or more parcels of Units as trustee or nominee for, or otherwise on account of, another person, Unibail may in its sole discretion allow the Scheme Participant to make elections under clause 29.12(c) in relation to each of those parcels of Units, and an election made in respect of any such parcel, or an omission to make an election in respect of any such parcel, will not be taken to extend to the other parcels. The manner in which such election is to be made, and the information to be provided to Unibail in relation to that, is to be as advised by Unibail to the Scheme Participant. Whether or not such an election is accepted is in the sole discretion of Unibail.
- (e) Subject to clause 29.12(d), a Currency Election made or deemed to be made by a Scheme Participant under this clause 29.12 will be deemed to apply in respect of the Scheme Participant's entire registered holding of Scheme Units on the Scheme Record Date, regardless of whether the Scheme Participant's holding of Scheme Units on the Scheme Record Date is greater or less than the Scheme Participant's holding at the time it made its Currency Election.
- (f) An election made under clause 29.12(c) may be revoked in accordance with the instructions on the Election Form.
- (g) Unibail will determine, in its sole discretion, all questions as to the correct completion of an Election Form, and time of receipt of an Election Form. Unibail is not required to communicate with any Scheme Participant prior to making this determination. The determination of Unibail will be final and binding on the Scheme Participant.
- (h) All Cash Consideration payable to a Scheme Participant in the form of A\$ will be based on the US\$/A\$ exchange rate on the Scheme Record Date, being such market-published rate selected by Unibail.
- (i) Notwithstanding anything else in this constitution:
 - (i) no Cash Consideration will be paid to a Scheme Participant in the form of A\$ if that Scheme Participant is to receive cash consideration under the WCL Scheme in the form of US\$ rather, the Scheme Participant will receive Cash Consideration in the form of US\$; and
 - (ii) no Cash Consideration will be paid to a Scheme Participant in the form of US\$ if that Scheme Participant is to receive cash consideration under the WCL Scheme in the form of A\$ rather,

the Scheme Participant will receive Cash Consideration in the form of A\$.

29.13 Scrip Consideration

- (a) In accordance with the Transaction Steps, each Newco BI Share transferred under the Scheme will on the Implementation Date be stapled to a New Unibail Share issued under the WCL Share Scheme and WFDT Trust Scheme to form a New Unibail Stapled Security.
- (b) If the Registered Address of a Scheme Participant as at the Scheme Record Date is within Australia and the Scheme Participant is not an Ineligible Foreign Holder:
 - (i) the Newco BI Shares to which the Scheme Participant is entitled under the Scheme will be in the form of Newco BI CDIs; and
 - (ii) the Scheme Participant will receive Newco BI CDIs.

However, if either:

- (iii) the condition precedent in clause 3.1(n)(ii) of the Implementation Agreement is waived in accordance with the Implementation Agreement; or
- (iv) Unibail in its sole discretion determines to do so following a valid election made by the Scheme Participant in accordance with clause 29.13(d),

that Scheme Participant will receive their Scrip Consideration in the form of Newco BI Shares.

- (c) If the Registered Address of a Scheme Participant as at the Scheme Record Date is outside Australia and the Scheme Participant is not an Ineligible Foreign Holder the Scheme Participant will receive their Scrip Consideration in the form of Newco BI Shares. However, if Unibail in its sole discretion determines to do so following a valid election made by that Scheme Participant in accordance with clause 29.13(d), that Scheme Participant will receive their Scrip Consideration in the form of Newco BI CDIs.
- (d) An election by a Scheme Participant to receive their Scrip Consideration in a form other than as prescribed by clause 29.13(b) or 29.13(c) (as applicable) (being a **Scrip Election**) will be valid only if the following requirements are satisfied to Unibail's sole satisfaction (or waived by Unibail in its sole discretion) in respect of any particular Scheme Participant:
 - (i) completing and signing an election form (that will be provided by the Manager to the Scheme Participant upon request) in accordance with the instructions on the election form; and
 - (ii) the election form is received by the Unibail Registry before the Election Time at the address specified by Unibail on the Election Form.
- (e) If a Scheme Participant is noted on the Register as holding one or more parcels of Units as trustee or nominee for, or otherwise on account of, another person, Unibail may in its sole discretion allow the Scheme Participant to make separate US dollar and Australian dollar elections under clause 29.13(d) in relation to each of those parcels of Units, and

an election made in respect of any such parcel, or an omission to make an election in respect of any such parcel, will not be taken to extend to the other parcels. The manner in which such election is to be made, and the information to be provided to Unibail in relation to that, is to be as advised by Unibail to the Scheme Participant. Whether or not such an election is accepted is in the sole discretion of Unibail.

- (f) Subject to clause 29.13(e), an election made or deemed to be made under this clause 29.13 will be deemed to apply in respect of a Scheme Participant's entire registered holding of Scheme Units at the Scheme Record Date, regardless of whether the Scheme Participant's holding of Scheme Units at the Scheme Record Date is greater or less than the Scheme Participant's holding at the time it made its election under clause 29.13(d).
- (g) An election made under clause 29.13(d) may be revoked in accordance with the instructions on the Election Form.
- (h) Unibail will determine, in its sole discretion, all questions as to the correct completion of an Election Form, and time of receipt of an Election Form. Unibail is not required to communicate with any Scheme Participant prior to making this determination. The determination of Unibail will be final and binding on the Scheme Participant.
- (i) Notwithstanding anything else in this constitution:
 - (i) no Newco BI Shares will be transferred to or in respect of a Scheme Participant under the Scheme if that Scheme Participant is to be issued New Unibail CDIs under the WCL Scheme rather, such Scheme Participant's Scrip Consideration will be in the form of Newco BI CDIs; and
 - (ii) no Newco BI CDIs will be issued to a Scheme Participant under the Scheme if that Scheme Participant is to be issued New Unibail Shares under the WCL Scheme rather, such Scheme Participant's Scrip Consideration will be in the form of Newco BI Shares.

29.14 Minimum Holders

- (a) Subject to clause 29.14(b), Unibail will be under no obligation to procure the issue or transfer, and must not procure the issue or transfer of, any Newco BI CDIs or Newco BI Shares under the Scheme to or in respect of any Minimum Holder and, instead, clause 29.16 will apply.
- (b) This clause 29.14 only applies if Westfield obtains any necessary relief from the operation of section 601FC(1)(d) of the Corporations Act to enable it to apply.

29.15 Ineligible Foreign Holders

Unibail will be under no obligation to procure the issue or transfer, and must not procure the issue or transfer of, any Newco BI CDIs or Newco BI Shares under the Scheme to or in respect of any Ineligible Foreign Holder and, instead, clause 29.16 will apply.

29.16 Sale Facility

In respect of each Scheme Participant which is a Minimum Holder (if clause 29.14 applies) or an Ineligible Foreign Holder:

- (a) Unibail must not issue or provide to or in respect of the Scheme Participant any Newco BI CDIs or Newco BI Shares that would otherwise be issued or provided to the Scheme Participant under the Scheme and, instead, Unibail must on the Implementation Date transfer to the Sale Agent the Newco BI Shares that would otherwise have been transferred to or in respect of the Scheme Participant (and all such Newco BI Shares together with the New Unibail Shares to which they will be stapled shall be referred to as the **Sale Securities**).
- (b) Unibail must procure that as soon as practicable after the Implementation Date and, in any event, not more than 20 Business Days after the Implementation Date:
 - (i) the Sale Agent, in consultation with Unibail, sells or procures the sale (including on an aggregated or partially aggregated basis), on Euronext Paris and/or Euronext Amsterdam, of all the Sale Securities in such manner, at such price or prices and on such other terms as the Sale Agent determines in good faith; and
 - (ii) remits to the Manager in US dollars the proceeds of the sales, after deduction of any applicable brokerage, taxes, duty, currency conversion costs and other costs and charges (converted into US dollars in such manner, at such price or prices and on such other terms as the Sale Agent determines in good faith) (Net Sale Proceeds).
- (c) Within 10 Business Days after receiving the Net Sale Proceeds in respect of the sale of all of the Sale Securities, the Manager must pay, or procure the payment, to each Selling Scheme Participant (in accordance with this clause 29.16) an amount calculated as follows:

$A \div B \times C$

Where:

A = the Net Sale Proceeds;

B = the total number of Newco BI Shares transferred to the Sale Agent under clause 29.16(a).

C = the number of Newco BI Shares transferred to the Sale Agent under clause 29.16(a) in respect of that Selling Scheme Participant (which, for the avoidance of doubt, must not include a fraction of a Newco BI Share).

Such payment represents each Selling Scheme Participant's entitlement to part of the Net Sale Proceeds under both the Scheme and the WCL Scheme.

- (d) None of Westfield, Unibail or the Sale Agent gives any assurance as to the price that will be achieved for the sale of Sale Securities described in clause 29.16(b). The sale of Sale Securities under this clause 29.16 will be at the risk of the Selling Scheme Participant.
- (e) The Manager must make payments to Selling Scheme Participants under clause 29.16(c) by either (in the absolute discretion of the Manager):
 - (i) where a Selling Scheme Participant has, before the Scheme Record Date, made a valid election in accordance with the requirements of the Registrar to receive distribution payments from Westfield by electronic funds transfer to a bank account nominated by the Selling Scheme Participant denominated in US dollars, paying, or procuring the payment of, the relevant

- amount in US dollars by electronic means in accordance with that election: or
- (ii) otherwise, whether or not the Selling Scheme Participant has made an election referred to in clause 29.16(e)(i), dispatching, or procuring the dispatch of, a cheque for the relevant amount in US dollars to the Selling Scheme Participant by prepaid post to their Registered Address (as at the Scheme Record Date), such cheque being drawn in the name of the Selling Scheme Participant (or in the case of joint holders, in accordance with the procedures set out in clause 29.19).
- (f) If the Manager receives professional advice that any withholding or other tax is required by law to be withheld from a payment to a Selling Scheme Participant, the Manager is entitled to withhold the relevant amount before making the payment to the Selling Scheme Participant (and payment of the reduced amount shall be taken to be full payment of the relevant amount for the purposes of the Scheme, including clause 29.16(c)). The Manager must pay any amount so withheld to the relevant taxation authorities within the time permitted by law, and, if requested in writing by the relevant Selling Scheme Participant, provide a receipt or other appropriate evidence of such payment (or procure the provision of such receipt or other evidence) to the relevant Selling Scheme Participant.
- (g) Each Selling Scheme Participant appoints the Manager as its agent to receive on its behalf any financial services guide or other notices (including any updates of those documents) that the Sale Agent is required to provide to Selling Scheme Participant under the Corporations Act.

29.17 Provision of Cash Consideration

- (a) Unibail must, and the Manager must use its best endeavours to procure that Unibail does by no later than the Business Day before the Implementation Date, deposit in cleared funds an amount equal to:
 - (i) the aggregate amount in US dollars of the Cash Consideration which is payable to Scheme Participants in US dollars by operation of clause 29.12, into a US dollar denominated trust account operated by the Manager as trustee for the Scheme Participants (provided that any interest on the amounts deposited (less bank fees and other charges) will be credited to Unibail's account); and
 - (ii) the aggregate amount in Australian dollars of the Cash Consideration which is payable to Scheme Participants in Australian dollars by operation of clause 29.12 into an Australian dollar denominated trust account operated by the Manager as trustee for the Scheme Participants (provided that any interest on the amounts deposited (less bank fees and other charges) will be credited to Unibail's account).
- (b) Upon Unibail having complied with its obligations in clause 29.17(a), Unibail will be taken to have satisfied its obligations to provide or procure the provision of Cash Consideration to each Scheme Participant
- (c) On the Implementation Date, subject to funds having been deposited in accordance with clause 29.17(a), the Manager must pay or procure the payment, from the trust accounts referred to in clause 29.17(a), to each Scheme Participant such amount of cash from the cash component of

- the Scheme Consideration as that Scheme Participant is entitled under this clauses 29.10 to 29.22 (if any).
- (d) The obligations of the Manager under clause 29.17(c) will be satisfied by the Manager (in its absolute discretion):
 - (i) where a Scheme Participant has, before the Scheme Record Date, made a valid election in accordance with the requirements of the Registrar to receive distribution payments from the Manager by electronic funds transfer to a bank account nominated by the Scheme Participant which is denominated in the currency specified in a Currency Election, paying, or procuring the payment of, the relevant amount in the currency of that Scheme Participant's Currency Election by electronic means in accordance with that election; or
 - (ii) otherwise, whether or not the Scheme Participant has made an election referred to in clause 29.17(d)(i), dispatching, or procuring the dispatch of, a cheque for the relevant amount in the currency of that Scheme Participant's Currency Election to the Scheme Participant by prepaid post to their Registered Address (as at the Scheme Record Date), such cheque being drawn in the name of the Scheme Participant (or in the case of joint holders, in accordance with the procedures set out in clause 29.19).
- (e) To the extent that, following satisfaction of the Manager's obligations under clause 29.17(b) and 29.20, there is a surplus in the amount held by the Manager as trustee for the Scheme Participants in the trust account referred to in clause 29.17(a), that surplus shall be paid by the Manager to Unibail.
- (f) If, following satisfaction of Unibail's obligations under clause 29.17(a) but prior to the occurrence of all of the events described in clause 29.2, clause 28.4 applies:
 - the Manager must immediately repay (or cause to be repaid) to or at the direction of Unibail the funds that were deposited in the bank accounts described in clause 29.17(a) plus any interest on the amounts deposited (less bank fees and other charges); and
 - the obligation to transfer Scheme Units, together with all rights and entitlements attaching to the Scheme Units as at the Implementation Date, to US Newco under clause 29.2 will immediately cease;
 - (iii) US Newco must return the WAT Transfer Forms, if provided pursuant to clause 29.2(a)(i), and the Manager is no longer obliged to enter, or procure the entry of, the name of US Newco in the Register in accordance with clause 29.2(b).

29.18 Provision of Scrip Consideration

- (a) Subject to clauses 29.9, 29.11, 29.13, 29.14 and 29.16, the obligation of Unibail to provide the Scrip Consideration to each Scheme Participant will be satisfied:
 - (i) in the case of Scrip Consideration that is required to be provided to or in respect of each Scheme Participant in the form of Newco BI Shares, by US Newco delivering to the Manager a copy of the duly executed Newco BI Transfer Form;

- (ii) in the case of Scrip Consideration that is required to be provided to each Scheme Participant in the form of Newco BI CDIs, by:
 - (A) US Newco transferring to CDN the Newco BI Shares that will underlie those Newco BI CDIs;
 - (B) CDN issuing the Newco BI CDIs to the Scheme Participant; and
 - (C) the name of each such Scheme Participant being entered in the records maintained by CDN as the holder of those Newco BI CDIs; and
- (iii) in respect of each Scheme Participant which is a Minimum Holder (if clause 29.14 applies) or an Ineligible Foreign Holder, by Unibail complying with clause 29.16(a).
- (b) Unibail must procure that, on or before the date that is two Business Days after the Implementation Date, a statement is sent to the Registered Address of each Scheme Participant in respect of whom Newco BI Shares or Newco BI CDIs are transferred or issued in accordance with clause 29.18(a) stating the number of New Unibail Stapled Securities or New Unibail CDIs transferred or issued to that Scheme Participant pursuant to the Scheme and the WCL Scheme.

29.19 Joint holders

In the case of Scheme Units held in joint names:

- (a) subject to clause 29.17(d), any amount comprising the cash component of the Scheme Consideration payable in respect of those Scheme Units is payable to the joint holders and any cheque required to be sent under the Scheme will be made payable to the joint holders and sent to either, at the sole discretion of the Manager, the holder whose name appears first in the Register as at the Scheme Record Date or to the joint holders;
- (b) any Newco BI Shares or Newco BI CDIs to be transferred or issued under the Scheme in respect of such Scheme Units must be transferred or issued to such joint holders; and
- (c) any document required to be sent under the Scheme, will be forwarded to either, at the sole discretion of the Manager, the holder whose name appears first in the Register as at the Scheme Record Date or to the joint holders.

29.20 Unclaimed monies

- (a) The Manager may cancel a cheque issued under this clause 29 if the cheque:
 - (i) is returned to the Manager; or
 - (ii) has not been presented for payment within six months after the date on which the cheque was sent.
- (b) During the period of 12 months commencing on the Implementation Date, on request in writing from a Scheme Participant to the Manager (or the Registrar) (which request may not be made earlier than the date which is 20 Business Days after the Implementation Date), the Manager must reissue a cheque that was previously cancelled under this clause 29.20.

(c) The *Unclaimed Money Act 1995* (NSW) will apply in relation to any amount payable to a Scheme Participant under the Scheme which becomes 'unclaimed money' (as defined in section 7 of the *Unclaimed Money Act 1995* (NSW)).

29.21 Orders of a court or Regulatory Authority

- (a) The Manager may deduct and withhold from any consideration which would otherwise be payable by the Manager to a Scheme Participant in accordance with this clause 29, any amount which Unibail and the Manager determine is required to be deducted and withheld from that consideration under any applicable law, including any order, direction or notice made or given by a court of competent jurisdiction or by another Regulatory Authority.
- (b) To the extent that amounts are so deducted or withheld, such deducted or withheld amounts will be treated for all purposes under the Scheme as having been paid to the person in respect of which such deduction and withholding was made, provided that such deducted or withheld amounts are actually remitted to the appropriate taxing agency.
- (c) If written notice is given to the Manager (or the Registrar) of an order, direction or notice made or given by a court of competent jurisdiction or by another Regulatory Authority that:
 - (i) requires consideration which would otherwise be payable or provided to a Scheme Participant by the Manager in accordance with this clause 29 must instead be paid or provided to a Regulatory Authority or other third party (either through payment of a sum or the issuance of a security), then the Manager shall be entitled to procure that payment or provision of that consideration is made in accordance with that order, direction or notice (and payment or provision of that consideration in accordance with that order, direction or notice will be treated for all purposes under the Scheme as having been paid or provided to that Scheme Participant); or
 - (ii) prevents the Manager from providing consideration to any particular Scheme Participant in accordance with this clause 29, or the payment or provision of such consideration is otherwise prohibited by applicable law, the Manager shall be entitled to (as applicable):
 - (A) retain an amount equal to the cash component of the Scheme Consideration to which that Scheme Participant would otherwise be entitled to under clause 29.3; and
 - (B) direct Unibail not to issue or provide (or procure the issue or provision of), or to issue or provide to a trustee or nominee, such number of Newco BI Shares or Newco BI CDIs as that Scheme Participant would otherwise be entitled to under clause 29.3,

until such time as payment or provision of the consideration in accordance with this clause 29 is permitted by that order or direction or otherwise by law.

29.22 Status of Newco BI Shares and Newco BI CDIs

Unibail must:

- (a) transfer all Newco BI Shares under the Scheme to ensure that each Newco BI Share transferred under the Scheme ranks equally in all respects with all other Newco BI Shares on issue at the time:
- (b) ensure that each such Newco BI Share is duly and validly issued in accordance with all applicable laws and the constituent and organisational documents of Newco BI, fully paid and free from any Encumbrance; and
- (c) use all reasonable endeavours to ensure that:
 - (i) the New Unibail Stapled Security of which each Newco BI Share transferred under the Scheme is a part is, from the Business Day following the date the Scheme becomes Effective (or such later date as the relevant exchange requires), quoted for trading on Euronext Paris and Euronext Amsterdam; and
 - (ii) each New Unibail CDI issued under the Scheme is, from the Business Day following the date the Scheme becomes Effective (or such later date as ASX requires), quoted for trading on the ASX initially on a deferred settlement basis and thereafter on an ordinary settlement basis.

29.23 Determination of Scheme Units

To establish the identity of the Scheme Participants, dealings in Scheme Units will only be recognised by the Manager if:

- (a) in the case of dealings of the type to be effected using CHESS, the transferee is registered in the Register as the holder of the relevant Scheme Units on or before the Scheme Record Date; and
- (b) in all other cases, registrable transmission applications or transfers in registrable form in respect of those dealings are received on or before the Scheme Record Date at the place where the Register is kept.

29.24 Register

The Manager must register any registrable transmission applications or transfers of Units received in accordance with clause 29.23(b) of the Scheme on or before the Scheme Record Date.

29.25 No disposals after Effective Date

- (a) If the Scheme becomes Effective, a holder of Scheme Units (and any person claiming through that holder) must not dispose of or purport or agree to dispose of any Scheme Units or any interest in them after the Effective Date in any way except as set out in the Scheme and any such disposal will be void and of no legal effect whatsoever.
- (b) The Manager will not accept for registration or recognise for any purpose any transmission, application or transfer in respect of Scheme Units received after the Scheme Record Date (except a transfer to US Newco pursuant to the Scheme or any subsequent transfer by US Newco or its successors in title).

29.26 Maintenance of Register

For the purpose of determining entitlements to the Scheme Consideration, the Manager will maintain the Register in accordance with the provisions of this clause 29.26 until the Scheme Consideration has been provided to the Scheme Participants and US Newco has been entered in the Register as the holder of all

the Scheme Units. The Register in this form will solely determine entitlements to the Scheme Consideration.

29.27 Effect of certificates and holding statements

Subject to provision of the Scheme Consideration and registration of the transfer to US Newco contemplated in clauses 29.2 and 29.26 of the Scheme, any statements of holding in respect of Scheme Units will cease to have effect after the Scheme Record Date as documents of title in respect of those shares (other than statements of holding in favour of US Newco and its successors in title). After the Scheme Record Date, each entry current on the Register as at the Scheme Record Date (other than entries in respect of Unibail or its successors in title) will cease to have effect except as evidence of entitlement to the Scheme Consideration.

29.28 Details of Scheme Participants

Within 3 Business Days after the Scheme Record Date, the Manager will ensure that details of the names, Registered Addresses and holdings of Scheme Units for each Scheme Participant, as shown in the Register at the Scheme Record Date are available to US Newco in such form as US Newco or Unibail reasonably requires.

29.29 Quotation of Units

Suspension of trading on ASX in Units will occur from the close of trading on ASX on the Effective Date.

29.30 Termination of quotation of Units

After the Scheme has been implemented, the Manager will apply:

- (a) for termination of the official quotation of Units on ASX; and
- (b) to have itself removed from the official list of the ASX.

29.31 Power of attorney

Each Scheme Participant, without the need for any further act by any Scheme Participant, irrevocably appoints the Manager and each of its directors and secretaries (jointly and each of them individually) as its attorney and agent for the purpose of:

- (a) executing any document necessary or expedient to give effect to the Scheme and the Transaction Steps including:
 - (i) executing any WAT Transfer Forms;
 - (ii) executing any Newco BI Transfer Form; and
 - (iii) applying for shares in (and agreeing to become a member of) any of the entities in which Scheme Participants are to become members in accordance with the Transaction Steps; and
- (b) enforcing the Deed Poll against Unibail,

and the Manager accepts such appointment.

29.32 Notices

(a) If a notice, transfer, transmission application, direction or other communication referred to in the Scheme is sent by post to the Manager, it will not be taken to be received in the ordinary course of post or on a

date and time other than the date and time (if any) on which it is actually received at the Manager's registered office or at the office of the Registrar.

(b) The accidental omission to give notice of the Scheme Meeting or the non-receipt of such a notice by any Member will not, unless so ordered by the Court, invalidate the Scheme Meeting or the proceedings of the Scheme Meeting.

29.33 Further action by the Manager

The Manager will execute all documents and do all things (on its own behalf and on behalf of each Scheme Participant) necessary or expedient to implement, and perform its and each Scheme Participant's obligations under, the Scheme and the Transaction Steps.

29.34 No liability when acting in good faith

Neither the Manager nor Unibail, nor any of their respective officers, will be liable for anything done or omitted to be done in the performance of the Scheme in good faith.

29.35 Stamp duty

Unibail will pay all stamp duty (including any fines, penalties and interest) payable in respect of the Scheme.

29.36 Limitation of liability

Without limiting clause 14.3 but subject to the Corporations Act, the Manager will not have any liability of any nature whatsoever to Members, beyond the extent to which the Manager is actually indemnified out of the Assets, arising, directly or indirectly, from the Manager doing or refraining from doing any act (including the execution of a document), matter or thing pursuant to or in connection with the implementation of the Scheme.



WFD Trust Supplemental Deed

Dated

Westfield America Management Limited (ABN 66 072 780 619) ("Manager")

King & Wood Mallesons

Level 61
Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia
T +61 2 9296 2000
F +61 2 9296 3999
DX 113 Sydney
www.kwm.com

WFD Trust Supplemental Deed

Details

Manager	Name		Westfield America Management Limited
	ABN		66 072 780 619
	Capac	ty	Responsible entity of the WFDT
	Addres	SS	Level 29, 85 Castlereagh Street, Sydney NSW 2000, Australia
Governing law	New South Wales		
Recitals	Α	Manager is the responsible entity of the trust known WFD Trust (ARSN 168 765 875) (WFDT). WFDT is registered as a managed investment scheme under Chapter 5C of the Corporations Act.	
	В		
	C WFDT was		s constituted under the Constitution.
	D	WFDT is admitted to the official list of the ASX along with WAT and WCL. One share in WCL and one unit in each of WAT and WFDT are stapled together and are officially quoted on the stock market conducted by ASX as Westfield Stapled Securities.	
	E	Westfield and Unibail have agreed, by executing an implementation agreement dated 12 December 2017 (the Implementation Agreement), to propose and implement the Scheme, the WAT Trust Scheme and the WCL Scheme to Westfield Securityholders, and each of WAT, WCL, WFDT and Unibail has agreed to take certain steps to give effect to the Scheme (including the Transaction Steps).	
	F	The Const	itution must be amended to facilitate the Scheme.
	G	Section 601GC(1) of the Corporations Act provides that the constitution of a registered scheme may be modified, or repealed and replaced with a new constitution:	
		(a) by	y special resolution of the members of the scheme;
		th	y the responsible entity if it reasonably considers be change will not adversely affect members' ghts.
	1.1		and the Constitution the Constitution may be

Under clause 24 of the Constitution, the Constitution may be modified by Resolution and Manager may give effect to the

modification by executing a supplemental deed.

Н

I Manager wishes to modify the Constitution, as set out in this document, to give effect to the Resolution to modify the Constitution that was passed by Members of WAT at the meeting held on [insert date].

WFD Trust Supplemental Deed

General terms

1 Interpretation

1.1 Definitions

In this document, these words and phrases have the following meanings and any other words and phrases have the meaning given to them in the Constitution, unless the contrary intention appears:

Constitution means the document dated 26 March 2014 under which the WFDT Trust was constituted as amended from time to time.

Effective Time means the date on and time at which a copy of this Supplemental Deed is lodged with the Australian securities and Investments Commission under section 601GC(2) of the Corporations Act.

Scheme means the arrangement under which TH Newco acquires all of the Scheme Units from each Westfield Securityholder as part of the Transaction.

Scheme Unit means a Unit held by the Scheme Participants as at the Scheme Record Date.

Scheme Participant means a person registered in the Register as a holder of one or more Scheme Units as at the Scheme Record Date.

Scheme Record Date means 5.00pm on the fourth Business Day following the Effective Time or such other date as the Manager and Unibail may agree in writing.

Stapling Deed means the stapling deed between WCL and Manager (as responsible entity for WAT and for WFDT) dated 30 June 2014.

TH Newco means a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated or to be incorporated in the Netherlands.

Transaction means the acquisition of Westfield by Unibail contemplated by the Implementation Agreement, including the implementation of all of the Transaction Steps in accordance with the Scheme, the WCL Scheme, the WAT Trust Scheme and the Implementation Agreement.

Transaction Steps means each of the steps set out in Schedule 2 of the Implementation Agreement.

Unibail means Unibail-Rodamco SE (Registration Number 682 024 096).

US Newco means a corporation incorporated or to be incorporated in the United States as referred to in the WAT Trust Scheme.

WAT means Westfield America Trust (ARSN 092 058 499).

WAT Trust Scheme means the arrangement under which US Newco acquires all of the WAT Units from each Westfield Securityholder as part of the Transaction, facilitated by amendments to the constitution of WAT, as set out in the supplemental deed poll under which the Manager will amend the constitution of WFDT.

WAT Unit means a fully paid ordinary unit in WAT.

WCL means Westfield Corporation Limited (ABN 12 166 995 197).

WCL Scheme means the scheme of arrangement under Part 5.1 of the Corporations Act between WCL and each Westfield Securityholder as part of the Transaction, subject to any alterations or conditions made or required by the Court pursuant to section 411(6) of the Corporations Act to the extent they are approved in writing by WCL and Unibail.

WCL Share means a fully paid ordinary share in the capital of WCL.

Westfield Securityholder means a person who is registered in the Register, WCL share register of members and WAT unit register of members as the holder of one or more Westfield Stapled Securities.

Westfield Stapled Security means a WCL Share, a WAT Unit and a Unit stapled to each other in accordance with the provisions of the Stapling Deed, the Constitution and the constitutions of WCL and WAT.

WFDT means WFD Trust (ARSN 168 765 875).

1.2 Interpretation

Clause 31.2 of the Constitution applies to this document as if set out in this document, except that references to "this constitution" are references to this document.

1.3 Document supplemental to Constitution

This document is supplemental to the Constitution.

2 Benefit of this document

This document is made by the Manager with the intent that the benefit of this document shall ensure to the benefit of Members, jointly and severally.

3 Conditions

This document is conditional upon, and will have no force of effect until, the satisfaction of each of the conditions precedent stipulated in clause 3.1 of the WCL Scheme (as defined in the Implementation Agreement) other than the coming into effect of the amendments made by this document.

4 Modifications to the Constitution

The Constitution is modified on and from the Effective Time in the manner set out in Schedule 1.

5 No redeclaration

The Manager is not, by this document:

- (a) redeclaring or resettling WFDT;
- (b) declaring any trust; or
- (c) causing the transfer, vesting or accruing of property in any person.

6 Governing law

This document is governed by the laws in force in the place specified in the Details. Each person affected by it must submit to the non-exclusive jurisdiction of the courts of that place and the courts of appeal from them.

EXECUTED as a deed poll	
DATED:	
EXECUTED by WESTFIELD AMERICA MANAGEMENT LIMITED as responsible entity of WFD TRUST in accordance with section 127(1) of the Corporations Act 2001 (Cth) by authority of its directors:	
Signature of director	Signature of director/company secretary* *delete whichever is not applicable
Name of director (block letters)	Name of director/company secretary* (block letters) *delete whichever is not applicable

WFD Trust Supplemental Deed

Schedule 1

32 Interpretation

32.1 Definitions

In clause 32 and 33 the following definitions are inserted in alphabetical order:

Unless the contrary intention appears, these meanings apply:

Business Day means a business day as defined in the Listing Rules and on which trading will take place on Euronext Paris and Euronext Amsterdam and the New York Stock Exchange.

CHESS means the clearing house electronic sub register system of share transfers operated by ASX Settlement and Transfer Corporation Pty Ltd.

Court means the Federal Court of Australia, Sydney Registry, or such other court as is agreed in writing by Unibail and the Manager.

Deed Poll means the deed poll dated [insert date] executed by Unibail under which Unibail covenants in favour of each Scheme Participant to perform its obligations under the Scheme.

Effective Date means the date and time at which the supplemental deed making amendments to this constitution to facilitate the Scheme, take effect pursuant to section 601GC(2) of the Corporations Act.

Encumbrance means any mortgage, charge, lien, pledge, trust, power or title retention, flawed deposit arrangement, "security interest" as defined in sections 12(1) or (2) of the *Personal Property Securities Act 2009* (Cth) (Australia), or interest of a third party of any kind, whether legal or otherwise, or any agreement to create any of them or allowing them to exist.

Euronext Amsterdam means the regulated market operated by Euronext Amsterdam NV.

Euronext Paris means the regulated market operated by Euronext Paris S.A.

Implementation Agreement means the implementation agreement dated 12 December 2017 between Westfield and Unibail under which, amongst other things, the Manager has agreed to propose the Scheme to Members, and each of Westfield and Unibail has agreed to take certain steps to give effect to the Scheme (including the Transaction Steps).

Implementation Date means the fifth Business Day following the Scheme Record Date or such other date after the Scheme Record Date as is agreed in writing by Unibail and the Manager.

Ineligible Foreign Holder means a Scheme Participant whose address shown in the Register on the Scheme Record Date is a place outside Australia and its external territories and New Zealand, unless Unibail determines that it is lawful and not unduly onerous or impracticable to issue that Scheme Participant with TH Newco Shares under the Scheme.

Listing Rules means the Listing Rules of the ASX.

Registered Address means, in relation to a Member, the address shown in the Register.

Scheme means the arrangement under which TH Newco acquires all of the Scheme Units facilitated by amendments to this constitution.

Scheme Consideration means, in respect of each Scheme Unit, one TH Newco Share.

Scheme Meeting means the meeting of Members held on [●] for the purpose of, inter alia, voting on the Scheme Resolutions.

Scheme Participant means a person registered in the Register as a holder of one or more Scheme Units as at the Scheme Record Date.

Scheme Record Date means 5.00pm on the fourth Business Day following the Effective Date or such other date as the Manager and Unibail may agree in writing.

Scheme Resolutions means the resolutions of Members to approve the Scheme, being:

- (a) an ordinary resolution approving for all purposes, including item 7 of section 611 of the Corporations Act, the acquisition by TH Newco of all the Scheme Units; and
- (b) a special resolution for the purpose of section 601GC(1) of the Corporations Act to approve amendments to this constitution to facilitate the implementation of the Scheme.

Scheme Unit means a Unit on issue as at the Scheme Record Date.

TH Newco means a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated or to be incorporated in the Netherlands.

TH Newco Deed of Issue means the notarial deed of issue by which the TH Newco Scheme Shares are issued.

TH Newco Scheme Shares means the TH Newco Shares which are issued on the Implementation Date under the Scheme.

TH Newco Share means an ordinary share in the share capital of TH Newco.

Third Party Nominee means an entity, appointed by Unibail, that will be issued the TH Newco Scheme Shares.

Transaction means the acquisition of Westfield by Unibail contemplated by the Implementation Agreement.

Transaction Steps means each of the steps set out in Schedule 2 of the Implementation Agreement.

Unibail means Unibail-Rodamco SE (Registration Number 682 024 096).

WAT means Westfield America Trust (ARSN 092 058 499).

WCL means Westfield Corporation Limited (ABN 12 166 995 197).

WCL Scheme means the scheme of arrangement under Part 5.1 of the Corporations Act between WCL and each WCL shareholder as at the Scheme

Record Date as part of the Transaction, subject to any alterations or conditions made or required by the Court pursuant to section 411(6) of the Corporations Act to the extent they are approved in writing by WCL and Unibail.

Westfield means WCL, the Manager as responsible entity of the Trust and as responsible entity of WAT (together or any of them individually as the context requires).

WFDT Transfer Form means, for each Scheme Participant, a duly completed and executed proper instrument of transfer of the Units held by that Scheme Participant, which may be a master transfer of all Scheme Units.

32.2 General interpretation

Headings and labels used for definitions are for convenience only and do not affect interpretation. Unless the contrary intention appears, in this clause 32 and 33 only:

- (a) the singular includes the plural and vice versa;
- (b) a reference to a document includes any agreement or other legally enforceable arrangement created by it (whether the document is in the form of an agreement, deed or otherwise);
- (c) a reference to a document also includes any variation, replacement or novation of it;
- (d) the meaning of general words is not limited by specific examples introduced by "including", "for example", "such as" or similar expressions;
- (e) a reference to "person" includes an individual, a body corporate, a partnership, a joint venture, an unincorporated association and an authority or any other entity or organisation;
- (f) a reference to a particular person includes the person's executors, administrators, successors, substitutes (including persons taking by novation) and assigns;
- (g) a reference to a time of day is a reference to Sydney time;
- (h) where an act must take place on a day or a Business Day it must occur at a time of day that is both part of a day of the same date in Sydney, Australia and part of the same day of the same date in Paris, France;
- (i) a reference to:
 - (i) US\$ or US dollars is a reference to the currency of the United States of America;
 - (ii) A\$ or Australian dollars is a reference to the currency of Australia; and
 - (iii) euro or € is a reference to the currency of the European Union;
- a reference to any legislation includes regulations under it and any consolidations, amendments, re-enactments or replacements of any of them:
- (k) a reference to a group of persons is a reference to any 2 or more of them jointly and to each of them individually;

- (I) a reference to any thing (including an amount) is a reference to the whole and each part of it;
- (m) a period of time starting from a given day or the day of an act or event, is to be calculated exclusive of that day;
- (n) if a party must do something under this constitution on or by a given day and it is done after 5.00pm on that day, it is taken to be done on the next day; and
- (o) if the day on which a party must do something under this constitution is not a Business Day, the party must do it on the next Business Day.

32.3 Inconsistency

In this clause 32 and clause 33 defined terms in clause 32.1 and the interpretation principles in clause 32.2 apply to the exclusion of definitions or principles in the remainder of this constitution, to the extent of any inconsistency.

32.4 Cessation of operation

Clauses 32 and 33 cease to have any force or operation if the Implementation Agreement or Deed Poll is terminated in accordance with its terms.

33 Scheme

33.1 Implementation of Scheme

- (a) Each Scheme Participant and the Manager must do all things and execute all deeds, instruments, transfers or other documents as the Manager considers are necessary or desirable to give full effect to the terms of the Scheme and the transactions contemplated by it.
- (b) Without limiting the Manager's other powers under this clause 33, the Manager has power to do all things that it considers necessary or desirable to give effect to the Scheme and the Implementation Agreement.
- (c) Subject to the Corporations Act, the Manager, the other Westfield entities, Unibail or any of their respective directors, officers, employees or associates, may do any act, matter or thing described in or contemplated by this clause 33 even if they have an interest (financial or otherwise) in the outcome.
- (d) Clause 32 and this clause 33:
 - (i) binds the Manager and all of the Members (including those who do not attend the Scheme Meeting, those who do not vote at the Scheme Meeting and those who vote against the Scheme Resolutions); and
 - (ii) to the extent of any inconsistency, overrides the other provisions of this constitution (but, for the avoidance of doubt, remains subject to the Corporations Act and the Listing Rules).

33.2 Transfer and registration of Scheme Units

On the Implementation Date, at the time specified in the Transaction Steps:

(a) subject to the provision of the Scheme Consideration in the manner contemplated by clause 33.10 and Unibail having provided the Manager with written confirmation thereof, the Scheme Units, together with all

rights and entitlements attaching to the Scheme Units, must be transferred to TH Newco without the need for any further act by any Scheme Participant (other than acts performed by the Manager as attorney and agent for Scheme Participants under clause 33.21 of the Scheme) by:

- the Manager delivering to TH Newco duly completed and executed WFDT Transfer Forms executed on behalf of the Scheme Participants; and
- (ii) TH Newco duly executing the WFDT Transfer Forms and delivering them to the Manager for registration; and
- (b) as soon as practicable after receipt of the duly executed WFDT Transfer Forms the Manager must enter, or procure the entry of, the name of TH Newco in the Register in respect of all Scheme Units transferred to TH Newco in accordance with the terms of the Scheme.

33.3 Entitlement to receive Scheme Consideration

On the Implementation Date, in consideration for the transfer to TH Newco of the Scheme Units, each Scheme Participant will be entitled to receive the Scheme Consideration in respect of each of their Scheme Units in accordance with clause 33.10 of the Scheme.

33.4 Title and rights in Scheme Units

Subject to the provision of the Scheme Consideration for the Scheme Units as contemplated by clause 33.10 of the Scheme, on and from the Implementation Date TH Newco will be beneficially entitled to the Scheme Units transferred to it under the Scheme, pending registration by the Manager of TH Newco in the Register as the holder of the Scheme Units.

33.5 Scheme Participants' agreements

Under the Scheme, each Scheme Participant irrevocably:

- (a) acknowledges that clause 32 and this clause 33 binds the Manager and all of the Members from time to time (including those who do not attend the Scheme Meeting, do not vote at the Scheme Meeting or vote against the Scheme Resolutions);
- agrees to the transfer of their Scheme Units, together with all rights and entitlements attaching to those Scheme Units, in accordance with the terms of the Scheme;
- (c) agrees to the transfer to Unibail of all TH Newco Scheme Shares that are issued to it (or to the Third Party Nominee on its behalf) under the Scheme, together with all rights and entitlements attaching to those TH Newco Scheme Shares, in accordance with the terms of the WCL Scheme;
- (d) agrees that, if the Scheme Participant is an Ineligible Foreign Holder, all TH Newco Scheme Shares that would otherwise have been issued to it but instead are issued to the Third Party Nominee will be transferred to Unibail in accordance with the terms of the WCL Scheme, together with all rights and entitlements attaching to those TH Newco Scheme Shares;
- (e) agrees to the variation, cancellation or modification of the rights attached to their Scheme Units constituted by or resulting from the Scheme; and

(f) consents to the Manager and Unibail doing all things and executing all deeds, instruments, transfers or other documents as may be necessary or desirable to give full effect to this clause 33 and the transactions contemplated by it.

33.6 Warranty by Scheme Participants

Each Scheme Participant warrants to TH Newco and is deemed to have authorised the Manager to warrant to TH Newco as agent and attorney for the Scheme Participant by virtue of this clause 33.6, that:

- (a) all their Scheme Units (including any rights and entitlements attaching to those securities) transferred to TH Newco under the Scheme will, as at the date of the transfer, be fully paid and free from all Encumbrances;
 and
- (b) they have full power and capacity to sell and to transfer their Scheme Units (including any rights and entitlements attaching to those securities) to TH Newco under the Scheme.

33.7 Transfer free of encumbrances

To the extent permitted by law, all Scheme Units (including any rights and entitlements attaching to those securities) which are transferred to TH Newco under the Scheme will, at the date of the transfer of them to TH Newco, vest in TH Newco free from all Encumbrances and interests of third parties of any kind, whether legal or otherwise, and free from any restrictions on transfer of any kind not referred to in the Scheme.

33.8 Appointment of TH Newco as sole proxy

Subject to the provision of the Scheme Consideration for the Scheme Units as contemplated by clause 33.10 of the Scheme, on and from the Implementation Date until the Manager registers TH Newco as the holder of all of the Scheme Units in the Register, each Scheme Participant:

- (a) irrevocably appoints the Manager as attorney and agent (and directs the Manager in such capacity) to appoint TH Newco and each of its directors from time to time (jointly and each of them individually) as its sole proxy, and where applicable corporate representative, to attend securityholders' meetings, exercise the votes attaching to Units registered in its name and sign any securityholders resolution, and no Scheme Participant may itself attend or vote at any of those meetings or sign any resolutions, whether in person, by proxy or by corporate representative (other than pursuant to this clause 33.8(a)); and
- (b) must take all other actions in the capacity of the registered holder of Units as TH Newco directs.

The Manager undertakes in favour of each Scheme Participant that it will appoint TH Newco and each of its directors from time to time (jointly and each of them individually) as that Scheme Participant's proxy or, where applicable, corporate representative in accordance with clause 33.8(a) of the Scheme.

33.9 Transaction Steps

- (a) The Manager must implement, and must procure that WCL and WAT implement, the Transaction Steps.
- (b) Unibail must implement or procure the implementation of the Transaction Steps.

The Third Party Nominee must execute or provide duly executed powers (c) of attorney to such person(s) and in such form as indicated by Unibail for the execution of the TH Newco Deed of Issue

33.10 Provision of Scheme Consideration

The obligation of TH Newco to provide the Scheme Consideration to each Scheme Participant will be satisfied by TH Newco delivering to the Manager a copy of the duly executed TH Newco Deed of Issue and:

- (a) each Scheme Participant which is not an Ineligible Foreign Holder will hold beneficial title to one TH Newco Share for each Scheme Unit they hold: and
- (b) the Third Party Nominee will hold legal and beneficial title to each TH Newco Share in respect of which a Scheme Participant would have held beneficial title if they were not an Ineligible Foreign Holder.

33.11 Joint holders

In the case of Scheme Units held in joint names, any other document required to be sent under the Scheme, will be forwarded to either, at the sole discretion of the Manager, the holder whose name appears first in the Register as at the Scheme Record Date or to the joint holders.

33.12 Status of TH Newco Scheme Shares

Unibail must:

- issue the TH Newco Scheme Shares to ensure that each TH Newco (a) Scheme Share ranks equally with all other TH Newco Shares on issue at the time; and
- ensure that each TH Newco Scheme Share is duly and validly issued in (b) accordance with all applicable laws and the constituent and organisational documents of TH Newco free from any Encumbrance.

33.13 Determination of Scheme Units

To establish the identity of the Scheme Participants, dealings in Scheme Units will only be recognised by the Manager if:

- (a) in the case of dealings of the type to be effected using CHESS, the transferee is registered in the Register as the holder of the relevant Scheme Units on or before the Scheme Record Date; and
- (b) in all other cases, registrable transmission applications or transfers in registrable form in respect of those dealings are received on or before the Scheme Record Date at the place where the Register is kept.

33.14 Register

The Manager must register any registrable transmission applications or transfers of Units received in accordance with clause 33.13(b) of the Scheme on or before the Scheme Record Date.

33.15 No disposals after Effective Date

A holder of Scheme Units (and any person claiming through that holder) (a) must not dispose of or purport or agree to dispose of any Scheme Units or any interest in them after the Effective Date in any way except as set out in the Scheme and any such disposal will be void and of no legal effect whatsoever.

(b) The Manager will not accept for registration or recognise for any purpose any transmission, application or transfer in respect of Scheme Units received after the Scheme Record Date (except a transfer to TH Newco pursuant to the Scheme or any subsequent transfer by TH Newco or its successors in title).

33.16 Maintenance of Register

For the purpose of determining entitlements to the Scheme Consideration, the Manager will maintain the Register in accordance with the provisions of this clause 33.16 until the Scheme Consideration has been provided to the Scheme Participants and TH Newco has been entered in the Register as the holder of all the Scheme Units. The Register in this form will solely determine entitlements to the Scheme Consideration.

33.17 Effect of certificates and holding statements

Subject to provision of the Scheme Consideration and registration of the transfer to TH Newco contemplated in clauses 33.2 and 33.16 of the Scheme, any statements of holding in respect of Scheme Units will cease to have effect after the Scheme Record Date as documents of title in respect of those shares (other than statements of holding in favour of TH Newco and its successors in title). After the Scheme Record Date, each entry current on the Register as at the Scheme Record Date (other than entries in respect of Unibail or its successors in title) will cease to have effect except as evidence of entitlement to the Scheme Consideration.

33.18 Details of Scheme Participants

Within 3 Business Days after the Scheme Record Date, the Manager will ensure that details of the names, Registered Addresses and holdings of Scheme Units for each Scheme Participant, as shown in the Register at the Scheme Record Date are available to TH Newco in such form as TH Newco or Unibail reasonably requires.

33.19 Quotation of Units

Suspension of trading on ASX in Units will occur from the close of trading on ASX on the Effective Date.

33.20 Termination of quotation of Units

After the Scheme has been implemented, the Manager will apply:

- (a) for termination of the official quotation of Units on ASX; and
- (b) to have itself removed from the official list of the ASX.

33.21 Power of attorney

Each Scheme Participant, without the need for any further act by any Scheme Participant, irrevocably appoints the Manager and each of its directors and secretaries (jointly and each of them individually) as its attorney and agent for the purpose of:

- (a) executing any document necessary or expedient to give effect to the Scheme and the Transaction Steps including:
 - (i) executing any WFDT Transfer Forms; and
 - (ii) applying for shares in (and agreeing to become a member of) any of the entities in which Scheme Participants are to become members in accordance with the Transaction Steps; and

(b) enforcing the Deed Poll against Unibail,

and the Manager accepts such appointment.

33.22 Notices

- (a) If a notice, transfer, transmission application, direction or other communication referred to in the Scheme is sent by post to the Manager, it will not be taken to be received in the ordinary course of post or on a date and time other than the date and time (if any) on which it is actually received at the Manager's registered office or at the office of the Registrar.
- (b) The accidental omission to give notice of the Scheme Meeting or the non-receipt of such a notice by any Member will not, unless so ordered by the Court, invalidate the Scheme Meeting or the proceedings of the Scheme Meeting.

33.23 Further action by the Manager

The Manager will execute all documents and do all things (on its own behalf and on behalf of each Scheme Participant) necessary or expedient to implement, and perform its and each Scheme Participant's obligations under, the Scheme and the Transaction Steps.

33.24 No liability when acting in good faith

Neither the Manager nor Unibail, nor any of their respective officers, will be liable for anything done or omitted to be done in the performance of the Scheme in good faith.

33.25 Stamp duty

Unibail will pay all stamp duty (including any fines, penalties and interest) payable in respect of the Scheme.

33.26 Limitation of liability

Without limiting clause 19 but subject to the Corporations Act, the Manager will not have any liability of any nature whatsoever to Members, beyond the extent to which the Manager is actually indemnified out of the Assets, arising, directly or indirectly, from the Manager doing or refraining from doing any act (including the execution of a document), matter or thing pursuant to or in connection with the implementation of the Scheme.